Effective May 7, 2025



# **Table of Contents**

| Section # | New Name/Title                      | Page |
|-----------|-------------------------------------|------|
|           | Article I. Introduction             | 5    |
| 1.1       | Statement of Purpose                | 5    |
| 1.2       | Zoning Map Establishment            | 5    |
| 1.3       | Interpretation of Zoning Boundaries | 5    |
| 1.4       | Establishment of Zoning Districts   | 6    |
| 1.5       | Conformance with Regulations        | 7    |

|     | Article II. Interpretation | 8 |
|-----|----------------------------|---|
| 2.1 | Definitions                | 8 |

|      | Article III. General Regulation               | 21 |
|------|---|----|
| 3.1  | Lot in Two Zones                              | 21 |
| 3.2  | Lot of Record                                 | 21 |
| 3.3  | Change of Lot Area and Lot Dimensions         | 21 |
| 3.4  | One Main Building on a Lot                    | 21 |
| 3.5  | Yards and Open Space for Every Building       | 21 |
| 3.6  | Obstructions in Yards                         | 22 |
| 3.7  | Visibility at Intersections and Curves        | 22 |
| 3.8  | Building Lines                                | 22 |
| 3.9  | Lot Frontage                                  | 23 |
| 3.10 | Lots on Narrow Streets                        | 23 |
| 3.11 | Usable Lot Area                               | 23 |
| 3.12 | Rear Lots (Flag Lots)                         | 23 |
| 3.13 | First Floor Elevation                         | 23 |
| 3.14 | Exceptions to Height Requirements             | 24 |
| 3.15 | Required Street Frontage                      | 24 |
| 3.16 | Accessory Buildings                           | 24 |
| 3.17 | Location of Accessory Buildings               | 24 |
| 3.18 | Removal and Replacement of Topsoil            | 25 |
| 3.19 | Drainage                                      | 25 |
| 3.20 | Recreational Vehicles                         | 25 |
| 3.21 | Continuing Existing Nonconforming Uses        | 26 |
| 3.22 | Changes in Regulations                        | 26 |
| 3.23 | Residential Uses in Non-Residential Districts | 27 |

|     | Article IV. Residential Districts    | 29 |
|-----|--------------------------------------|----|
| 4.1 | Explanation of Use Tables            | 29 |
| 4.2 | Schedule of Permitted Main Use       | 29 |
| 4.3 | Schedule of Permitted Accessory Uses | 30 |

| 4.4  | Supplementary Regulations to the Tables of Permitted Uses | 31 |
|------|---|----|
| 4.5  | Conversion for Related Family Unit (In-Law Apartments)    | 33 |
| 4.6  | Home Occupations and Professional Offices                 | 33 |
| 4.7  | Conversion for Bed and Breakfast                          | 34 |
| 4.8  | Open Space Districts (OS)                                 | 34 |
| 4.9  | Dimensional Standards for Residential Districts           | 35 |
| 4.10 | Apartment Buildings                                       | 35 |
| 4.11 | Standards for Multi-Family Uses                           | 36 |
| 4.12 | Multi-Family Permitted Uses                               | 36 |
| 4.13 | Homeowners' Associations                                  | 36 |
| 4.14 | Criteria for Building and Sites                           | 36 |
| 4.15 | Public Improvements                                       | 37 |
| 4.16 | Maintenance and Operation                                 | 37 |
| 4.17 | Parking, Landscaping, and Buffers                         | 36 |
| 4.18 | Filing Requirements                                       | 38 |

|      | Article V. Commercial Districts                           | 39 |
|------|---|----|
| 5.1  | Explanation of Tables                                     | 39 |
| 5.2  | Schedule of Permitted Main Use                            | 39 |
| 5.3  | Schedule of Permitted Accessory Uses                      | 41 |
| 5.4  | Supplementary Regulations to the Tables of Permitted Uses | 41 |
| 5.5  | Performance Standards – Business and Industrial Uses      | 43 |
| 5.6  | Work/Live Floating Zone District                          | 43 |
| 5.7  | Standards for Business and Industrial Districts           | 46 |
| 5.8  | Consolidation of Parcels                                  | 46 |
| 5.9  | Smaller Lots in Business Zones                            | 46 |
| 5.10 | Commercial and Industrial Development                     | 47 |
| 5.11 | Criteria for Approval of Special Use Permit               | 49 |
| 5.12 | Village Overlay District                                  | 50 |
| 5.13 | Adaptive Reuse  | 56 |

|     | Article VI. Site Development Standards     | 59 |
|-----|--|----|
| 6.1 | Screening and Landscaping Standards        | 59 |
| 6.2 | Off-Street Parking and Loading Regulations | 61 |
| 6.3 | Sign Regulations                           | 68 |

|     | Article VII. Special Regulations                      | 74 |
|-----|---|----|
| 7.1 | Seasonal Dwellings                                    | 74 |
| 7.2 | Earth Removal and Blasting                            | 74 |
| 7.3 | Regulations Concerning Kennels                        | 79 |
| 7.4 | Regulations Concerning the Keeping of Animals         | 81 |
| 7.5 | Regulations Concerning Mobile Homes and Camping Areas | 84 |
| 7.6 | Regulations Concerning Automotive Sales               | 86 |
| 7.7 | Car Wash  | 88 |
| 7.9 | Pistol, Rifle and Skeet Trap Shooting                 | 88 |

| 7.10 | Forestry-Repealed 8-16-18                          | 88  |
|------|--|-----|
| 7.11 | Day Care   | 88  |
| 7.12 | Adult Bookstores                                   | 89  |
| 7.13 | Alcoholic Beverages                                | 90  |
| 7.14 | Flood Plain District                               | 91  |
| 7.15 | Soil Erosion and Sedimentation Control Regulations | 98  |
| 7.16 | Telecommunications Towers                          | 100 |
| 7.17 | Age-Restrict Housing                               | 102 |
| 7.18 | Solar Energy Systems                               | 107 |
| 7.19 | Accessory Agricultural Activities                  | 113 |
| 7.20 | Recreational Club                                  | 115 |
| 7.21 | Medical Marijuana                                  | 115 |
| 7.22 | Recreational Cannabis                              | 116 |
| 7.23 | Lawn Care / Tree Removal Service                   | 119 |

|      | Article VIII. Administration and Enforcement                            | 119 |
|------|---|-----|
| 8.1  | Zoning Permits  | 120 |
| 8.2  | Certificate of Zoning Compliance  | 120 |
| 8.3  | Site Plan Requirements  | 121 |
| 8.4  | Changes and Modification to an Approved Site Plan or Special Use Permit | 122 |
| 8.5  | Time Period of Permits  | 123 |
| 8.6  | One- and Two-Family Dwellings   | 123 |
| 8.7  | Waiver of Requirements  | 124 |
| 8.8  | Planning Review   | 124 |
| 8.9  | Applications  | 124 |
| 8.10 | Performance Bond  | 124 |
| 8.11 | Procedures for Approval of Applications                                 | 124 |
| 8.12 | All Other Applications for Permits                                      | 128 |
| 8.13 | Public Hearings   | 128 |
| 8.14 | Zoning Amendments   | 129 |
| 8.15 | Enforcement   | 130 |
| 8.16 | Zoning Board of Appeals   | 130 |
| 8.17 | Schedule of Fees - Repealed   | 132 |
| 8.18 | Expert Review of Applications   | 132 |
| 8.19 | Continuity Clause   | 132 |
| 8.20 | Validity  | 133 |
| 8.21 | Amendments  | 133 |
| 8.22 | Repealing Clause  | 133 |
| 8.23 | Revised Dates   | 134 |

#### **Article I**

#### Introduction

#### 1.1 Statement of Purpose:

Pursuant to the authority granted by Chapter 124 of the General Statutes of the State of Connecticut, as amended, and in order to promote the health, safety, and general welfare of the residents of the Town of Stafford; to secure safety from fire, panic, and other dangers; to provide adequate light, and air; to prevent the overcrowding of land; to avoid undue concentration of population; to facilitate adequate provision for transportation, water, sewerage, schools, parks, and other requirements; to avoid congestion in the streets; to encourage the most appropriate use of land throughout the Town: now therefore, the Planning and Zoning Commission of the Town of Stafford hereby adopts the following Zoning Regulations.

#### 1.2 Zoning Map Establishment:

The boundaries of the zoning districts described in Section 1.3 are shown on a map entitled "Town of Stafford Official Zoning Map" dated July 2010 and filed in the office of the Town Clerk. The map described and any amendments thereto are hereby made part of these Regulations.

#### 1.3 Interpretation of Zoning Boundaries:

Unless otherwise indicated on the Official Zoning Map, the boundary lines of the zoning districts are: the center line of streets, or the middle of the channel of waterways, or the center line of utility rights-of-way, or the boundary lines of state forest, parks, or reservations, or the center line of main tracks of railroad lines. Where the zoning district boundary is a contour line, the line shall be designated on the Official Zoning Map by its elevation in feet above mean sea level using U.S. Geological Survey datum. Where a zoning district boundary is shown parallel to a street, such a boundary shall be interpreted as running parallel to the nearest street line and at such a distance thereon as indicated on the official Zoning Map. If appropriate distance is not indicated, such a distance shall be as follows measured at right angle from the center line of the street right-of-way.

- 750 feet for rural and single-family residence district AAA
- 300 feet for residence districts AA and A
- 300 feet for residence district B
- 500 feet for residence district C
- 250 feet for local business district LB
- 250 feet for highway business district HB
- 300 feet for industrial district IN

• 300 feet for highway industrial district

Where a lot of record falls into two or more zoning districts, any use allowed in either district shall be permitted on the lot; but in no case shall the uses allowed in one district but prohibited in the other be extended more than thirty (30) feet into the district prohibiting such uses.

In case of uncertainty as to the location of any district boundary line, the determination thereof shall be made by the Commission.

#### 1.4 Establishment of Zoning Districts:

For the purpose of these Regulations, the Town of Stafford is divided into the following zoning districts:

#### A. Residence districts:

AAA — Rural and single-family residence

AA — Single-family

A — Single family residence

B — General residence

C — General residence

#### B. Commercial districts:

LB — Local Business

HB — Highway Business

CB — Central Business

#### C. Industrial districts:

IN — Industrial districts

HI – Highway Industrial

#### D. Miscellaneous districts:

WM — Water Management

OS — Open Space

WL – Work/Live Floating Zone District. The district is created but has no established boundaries. Such WL district, at the Commission's discretion through the zone change process, may be established anywhere within the Town. (Effective 12/15/04)

VOD — Village Overlay District

# 1.5 Conformance with Regulations:

Except as hereinafter specified, no land, building, or premises, or part thereof shall hereafter be used and no building, or part thereof, or other structure shall be constructed, reconstructed, extended, enlarged, moved, or altered except in conformity with the regulations herein prescribed for the zone in that it is located.

# **Article II** Interpretation

#### 2.1 Definitions

**Abut, Abutting:** Having a common boundary.

Accessory Building, or Use: A building, or use customarily incidental to the principal building, or to the principal use of the land. Except in the case of barns or sheds used for the indoor storage of agricultural products, or equipment, any accessory building, or use shall be located on the same lot under the same ownership. *Effective* 11/15/03 accessory uses located in building housing the principal use are limited to 25% of the floor area of the building.

**Adjacent**: Near, close by. The term "adjacent" includes the meaning of the terms *abut* and *adjoin* as defined; however, it is not intended to imply any definite distance or boundary.

**Adjoin, Adjoining:** Having a common boundary with or without the intervention of a street. Thus the term *adjoin* includes the meaning of the term *abut* as defined.

**Adult Bookstore:** An establishment that has a substantial, or significant portion of its stockin-trade, books, magazines, newspapers, films, audio and video tapes, devices, slides, or other photographic, written, or audio material for rent, or sale, that might be considered "obscene as to minors" as defined in Connecticut State Statute Section 53a-193 of the Connecticut General Statutes.

**Age-Restricted Housing Developments:** A residential community restricted to residents aged 55 and over, their spouses, and children at least 18 years old. (Effective 6/11/08)

**Agriculture** Connecticut General Statutes, Sec. 1-1 (q) Except as otherwise specifically defined, the words "agriculture" and "farming" shall include cultivation of the soil, dairying, forestry, raising, or harvesting any agricultural or horticultural commodity, including the raising, shearing, feeding, caring for, training, and management of livestock, including horses, bees, poultry, fur-bearing animals and wildlife, and the raising or harvesting of oysters, clams, mussels, other molluscan shellfish or fish; the operation, management, conservation, improvement or maintenance of a farm and its buildings, tools and equipment, or salvaging timber or cleared land of brush or other debris left by a storm, as an incident to such farming operations; the production or harvesting of maple syrup or maple sugar, or any agricultural commodity, including lumber, as an incident to ordinary farming operations or the harvesting of mushrooms, the hatching of poultry, or the construction, operation or maintenance of ditches, canals, reservoirs, or waterways used exclusively for farming purposes; handling, planting, drying, packing, packaging, processing, freezing, grading, storing or delivering to storage or to market, or to a carrier for transportation to market, or for direct sale any agricultural or horticultural commodity as an incident to ordinary farming operations, or, in the case of fruits and vegetables, as an incident to the preparation of such fruits or vegetables for market or for direct sale. The term "farm" includes farm buildings, and accessory buildings thereto, nurseries, orchards, ranges, greenhouses, hoop houses and other temporary structures or other structures used primarily for the raising and, as an incident to ordinary farming operations, the sale of agricultural or horticultural commodities.

The term "aquaculture" means the farming of the waters of the state and tidal wetlands and the production of protein food, including fish, oysters, clams, mussels and other molluscan shellfish, on leased, franchised and public underwater farm lands.

**Agriculture – Commercial:** A Farm producing farm products for sale by wholesale, or for sale at locations (not including farm markets) other than farm property on which they were produced or a Company or individuals(s) required to file IRS Form 1040f.

**Amusement Arcade:** A facility that primarily contains amusement devices. (Effective 11/15/03)

Amusement Devices: Game devises intended for entertainment purposes, including, but not limited to, pool tables, electronic devices, or video devices. (Effective 11/15/03)

**Animal Enclosure:** A fence, or other method, used to contain livestock and other domestic animals within a predetermined outdoor area, excluding common house hold pets such as dogs and cats.

**Animal Management**: addresses shelter provisions, outdoor keeping area, pasture and manure management needs depending on the specific animal. Information available from the CT Department of Agriculture, CT Farm Bureau, and or the USDA as well as Natural Resources Conservation Service. These references should be utilized in preparing a specific management plan.

**Apartment:** An apartment is a segregated portion of a dwelling that is provided with bathroom and toilet facilities and with kitchen facilities in a kitchen or kitchen alcove.

**Apartment House:** An apartment house is a building arranged, intended, or designed to be occupied by three, or more families living independently of each other and doing their cooking upon the premises, or by three or more individuals, or groups of individuals living independently but having a common heating system and general dining room.

**Barn:** A building for storing hay, grain, etc., and often for housing livestock. (Effective 2/15/95)

**Basement:** The lowest story of a building below the main floor and wholly or partly below the average level of the ground surrounding the building or structure.

**Bed and Breakfast**: An owner-occupied one-family detached building in that food and/or lodging may be provided to overnight paying guests. The provision of such services is subordinate to the principal use of the building as a residence. The lodging may be provided through the services of an online transient reservation system.

**Boarding House:** Same as a *rooming house* except that meals are normally served to lodgers.

**Brewery:** A facility where beer or cider is manufactured, stored, bottled and sold at wholesale or at retail in sealed containers for consumption off premises or offered for on the premises tastings.

**Brewpub:** A facility where beer or cider is manufactured, stored, bottled, sold at wholesale or at retail in sealed bottles or other sealed containers for consumption off premises or sold to be consumed on the premises in a room that is ancillary to the production of beer or cider, with or without the sale of food, and in accordance with Section 7.13.

**Building:** Any structure having a roof and intended for the shelter, housing, or enclosures of persons, animals, or materials. Any other structure more than eight feet high shall be considered as a building, including a fence, or wall, excluding a public utility pole, or a flagpole.

**Building Coverage:** That percentage of total lot area covered by the combined area of all buildings on the lot.

**Building Height:** The vertical distance from the finished grade at any point under consideration to the highest point of flat, or mansard roofs, or to the mean level between the eave and the ridge for gable, hip, or gambrel roofs. Chimneys, spires, towers, and similar projections not intended for human occupancy shall not be included in the height, provided that any such projection does not have an aggregate area greater than 25% of the roof area.

**Building Line:** A line parallel to a street at a distance equal to the required front yard or at a greater distance when otherwise established by the Town of Stafford or when established by the owner and recorded in the land records of the Town of Stafford.

**Building, Main:** A building in that is conducted the main or principal use of the lot on that said building is situated.

**Building Permit:** A permit issued by the Building Inspector upon application, certifying that a proposed structure, or any extension, or structural alteration thereof, conforms with the requirements of the Building Code and all other regulations, or codes that are specified in the building permit.

**Cabin:** A building having a design or character suitable for seasonal or temporary living purposes.

**Cafe:** As defined by the State of Connecticut Liquor Control Commission. (Effective 2/15/95)

**Camp:** An area of land under the same ownership on that is located a cabin or other shelter suitable for seasonal or temporary living purposes.

Camper: See Recreational vehicle.

**Camping Area:** Any area devoted to camping by means of tents, camp trailers (either self-propelled or without motor power), two-wheel tent trailers, or any other object used for camping where a charge is made to the camper for the use of the area on a limited basis. For the purpose of these regulations, *camping area* shall also include *boarding camp*.

**Canopy:** A roofed structure with open sides normally attached to and considered a part of a main building and observing all yard requirements.

**Club, Recreational:** An organization catering exclusively to member and their guests, provided that the purpose of the club is not conducted primarily for gain and that there are

not conducting any commercial activities except as required generally for the membership and purposes of this club.

**COMMERCIAL VEHICLE** - Any vehicle or equipment regularly used to carry, deliver, handle or move goods in the conduct of a business, commerce, profession or trade, and which has two or more of the following characteristics: 1. Exceeds (gross vehicle weight rating) GVWR of nine thousand pounds (9,000); 2. Exceeds seven (7) feet in height from the base of the vehicle to the top; 3. Exceeds twenty (20) feet in length; 4. Has more than two (2) axles; 5. Has more than four (4) tires in contact with the ground; 6. Used, designed and built to carry more the eight (8) passengers; 7. Designed to sell food or merchandise from the vehicle or trailer itself; 8. Banners, signs, logos, advertising or markings identifying the owner or registrant, a trade, business, service or commodity; 9. Has modifications such as but not limited to platform rack, ladder rack, or mechanical equipment such as a hoist used to facilitate the carrying of goods or equipment; 10. Commercial plate or registration. Commercial Vehicles used on a farm for activities associated with that farm, and which are not in violation of any other Town Regulation and/or Town Ordinance, are exempt from the definition of Commercial Vehicle. The following types of vehicles when regularly used to carry, deliver, handle or move goods in the conduct of business, commerce, profession or trade shall be considered commercial vehicles - step vans, cargo vans, box trucks, flat bed or stake bed trucks, buses semitrailers, tractor trailers dump trucks wreckers and trailers for commercial purposes. The following types of equipment shall also be considered commercial vehicles - earth moving equipment, cement mixers trenching and pipe laying equipment and other similar type of contractors/ construction/ site work equipment.

**Commission:** The Planning and Zoning Commission of the Town of Stafford.

**Common Driveway:** A driveway that provides access to two (2) or more lots. (Effective 11/15/03)

**Congregate Housing:** A multi-family elderly and/or handicapped facility in that residents have their own apartments and share meals in a central dining room. Other services may also be provided that help residents maintain their independence.

**Convalescent Home:** A home for the aged, or any establishment, other than hospitals, where three or more persons suffering from, or afflicted with, or convalescing from any infirmity, disease, or ailment are habitually kept, boarded, or housed for remuneration.

**Court:** A horizontal open space, other than a yard, on the same lot with a building that is bounded on two or more sides by opposite walls of one or more buildings.

**Coverage, building:** Lot area covered by the combined area of all structures on the lot, including porches, decks, ramps, balconies plus the area covered by pools, tennis courts and other structures.

**Coverage, building ratio:** That percentage of usable lot area (as defined in Section 3.11) covered by the combined area of all structures on the lot, including porches, decks, ramps, balconies plus the area covered by pools, tennis courts and other structures.

**Coverage, impervious:** The area of all buildings on a lot plus the area covered by parking and loading areas and driveways (whether paved or unpaved)

**Cutting:** The cutting or removal of forest tree species on a lot for the purpose of preparing a site for the construction of a building or other structure.

**Cutting Plan:** A plan showing the existing mix of forest trees, their approximate height, age, and density; a description of the cutting or removal activities to be undertaken and any other information that may be necessary and reasonably required.

**Drive through facilities:** An establishment that distributes goods and services directly to consumers within their vehicle. (Effective 4/1/06)

**Dwelling:** A building used exclusively as living quarters for one or more families. (Effective 2/15/95) The terms *dwelling*, *attached dwelling*, *detached dwelling*, and *dwelling unit* shall not be deemed to include hotel, motel, boarding, rooming house, convalescent or nursing home, trailer, tourist home, or tent. In the case of buildings having two or more portions divided by party walls forming a complete separation above the basement, each such portion shall be considered to be a separate dwelling.

**Dwelling, attached:** A dwelling having any portion of a wall in common with another dwelling.

**Dwelling, detached:** A dwelling with open spaces on all sides.

**Dwelling, In-Law-Apartment:** An accessory dwelling unit to a single-family home for an immediate family member.

**Dwelling, multi-family:** A dwelling containing more than two dwelling units.

**Dwelling, one-family:** A dwelling or portion thereof providing complete housekeeping facilities for one family only.

**Dwelling Unit:** A building or portion of a building used as living quarters for one family. (Effective 6/11/08)

**Earth Removal:** Commercial earth removal or mining operation for the purpose of excavation, processing, and/or blasting and removal of sand, gravel, stone, loam, dirt, or other earth products from the premises on a continuing basis; excluding bona fide farming operations who manufacture loam on site and permitted retail operations where products are delivered to and sold on the premises, retail only. (Effective 4/25/2007)

**Family:** One or more persons living together as a single non-profit housekeeping unit, including domestic servants and gratuitous guests.

**Farm:** A tract of land used principally for agriculture, with or without an associated single-family dwelling.

**Flight:** A combination of one landing and one takeoff by which a helicopter arrives at and departs from a heliport.

**Floor area, gross:** that sum of the horizontal area of all floors of a building as measured from the exterior walls. Such area shall not include any space with the headroom of less than seven (7) feet.

**Floor area ratio:** The ratio of the gross floor area divided by usable lot area (as defined in section 3.11)

**Food Trailer:** A non-motorized, towed vehicle, usually stationed at a specific location. It offers similar food services but requires a vehicle to tow it to different locations (

**Food Truck:** A motorized vehicle that can drive to different locations, typically serving a wide range of foods, from street food to gournet foods

**Garage:** A building or place for sheltering, cleaning, or repairing motor vehicles. (Effective 2/15/95)

**Grade, finished:** The completed surfaces of lawns, walks, and roads brought to grades as shown on official plans or designs relating thereto.

**Grooming Facility:** A commercial facility, other than a commercial kennel maintained for the purposes of grooming dogs and cats with no accommodations for overnight boarding of animals (Effective 12/19/2003) (Rev. 9/28/10)

**Gross Floor Area:** The sum of the horizontal area of all floors of a building, measured by exterior dimensions.

**Ground Cover:** A medium used in a confined area to check or prohibit the growth of undesirable plant materials. Ground cover may consist of plants such as pachysandra and myrtle or of materials such as white gravel, brick, or stone paving.

**Group Care Facility:** A supervised residence facility that houses persons who are handicapped, aged, or disabled, or in need of rehabilitation but are not acutely ill and are provided services to meet their needs. It does not include an institution as defined in Section 19-32 of the Connecticut General Statutes and required to be licensed pursuant to the provisions of Sections 19-32 to 19-42 of the Connecticut General Statutes.

**Hedge:** A hedge shall provide complete visual screening and consist of evergreens at least four (4) feet in height at the time of planting, and it shall be maintained at a height of at least six (6) feet.

**Heliport:** An area of defined dimensions, either at ground level or elevated on a structure, designated for the landing and takeoff of helicopters, which may be restricted solely for that purpose.

**Home for the Aged:** An establishment other than a hospital that furnishes for remuneration food, shelter, laundry, and other non-medical services to three or more persons over the age of sixty (60) years.

Home Occupation: Customary home occupation for gain carried on entirely within the dwelling by residents thereof and requiring only customary home equipment, provided the use: 1) is clearly incidental to the use of the dwelling for dwelling purposes; 2) does not change the residential character of the dwelling in any visible manner; 3) does not create objectionable noise, odor, vibrations, or unsightly conditions noticeable off the premises; 4) does not create interference with radio and television reception in the vicinity; 5) does not create a health or safety hazard; 6) does not employ a non-resident help for that purpose; 7) no trading in merchandise is carried on; 8) no external or internal alterations, or construction features not customarily found in a home are required. Such use shall be confined to not more than 25% of the livable floor area of the dwelling or to space in an outbuilding no larger than 25% of the ground floor area of the dwelling.

**Hotel:** A building or group of buildings providing lodging for persons, with or without meals, and intended, designed, and used primarily for the accommodations of transients. The term *hotel* shall be deemed to include the terms *motel*, *motor inn*, and *inn*.

**Immediate Family:** Spouse, mother, father, mother-in-law, father-in-law, daughter, son, sister, or brother.

**In-Law Apartment:** An apartment accessory to an owner-occupied single family dwelling to be occupied by a parent, grandparent, aunt, uncle, brother, sister, brother-in-law, sister-in-law, child, child-in-law or a care giver as certified by a licensed physician.

**Junk Yard:** Any place in or on that is stored or deposited old material. This includes glass, paper, metal, cordage, scrap, or other waste, or second-hand or unused substances, any material that has been or was intended to be a part of any fabricated object or of any vehicle or of any form of conveyance of any kind or description.

**Kennel**: The keeping of five (5) or fewer dogs as exempt under section 7.3 of these regulations. (Rev. 9/28/10)

**Kennel, Commercial:** A kennel maintained for boarding or grooming dogs or cats, and includes, but is not limited to, any veterinary hospital that boards or grooms dogs or cats for non-medical purposes. (Effective 12/19/2003) (Rev. 9/29/20)

**Kennel Non-commercial:** Six (6) or more dogs over the age of six (6) months kept under one ownership at a single location and bred for show, sport, or sale. Any owner or breeder who breeds more than two litters of dogs annually shall apply to the Town Clerk for a kennel license. (Effective 12/19/2003) (Rev. 9/28/10)

**Landscaping, Landscaped:** The term *landscaped* or *landscaping* shall mean that an area be at least covered with grass or ground cover. Any additional planting is either specifically required by the ordinance or left to the discretion of the property owner.

**Lawn Care Business:** Business dealing with the maintenance of lawns to include grass cutting, leaf removal, pruning, mulching, and planting or removal of trees, shrubs, plants, and grass. For the purpose of these regulations, the locations of a lawn care business shall be the primary location of the storage of related lawn care equipment. (Effective 5/1/2000)

**Licensed Medial Marijuana Dispensary (Dispensary):** A pharmacist licensed pursuant to Chapter 400j of the Connecticut General Statues, who the Department of Consumer Protection has licensed to acquire, possess, distribute and dispense medical marijuana pursuant to CGS Section 21a-408 to 21a408q inclusive of an 'Act Concerning the Palliative Use of Marijuana' and who is located on the premises of a pharmacy licensed by the Connecticut Commission of Pharmacy.

**Licensed Medical Marijuana Producer (Producer):** A person or organization licensed by the Connecticut Department of Consumer Protection as a producer under CGS 21a-408-1 through 21a-408-70, whose purpose is to cultivate marijuana for palliative use, including selling, delivering, transporting and distributing such marijuana, but only to State licensed dispensaries under Sections 1 through 15, inclusive of the Act Concerning the Palliative Use of Marijuana.

**Livable Floor Area:** The sum of the gross finished horizontal areas of each of several floors of a dwelling with structural head room of seven feet, six inches (7' 6") or more, provided with heat and suitable for year-round occupancy. Basement, attic, open or screened porch, breezeway, terrace, or other similar areas shall not be included in computing the minimum livable floor area.

**Living Space:** The total area within a unit designated for human habitation including, but not limited to, storage, closets, hallways, and bathrooms. Calculations for the unit are the sum of the unit's total area measured horizontally from its defining interior walls. (Effective 9/28/10)

**Livestock:** Shall mean domestic animals other than household pets including, but not limited to, horses, cows, goats, and sheep. (Rev. 9/28/10)

**Lot:** A plot or parcel of land occupied or capable of being occupied by one principal building and the accessory buildings or uses customarily incidental to it, including such open spaces as are required by these regulations. In the case of public, institutional, or commercial buildings, a group of buildings under the same ownership may be considered as occupying the same lot.

Lot Area: The required lot area shall be deemed to be the product of multiplying the average width of a lot by a lot depth not greater than three times such width, regardless of the depth of the lot.

**Lot Corner:** A lot having two adjacent sides facing a street or streets so that the interior angle of the intersection is not more than 120 degrees; also a lot having two adjacent sides forming tangents of a curve with an inside radius greater than fifty (50) feet, all of that face a street or streets.

**Lot Depth:** The mean horizontal distance between the front and rear lot lines, measured in the general direction of the side lot lines. In determining the required depth of a lot, any portion of said lot that is in excess of the minimum lot area need not be included.

**Lot Frontage** is the distance between lot sidelines measured along the street line. In the case of lots having frontage on or adjacent to a street curve, required lot frontage may be measured at the building line on such lots.

**Lot Line:** The property lines bounding a lot as defined herein.

**Lot Line, front:** In the case of a lot abutting upon only one street, the line separating the lot from the street.

**Lot line, rear:** The lot line that is generally opposite the front lot line: If the rear lot line is less than ten (10) feet in length or if the lot comes to a point at the rear, the rear lot line shall be deemed to be a line parallel to the front line not less than ten (10) feet long, lying wholly within the lot and farthest from the front line.

**Lot Line, side:** Any lot line that is not a front lot line or a rear lot line, as defined herein.

Lot, minimum width of: The distance between the side lot lines measured in a straight line at right angles to the mean direction of such side lot lines that line of measurement shall touch, but not be in front of, the building line.

Main Building: See Building, main.

Main Use: See Use, main.

**Minor Landscaping:** Less than one (1) acre of disturbed area, associated with construction, building alterations, or installation of subsurface utilities. (Effective 4/25/2007)

**Mobile Home:** A structure capable of being transported on wheels that is designed to permit permanent occupancy thereof as a dwelling for one or more persons.

**Mobile Home Park:** A tract of land where two or more mobile homes are parked or a tract of land that is used or intended for the purpose of supplying to the public parking spaces for two or more mobile homes.

**Mobile Office:** Similar to a mobile home except that such vehicle is not intended for dwelling purposes.

**Non-conforming Building:** A building that does not conform to all the applicable provisions of these regulations.

**Non-conforming Use:** A use of land, building, or premises that is not a use permitted by the provisions of these regulations for the district in that such land, building, or premises is situated.

Nursing Home: See definition of *Convalescent Home*.

**Office:** A room, set of rooms, or building where the business of a commercial or industrial organization or of a professional person is transacted. Absent "Special Approval" from the Planning & Zoning Commission, an office shall not contain residential amenities such as a full bath or full kitchen. (Effective 2-15-95)

**Open Space:** An unoccupied space open to the sky on the same lot as the building.

**Opposite Wall:** For the purpose of defining the term *court*, walls shall be deemed opposite wherever the walls are parallel, nearly parallel, or at angles of less than 90 degrees.

**Owner of Record:** Wherever the term *owner of record* is used in these regulations in conjunction with a public hearing or public notice, it shall mean the owner whose name is recorded in the Town Clerk's office at the time when the mailing lists for said hearings and notices are prepared.

**Parking Space:** A reasonably level space, available for the parking of one (1) motor vehicle exclusive of passageways, driveways, and other means of circulation or access. Spaces shall be nine feet by eighteen feet (9' x 18') minimum.

**Premises:** That portion of a lot or structure or building actually in use for the specific purpose or use under consideration.

**Recreational Vehicle:** A vehicle that can be registered for highway use and that is capable of being occupied with sleeping and/or cooking accommodations, on a temporary basis and may or may not contain sanitary facilities.

**Re-use.** Renovation of an existing historic structure and conversion of the existing use to a use that is not otherwise allowed in the zone in which the structure is located. Uses allowable

as adaptive re-use include any use that is permitted by site plan or special permit approval in any zoning district, as specified by these Regulations.

**Roadside Stand:** A direct marketing operation up to eight hundred (800) square feet that is seasonal in nature and features on-farm produced as well as locally produced agricultural products, enhanced agricultural products and handmade crafts. At least fifty percent (50%) of the agricultural products must be raised on the property where the stand is located or on other parcels owned or leased by the proprietor. The stand can be within a temporary structure or an already existing permanent structure. All crafts must be produced on the property where the stand is located.

**Roadside Market:** A direct marketing operation with or without a permanent structure that offers outdoor and/or indoor commerce. Greenhouses or pick-your-own areas are not included in the size. Such an operation features on-farm produced as well as locally produced agricultural products, enhanced agricultural products, and handmade crafts. At least fifty per cent (50%) of the agricultural products must be raised on the property where the stand is located or on other parcels owned and/or leased by the proprietor. All crafts must be produced on the property where the market is located.

**Rooming House:** A dwelling, part of that is occupied by the owner or tenant of the building as his permanent residence, in that rooms are provided for compensation, to not more than three (3) persons. Supervised group quarters shall be considered a rooming house or a boarding house.

**Screening Fence or Screening Wall:** These are devices for complete visual screening. They shall be at least six (6) feet in height and three-quarters (3/4) solid.

**Sign** is any device for advertisement and/or visual communication that is used for the purpose of bringing the subject thereof to the attention of the public, including the devices displayed up to three (3) feet behind windows and visible from outside the building, but not including any flag or insignia of any governmental organization.

**Sign, ground** is any sign affixed to the ground by its own support and/or foundation.

**Sign, portable:** Any sign used or intended to be used in various locations.

**Sign Permit:** Same as *Building Permit*.

**Site Plan or Plot Plan:** A plan of property drawn to scale with locations of structures, grading, paving, and other improvements indicated thereon.

**Story:** That portion of a building that is between the surface of a floor and the surface of the next floor above or, in its absence, the next ceiling above.

**Street:** Any way that is an existing town or state highway or any way shown on a recorded subdivision map duly approved by the Planning and Zoning Commission, or any map duly recorded and approved prior to the adoption of Subdivision Regulations.

**Street Line:** The line separating the street right-of-way from adjoining property, not the paved or traveled roadway.

**Structure:** Anything constructed or erected, including a building, the use of that requires location on or under the ground, or attachment to something having location on the ground, except fences, or walls.

**Swimming Pool:** Swimming pools include above ground swimming pools or in-ground swimming pools. Swimming pools shall be considered an accessory structure for purposes of compliance with district standards. (Effective 11/15/03)

**Temporary Earth Removal Operation:** An operation associated with landscaping or agricultural use where the disturbed area exceeds one (1) acre; excluding minor private residential landscaping of less than one half (1/2) acre not associated with a commercial operation. (Effective 4/25/2007)

**Trailer:** A vehicle without means of propulsion, which can be used for hauling or storing of materials or goods and is capable of being readily moved by a tractor or other vehicle.

**Training Facility:** Any place, other than a commercial kennel or grooming facility, that is maintained as a business where dogs are trained. (Effective 12/19/2003)

**Trees, flowering:** Trees such as Dogwood, Redbud, or Crabapple.

**Trees, large:** Deciduous shade trees such as Sugar Maple, Red Oak, or London Plane, and conifers such as White Pine, Austrian Pine, or Canadian Hemlock. Three-quarters ( $\frac{3}{4}$ ) of the required large trees shall be at least two and one-half to three inches ( $\frac{21}{2}$ -3") in caliper at the time of planting and one-quarter ( $\frac{1}{4}$ ) shall be at least four to four and one-half inches ( $\frac{4}{4}$ - $\frac{41}{2}$ ") in caliper at the time of planting.

**Use, accessory:** A use customarily incidental and subordinate to a main use and located on the same lot with such main use.

**Use, main:** The specific purpose for that land, water, or a structure is designed, arranged, or intended, or for that it is, or may be occupied, or maintained.

**Visual Screening, complete:** A type of screening that affords a year-round effect and through that the screened object is obscured.

**Visual Screening, partial:** A type of screening through that the screened object is partially visible.

**Waiting Position:** A space in a line of approach nine feet (9') wide by twenty feet (20') long.

**Wholesale:** The sale of products in large quantities to commercial purchasers for the subsequent retail sale of such products to the general public. Compare to "retail sale."

**Work/Live Zone:** A zoning district for establishing and operating work/live occupancies as a primary commercial/industrial use, in that the proprietor would be allowed to reside as a secondary land use activity.

Yard: An open space on the same lot with a structure that lies between said structure and the nearest lot line and that is unoccupied except as may be specifically authorized in this regulation. In measuring a yard, as hereafter provided, the line of structure shall be deemed to mean a line parallel to the nearest lot line, drawn from a point of a structure nearest to such

lot line. Such measurement shall be taken at right angle from the line of the structure, as defined herein, to the nearest lot line.

**Yard, front:** A yard extending across the full width and/or length of the lot and lying between the front lot line and the nearest line of a structure.

**Yard, rear:** A yard extending across the full width of the lot and lying between the rear lot line and the nearest line of the building.

**Yard, side:** A yard between the sideline of a lot and the nearest line of the building and extending from the front yard to the rear yard or, in the absence of either such yards, to the front or rear lot line as the case may be.

**Zoning Permit:** A permit issued by the Zoning Enforcement Officer upon application, certifying that a proposed land use or any extension or alteration thereof, conforms with requirements of the Zoning Regulations.

#### THIS PAGE INTENTIONALLY LEFT BLANK

#### Article III.

# General Regulation

#### 3.1 Lot in Two Zones:

In the case of a lot of record lying in more than one district, the provision of the less restrictive district may be applied for a distance of not more than thirty (30) feet into a more restrictive district, provided that such lot has frontage on a street in the less restrictive district.

#### 3.2 Lot of Record:

The lot frontage and lot size requirements of these regulations or amendments thereto shall not prevent construction of a permitted building or establishment of a permitted use on a lot that:

- A. Was owned and maintained separately from any adjoining land prior to April 2, 1972, as evidenced by deed recorded in the Stafford land records; or
- B. Was shown on a valid filed plan of subdivision, approved by the Commission after February 15, 1972;
- C. And provided that such lot in a) or b) above has its frontage on an improved public street and sanitary sewage disposal and a potable water supply can be assured without hazard to public health;
- D. And provided construction on such lot can meet all yard requirements.

Where two (2) or more nonconforming adjoining lots of record are in the same ownership, then such lots shall be combined to meet or more nearly meet, the lot area and lot frontage requirements of these regulations.

#### 3.3 Change of Lot Area and of Lot Dimensions:

No lot shall be so reduced in area or changed in dimensions that any required yard or other open space will be smaller than prescribed by the Zoning Regulations.

#### 3.4 One Main Building on a Lot:

No building containing a residential use shall be erected, altered or maintained in the rear of any building on the same lot, and no building shall be erected, enlarged or maintained in front of a residential building, except for agricultural purposes, as expressly limited in Section 3.16 hereafter.

#### 3.5 Yards and Open Space for Every Building

A. No lot shall be occupied for residential purposes unless it has the required frontage in the applicable zone on a public street.

- B. Except as specifically provided herein, no part of any yard, or other open space required about any building by the provisions of these regulations may be included as part of a yard, or other open space required for any other building.
- C. No building shall be erected within fifty (50) feet of any watercourse as shown on the official Stafford Flood Insurance Rate Maps. Measurements shall be made from the seasonal high water mark of any watercourse.

#### 3.6 Obstruction in Yards:

No structures or projections from structures shall be permitted in any required yard except as follows:

- A. Minor projections of structures, such as window or door frames and sills, belt courses, cornices, or other architectural features may project no more than one (1) foot into any required yard.
- B. Major projections of structures such as chimneys, bay windows not longer than 25% of the wall from that they project, eaves, roofs over doorways, hatchways, areaways, and fire escapes may project not more than four (4) feet into any required yard, provided that they shall not be closer than four (4) feet to any lot line.
- C. Fences, in any residence district, shall not exceed six (6) feet in height above finished grade. Minor variations of less than one (1) foot in elevation of the adjoining finished grade shall not affect the measurements of height of such fence. Fences shall be erected so that the finished side faces abutting property if not used for agricultural purposes.
- D. No structure shall be erected between the building line and the street line, except a wall or fence not over four (4) feet in height and not more than one-half (1/2) solid and except signs as specified in Section 6.3 of these Regulations.
- E. In any rear yard, only accessory buildings shall be permitted, provided that they shall not occupy more than 25% of the required rear yard.
- F. An attached solar greenhouse or other passive solar facilities on the southerly side of the main structure may project up to ten (10) feet into a required yard if added to any existing structure.

#### 3.7 Visibility at Intersections and Curves:

No obstruction, hedge, bush, tree, or other growth, or any fence or sign shall be erected, maintained, or planted that obstructs or interferes with a clear view of drivers of vehicles on a curve or at any street intersection and that may endanger the safety of those traveling upon such streets or highways. The minimum vision clearance shall require a height not exceeding three (3) feet above the street grade within twelve (12) feet of the intersecting street lines bordering corner lots.

#### 3.8 Building Lines:

No structure or accessory building shall be erected or altered, except in accordance with these Regulations, on any lot nearer to the street than the building line.

#### 3.9 Lot Frontage:

Each lot in future subdivisions of land shall have the frontage required in the applicable zone upon a public street.

#### 3.10 Lots on Narrow Streets:

In the case of lots facing streets less than fifty (50) feet wide, the front yard required by the applicable provisions of these Regulations shall be increased by one-half ( $\frac{1}{2}$ ) the difference between fifty (50) feet and the actual width of the street.

#### 3.11 Usable Lot Area:

The area of a lot that is considered usable lot area shall be the area of a lot that does not include floodplains, wetlands, easements, or naturally occurring slopes of over fifteen percent (15%). A building lot shall have a usable lot area of at least fifty percent (50%) of the minimum required lot size for the district in that such property lies. The land area comprising the fifty percent (50%) minimum shall be one contiguous piece. (Effective 11/15/03)

#### **3.12 Rear Lots (flag lots)** Effective July 27, 2005:

The establishment of rear lots in a subdivision or resubdivision shall be allowed and shall be limited to not more than twenty percent (20%) of the number of lots in such subdivision or resubdivision. Rear lots shall comply with all dimensional regulation standards required by Section 4.9 with the exception of lot frontage requirements. Lot frontage requirements shall be in accordance with this section. The following standards shall apply to all rear lots.

- A. The minimum lot area shall be measured within the body of the lot, exclusive of the area contained within the access strip running from the street. The body of the lot shall begin at the point the lot width meets the lot frontage requirement of Section 4.9.
- B. Street frontage shall be not less than twenty-five (25) feet and the width of the access strip shall average at least twenty-five (25) feet. No structure will be permitted within such strip.
- C. Yards shall adhere to requirements of the zoning district in that the building lies.
- D. A building line shall be established at least one hundred (100) feet (Effective 11/15/03) from the boundary of the body of the lot closest to the street on that the lot fronts.
- E. Type D screening shall be installed within the front yard setback area. (Effective 11/15/03)

#### 3.13 First Floor Elevation:

Where the first floor of any proposed principle structure is not, on average, eighteen (18) inches above the center line of the street or streets abutting a lot, the owner must consult with the First Selectman or his agent prior to construction.

#### 3.14 Exceptions to Height Requirements:

The height limitations of the Zoning Regulations shall not apply to chimneys, antennas, church spires, air conditioning equipment and enclosures, or other usual accessory features carried above the roof. Free-standing structures such as towers or tanks not intended for human habitation that exceed the height limits of these Regulations may be permitted by the Commission under a special use permit. Such structures shall be erected only to such height as is necessary to accomplish the purpose they are intended to serve.

#### 3.15 Required Street Frontages:

No building or zoning permit shall be issued for any structures or land use unless the lot for that the permit is sought has the required frontage on a public street. The minimum required street frontage shall be contiguous.

#### 3.16 Accessory Buildings:

- A. No accessory buildings or private garage shall be built on an untenanted lot unless it is part of a structure on a continuous foundation that is under construction and is to be completed within one (1) year or if the building is used for agricultural purposes.
- B. An accessory building attached to a main building, including attachment by means of a breezeway or a roofed passageway shall comply with the requirements of these Regulations applicable to the main building.
- C. Any accessory building observing the same yards as required for the main building may be erected to the same height limits as the main building.
- D. No accessory building on the same lot with a main residence building shall be used for residence purposes nor shall such building contain full bath or full kitchen facilities. (Effective 2-15-95)
- E. Accessory buildings may include private garages with space for not more than three (3) non-commercial motor vehicles on one (1) lot up to one (1) acre in size and for one (1) additional such vehicle for a lot in excess of one (1) acre. Not more than one (1) such space may be occupied by a commercial motor vehicle provided such vehicle be of not more than thirteen thousand (13,000) pounds gross vehicle weight. Spaces (bays) shall not exceed twelve feet by twenty-four feet (12' x 24') for calculation purposes.
- F. Accessory buildings to farming use are regulated in section 7.4. Greenhouses as accessory buildings are limited to one structure on a residential lot. More than one greenhouse as an accessory use may be allowed on a commercial farm, but such buildings shall be accessory and incidental to the main agricultural use of the property.
- G. A temporary roadside stand under one-hundred and fifty (150) square feet is permitted as of right. The temporary roadside stand must be set back from the street line a minimum of twenty-five (25) feet and meet all other setback regulations. The temporary roadside stand must be removed when the selling season is over. Off street parking for at least two (2) cars must be maintained.

- H. A roadside stand greater than one hundred and fifty (150) square feet and less than eight hundred (800) square feet may be permitted by the Zoning Enforcement Officer (ZEO). The roadside stand must be set back from the street line a minimum of twenty five (25) feet and meet all other setback regulations. A minimum of one (1) parking space per two hundred (200) square feet of selling space is required. Submission of a site plan showing parking spaces, type, and location of structure, ingress and egress, and proximity to other structures is required.
- I. A roadside market eight hundred (800) square feet or larger may be permitted by special permit from the Planning and Zoning Commission (PZC) and requiring a public hearing. The roadside market must be set back from the street line a minimum of twenty-five (25) feet and meet all other setback regulations. A minimum of one (1) parking space per two hundred (200) square feet of selling space is required. Buffers shall be a minimum of ten (10) feet wide and planted with plant materials reaching a minimum of six (6) feet high in five (5) years and produce a continuous visual barrier or erect a solid fence six (6) feet tall. Maximum size is twelve hundred (1200) square feet. The PZC may impose further conditions as it deems required to insure safety and welfare of the community.

#### 3.17 Location of Accessory Buildings:

- A. Accessory buildings that are not more than two hundred (200) square feet in size and not more than twelve (12) feet in height may be located:
  - 1. No closer to the front line than the front plane of the primary structure, or forty (40) feet from the front line of the lot and no closer than five (5) feet of the side or rear lines of the lot. Accessory buildings to farming use are regulated in section 7.4f.
  - 2. All accessory buildings over 200 square feet must comply with the standards of 4.9 and 5.7

#### 3.18 Removal and Replacement of Topsoil:

The removal or destruction of topsoil of more than one thousand (1,000) square feet on any lot shall not be permitted except in connection with construction, regrading, or landscaping work. After completion of such work, the topsoil shall be replaced and seeded according to accepted landscaping practices.

#### 3.19 Drainage:

No structure shall be used, erected, or expanded and no land shall be graded or hard-surfaced unless provisions have been made and approved by the Town of Stafford for the proper disposal of drainage water, particularly from parking areas and driveways, from areas contiguous to property lines and from low areas that tend to collect drainage water.

#### 3.20 Recreational Vehicles:

A recreational vehicle or camper or a boat and/or boat trailer may be stored on a lot occupied by the owner's residence, provided such unit is stored behind the building line and it shall be adequately screened from view from the street and from neighboring

properties. No unit may be occupied for sleeping, living, or for carrying on a business in any zone except as provided elsewhere in these regulations.

#### 3.21 Continuing Existing Nonconforming Uses:

Any use or building lawfully existing and any premises or building or any portion of any premises or building in use for a nonconforming use at the time of adoption of these regulations, or any amendments thereto, may be continued and any building so existing that was designed, arranged, intended for, or devoted to a nonconforming use may be reconstructed and structurally altered, and the nonconforming use therein changed subject to the following regulations:

- A. Except hereinafter specified, no nonconforming use may be changed except to a conforming use or, with the approval of the Zoning Board of Appeals, to another nonconforming use more consistent with the uses permitted in the zone in that the premises are located
- B. No nonconforming commercial or industrial use shall be extended or expanded except that existing non-conforming uses in AA zoning districts, which uses existed at the time of adoption of these regulations, or any amendments thereto, may be extended, expanded, or changed to another nonconforming use more consistent with the uses permitted in the zone in the premises are located, under a Special Use Permit, under the following conditions
  - 1.) The property on which the existing nonconforming use exists consists of a contiguous parcel of land consisting of not less than 5 acres when the nonconforming use began.
  - 2.) The Commission shall determine, based upon the use being expanded, the appropriate commercial district standards and requirements applicable for site improvements, including but not limited to parking, lot area coverage, and building setbacks.
  - 3.) It is required that, prior to the submission of an official application for commercial or industrial development approval, the applicant initiate a pre-application conference with the Commission and subsequently prepare and present a preliminary plan for informal consideration by the Commission.
- C. No nonconforming use that has been abandoned shall be thereafter resumed.

#### 3.22 Changes in Regulations

- A. Nothing in this section shall require any change in the plans, construction, or designated use of a building for that a building permit in accordance with existing regulations has been issued and on that the construction shall have been commenced prior to the adoption of these regulations, or of any amendments thereto, and that shall be completed within one (1) year of the adoption of the same.
- B. A building containing a permitted use that does not conform, because of amendments to these regulations, to building height limit, minimum lot area or width, percentage of lot coverage, required yards, or minimum floor area on the effective date of these

Regulations shall not be considered a nonconforming building within the meaning of Sec. 3.22.

#### 3.23 Residential Uses in Non-Residential Districts:

In any case where there is an existing residential use in a local business, highway or industrial, or water management zone that use was in existence at the time said zone was created, the residential dwelling may be enlarged or altered and accessory structures may be constructed, provided that such enlargement, alteration, or accessory structure meets the percentage of lot coverage and the parking and yard requirements of a AAA zone.

#### THIS PAGE INTENTIONALLY LEFT BLANK

## **Article IV**

#### **Residential Districts**

## **4.1 Explanation of Use Tables**

- A. No structure shall be used, erected, or expanded; and no land use shall be established or expanded except in accordance with the Zoning Regulations. The schedules contained in Section IV list permitted uses of lands and buildings, and the maximum height of buildings, the required yards, open space, area of lots, and other requirements for the various districts in the Zoning Regulations.
- B. Any use marked "SU" in the following tables is a permitted use subject to the issuance of a building and/or zoning permit and subject to Sec. 8.11(A) "Special Use Permit."
- C. Any use marked "SP" in the following tables is a permitted use subject to the issuance of a building and/or zoning permit and subject to Sec. 8.3 "Site Plan."
- D. Any use marked "P" in the following tables is a permitted use subject to the issuance of a building and/or zoning permit.
- E. Any use not marked is not a permitted use in the particular zoning district. Uses that are not permitted are prohibited unless the Commission determines that a proposed use is sufficiently similar to a listed use.
- F. Any use marked "A" in the following tables is a permitted use as of right without a permit.

#### 4.2 Schedule of Permitted Main Uses:

| Permitted Main (Principal) Uses   |     | identia | l Zonii | ng Distri | icts |
|---|-----|---------|---------|-----------|------|
|   | AAA | AA      | A       | В         | C    |
| 1. One-family (1) dwelling, one (1)/lot                                   | P   | P       | P       | P         | P    |
| 2. Two-family (2) dwelling  |     |         |         | P         | P    |
| 3. In-Law-Apartment   | P   | P       | P       | P         | P    |
| 4. Multi-family   |     |         |         | SU        | SU   |
| 5. Age Restricted Housing (Effective 6/11/08)                             | SU  | SU      | SU      |           |      |
| 6. Agriculture,   | A   | Α       | A       | A         | Α    |
| 7. Poultry  | A   | Α       | SU      | SU        | SU   |
| 8. Boarding camp  | SU  |         |         |           |      |
| 9. Government uses, cemeteries, places of worship, recreational club      | SU  | SU      | SU      | SU        | SU   |
| 10. Hospital, convalescent, or nursing home, home for elderly             | SU  | SU      | SU      | SU        | SU   |
| 11. Wood cutting/sawmill  | SP  | SP      |         |           |      |
| 12. Public, parochial, or private school or college                       | SU  | SU      | SU      |           |      |
| 13. Group day care home, child play care center, nursery school, adult    | A   | Α       | Α       | A         | Α    |
| day care  |     |         |         |           |      |
| 14. Bed and breakfast or rentals via online transient reservation systems | SU  | SU      | SU      | SU        | SU   |
| 15. Clubhouse, lodge, or other non-profit social, civic, or fraternal     | SU  | SU      | SU      |           |      |
| house   |     |         |         |           |      |
| 16. Camping areas established prior to 4-7-72                             | SU  | SU      |         |           |      |
| 17. Shooting ranges   | SU  | SU      |         |           |      |
| 18. Utility transmission lines and substations                            |     | P       | P       | P         | P    |
| 19. Elderly housing   |     | SU      |         | SU        | SU   |

| 20. Noncommercial kennel (Effective 12/19/2003)             | SU | SU |    |    |    |
|---|----|----|----|----|----|
| 21. Veterinary hospital (Effective 12/19/2010)              | SU | SU |    |    |    |
| 22. Horse and pony farms, riding academies, boarding stable | SU | SU |    |    |    |
| 23. Funeral home  |    |    | SU | SU | SU |
| 24. The sale of alcoholic beverages                         |    |    |    |    |    |
| a. Club permit or nonprofit permit                          | SU | SU | SU |    |    |
| 25. Parking (See Note A)                                    | SU | SU | SU | SU | SU |
| 26. Earth removal (Effective 4/25/2007)                     | SU | SU |    |    |    |

- A. One family dwellings in WM zones are subject to site plan requirements. Such site plan shall include wetland, flood plain, and stream encroachment delineations.
- B. All buildings or uses larger than thirty five thousand (35,000) square feet of gross leasable floor area require the issuance of a special use permit in accordance with the requirements of Section 6.31 of these regulations.
- C. A special use permit is required for all drive-through facilities.

NOTE A: All parking areas will comply with parking requirements of Sections 6.2.

NOTE: The Commission shall determine, upon request, all like uses not specifically listed and, in its discretion, may allow such use under a special use permit whether such use is a main use or accessory use.

#### 4.3 Schedule of Permitted Accessory Uses

| Permitted Accessory Uses  |      | Zoning Districts |    |   |   |  |
|---|------|------------------|----|---|---|--|
|   | AAA  | AA               | A  | В | C |  |
| A. Keeping of not more than two (2) non-transient roomers or boarders in any dwelling unit.   | P    | P                | P  |   |   |  |
| B. Private garage for motor vehicles, not to include more than one (1) commercial vehicle not more than thirteen thousand (13,000) GVW. | al P | P                | P  | P | P |  |
| C. Building used for storing, processing, and manufacture of agricultural product   | s. P | P                | P  | P | P |  |
| D. Building for storage and repair on a farm of any number of motor vehicles and equipment for operation of such farm.                  | l P  | P                |    |   |   |  |
| E. Swimming pools in accordance with all applicable state and local regulations.  | P    | P                | P  | P | P |  |
| F. Non-commercial greenhouses, storage sheds, boathouses.   | P    | P                | P  | P | P |  |
| G. Off-street parking and loading facilities.   | P    | P                | P  | P | P |  |
| H. Home occupation.   | P    | P                | P  | P | P |  |
| I. Signs as specified in Sec. 6.3   | P    | P                | P  | P | P |  |
| J. Agriculture  | A    | A                | Α  | Α | Α |  |
| K. Accessory uses incidental to a permitted use on the same lot subject to all other special conditions contained in these regulations. | er P | P                | P  | P | P |  |
| L. Amusement devices shall be limited to three (3) such devices at each location (Effective November 15, 2003)                          |      |                  |    |   |   |  |
| M. Lawn care business, Minor (see note A)   | SP   | SP               | SP |   |   |  |
| Lawn Care business, Major (see Note A)  | SU   | SU               |    |   |   |  |
| N. Heliport   | SU   |                  |    |   |   |  |

#### 4.4 Supplementary Regulations to the Tables of Permitted Uses

- A. **Regulations for utility lines and substations:** Utility transmission lines and substations are permitted uses in all zoning districts provided that:
  - 1. The location, construction, and right-of-way of any transmission line shall be such as to prevent hazard to the public and surrounding property.
  - 2. A utility substation shall be located on a lot at least ten thousand (10,000) square feet in area. There shall be suitable fencing to protect the public and landscaping to effectively screen the substation from surrounding property. Yard setbacks of the district, in which the substation is located, shall be met. In addition, there shall be adequate off-street parking areas.
  - 3. Minor structures, such as hydrants, telephone, or light poles, or similar equipment, shall not be subject to these regulations.

#### B. Uses expressly prohibited in all districts:

- 1. Distillation of bones, offal, or dead animals rendering or dumping.
- 2. Blast furnaces or smelting of copper, iron, lead, tin, and zinc.
- 3. Coal or petroleum distillation or derivation of by-products.
- 4. Manufacture of cement, lime, gypsum, or plaster of parts; or chlorine; or carbolic, hydrochloric, nitric, picric, or sulfuric acid.
- 5. Manufacture or storage of explosives.
- Fertilizer manufacture.
- 7. Fat rendering in the manufacture of tallow, grease, and oil.
- 8. Refining and recovery of products from fish, animal refuse, or offal.
- 9. Petroleum gas manufacture and storage by others than a public utility, except that the storage for distributing purposes and the distribution of liquefied petroleum gas may be permitted by the Zoning Board of Appeals, provided that the standards established by the National Board of Fire Underwriters, NBFU pamphlet 58 and 59, and applicable state laws, including revisions, are complied with. Nothing shall prevent the storage for use on the premises of liquefied petroleum gas when installed and used in accordance with applicable Connecticut State laws.
- 10. Production, storage, and/or placement of any nuclear waste, hazardous materials or hazardous waste, explosives, including pyrotechnic and incendiary devices (except pyrotechnic displays), and pollutants.
- 11. Junkyards of any kind or storage or baling of scrap not associated with a permitted manufacturing or business use.
- 12. Trailer park or mobile home park.

- 13. Commercial radio, television, microwave relay, or other communication towers whose height exceeds thirty-five (35) feet from the ground, except that permitted are:
  - a. Amateur radio station towers that do not exceed fifty (50) feet in height from the ground.
  - b. Private business radio towers solely for the use of the business or its employees provided that said towers are located at the business address of the applicant and provided that said towers do not exceed fifty (50) feet in height from the ground.
  - c. Other structures in excess of thirty-five (35) feet may be allowed by the Commission under the special permit procedure.
- 14. Apartments and/or living quarters in accessory buildings.
- 15. Similar uses to the above that are dangerous by reason of fire or explosion; or injurious, noxious, or detrimental to the neighborhood because of emission of dust, fumes, odor, smoke, wastes, noise, or vibrations; or because of other objectionable features.
- C. Chemical and petroleum products storage: Commercial underground tanks and storage systems of fiberglass construction shall be designed, constructed, and installed in accordance with the standards of the state building and fire codes, National Fire Prevention Association, American Society for Testing Materials, and "Connecticut Regulations for Control of Non-Residential Underground Storage and Handling of Oil and Petroleum Liquids."
- D. **Domestic fuel oil storage**: Domestic underground fuel oil storage tanks are not permitted under these Regulations. Fuel tanks shall be located in cellars/basements or in approved structures above ground.

#### E. Heliports:

- 1. One private, noncommercial heliport, including any equipment necessary to permit safe flights, shall be permitted as an accessory use by the owner of a lot of not less than five (5) acres in the AAA district with approval of a Special Use Permit.
- 2. Fueling of the helicopter on the property is prohibited.
- 3. The applicant shall have a favorable air space determination from the U.S. Department of Transportation, Federal Aviation Administration for the heliport.
- 4. Flights shall be limited to the hours between 6:00 a.m. and 10:00 p.m. with no more than two (2) flights outside of those hours per month and shall be limited to twelve (12) flights per week. The operator of a heliport shall keep a log listing the date, time, and purpose of all flights. The log shall be retained for twelve (12) months and subject to review by the Zoning Enforcement Officer.
- 5. Lights associated with the heliport shall be lit only for so long as is reasonably necessary to permit the pilot to locate the heliport and complete the flight.

- 6. These provisions shall not apply to:
  - a. Air medical helicopter (e.g., LIFE STAR) flights for the purpose of emergency patient care.
  - b. Flights by federal, state or local government officials or their authorized agents in the exercise of government responsibilities, including training.

# 4.5 Conversion for Related Family Unit ("in-law apartment"):

Conversion of single-family homes to provide for one in-law apartment unit for immediate family may be permitted by the Zoning Enforcement Officer subject to reasonable safeguards and the following standards:

- A. The in-law apartment unit is to be occupied by a parent, grandparent, aunt, uncle, brother, sister, brother-in-law, sister-in-law, child, or child-in-law of the owner-occupant of the principal dwelling or a care giver as certified by a licensed physician.
- B. The in-law apartment shall be limited to a maximum of three (3) rooms exclusive of bathrooms, hallways, and closets. The in-law apartment shall also be limited to a maximum floor area of seven hundred and fifty (750) square feet. (as defined on the drawing by providing square feet of each room).
- C. The remainder of the structure shall contain the minimum livable floor area that is required for a single-family residence in that zone.
- D. The in-law apartment unit shall be in the same building as the principal dwelling unit.
- E. There shall be no separate utility service entrance for the in-law apartment unit.
- F. No structural alteration made to the exterior of the building shall detract from its single-family characteristics.
- G. No stairs above the first floor shall be added outside the existing exterior walls except at the rear wall of the building; and on a corner lot, any new stairs shall be added within the existing walls of the building or added in the form of a wing that will not detract from the building's single-family characteristics.
- H. An approval notice from the Stafford Health District as to the adequacy of on-site sewage disposal and water supply if utilized.
- I. One additional off-street parking space shall be provided for use by the occupant(s) of said in-law apartment.
- J. The owner of the accessory apartment property must file a deed restriction on the land records requiring the unit, if rented, be rented at or below prices that would qualify the apartment as "affordable housing", as defined in CGS 8-39G.
- K. Failure to meet any of the above requirements may be cause for revocation of the permit.

#### 4.6 Home Occupations and Professional Offices:

Traditional professional offices and home occupation that are clearly accessory to the primary use of a building as a single-family dwelling are permitted in all single-family residential districts. The following standards shall apply:

A. A site plan shall be submitted for approval by the Zoning Enforcement Officer.

- B. Office use of home occupation is restricted to the resident thereof.
- C. Professional offices may have up to two (2) non-resident employees.
- D. Adequate off-street screened parking shall be provided on the premises.
- E. The Z.E.O., at his/her discretion, may waive the requirement for a site plan, or he/she may require a public hearing as a special use if it is determined that a proposed use might have an effect on abutters.

#### 4.7 Conversion for Bed and Breakfast or rentals via online transient reservation systems:

Conversion of a single-family home to provide for a bed and breakfast facility when used in conjunction with primary residential use may be allowed by the Commission as a special permit use subject to reasonable safeguards and the following standards:

#### A. A bed and breakfast shall:

- 1. Not contain more than six (6) guest rooms;
- 2. Provide meals only to overnight guests;
- 3. Not change the residential character of the dwelling;
- 4. Have appropriate sewage disposal facilities to meet Health Code requirements or be connected to a public sewer.

#### B. Off-street parking shall:

- 1. Consist of one (1) space per rental room in addition to two (2) spaces for the dwelling;
- 2. Not be located between the street and the building line except for the dwelling;
- 3. Have lighting adequate for safety and security without creating a nuisance to abutters.

#### **4.8 Open Space Districts (OS)**

- A. The purpose of the open space districts established in these Regulations is designed to further promote the health, safety, and the general welfare of the residents of Stafford. These general goals include the following specific purposes:
  - To provide sufficient public space in appropriate locations for the retention of open spaces to: maintain aesthetic and scenic value; conserve and protect natural resources; serve as a natural buffer zone between incompatible land uses; and serve as a "land bank" where open land remains as a reserve for future development.
  - 2. To provide sufficient space in appropriate locations to meet the needs for active, natural, and passive recreational areas including: playgrounds and play fields and neighborhood parks.

- 3. To provide sufficient space in appropriate locations to meet needs for the expansion and addition to the community's educational plant.
- B. Open space districts may remain vacant and undeveloped or may be developed so as to provide natural, active, or passive recreation or educational facilities to meet the needs of the community. All plans and specifications for development within designated open space areas shall be subject to the review and findings of the Planning and Zoning Commission and conform to the land use objectives in the Plan of Development. Permits for the development of open space properties may only be granted by the Commission after lawfully advertised public hearing.

#### 4.9 Dimensional Standards for Residence Districts

| Minimum Dimensions              | AAA    | AA        | AA*    | A      | B****  | C****  | WM     |  |  |  |  |
|---------------------------------|--------|-----------|--------|--------|--------|--------|--------|--|--|--|--|
| Lot area – square feet          | 88,000 | 44,000*** | 30,000 | 15,000 | 40,000 | 44,000 | 88,000 |  |  |  |  |
| Lot frontage – feet             | 200    | 175       | 150    | 100    | 250    | 100    | 200    |  |  |  |  |
| Lot depth – feet                | 250    | 200       | 175    | 120    | 150    | 150    | 250    |  |  |  |  |
| Front yard – feet               | 50     | 40        | 50     | 25     | 50     | 15     | 50     |  |  |  |  |
| Side yard – feet                |        |           |        |        |        |        |        |  |  |  |  |
| Main building                   | 30     | 20        | 20     | 20     | 30     | 10     | 30     |  |  |  |  |
| Accessory building** (Effective | 30     | 20        | 20     | 12     | 30     | 10     | 30     |  |  |  |  |
| November 15, 2003)              |        |           |        |        |        |        |        |  |  |  |  |
| Parking area                    | 10     | 5         | 5      | 2      | 5      | 5      | 10     |  |  |  |  |
| Rear yard – feet                |        |           |        |        |        |        |        |  |  |  |  |
| Main building                   | 50     | 35        | 35     | 35     | 50     | 25     | 50     |  |  |  |  |
| Accessory building** (Effective | 50     | 35        | 35     | 35     | 50     | 25     | 50     |  |  |  |  |
| November 15, 2003)              |        |           |        |        |        |        |        |  |  |  |  |
| Parking area                    | 10     | 5         | 5      | 2      | 5      | 5      | 10     |  |  |  |  |
| Maximum height                  |        |           |        |        |        |        |        |  |  |  |  |
| Main building – stories         | 21/2   | 21/2      | 21/2   | 21/2   | 21/2   | 3      | 21/2   |  |  |  |  |
| Main building – feet            | 30     | 30        | 30     | 30     | 35     | 35     | 30     |  |  |  |  |
| Accessory building – stories**  | 30     | 30        | 30     | 30     | 35     | 35     | 30     |  |  |  |  |
| (Effective November 15, 2003)   |        |           |        |        |        |        |        |  |  |  |  |
| Lot area/dwelling unit          |        |           |        |        |        |        |        |  |  |  |  |
| Square feet                     | 88,000 | 30,000    | 30,000 | 15,000 | 11,000 | 7,330  | 88,000 |  |  |  |  |
| Lot coverage – all buildings    |        |           |        |        |        |        |        |  |  |  |  |
| Percentage                      | 10     | 15        | 20     | 20     | 20     | 20     | 10     |  |  |  |  |

<sup>\*</sup> Denotes AA District within the former Borough of Stafford Springs.

#### 4.10 Apartment Buildings:

Apartments shall be constructed with the following minimum livable floor areas:

<sup>\*\*</sup> Accessory buildings that are not more than two hundred (200) square feet in size and not more than twelve (12) feet in height shall comply with Section 3.17.

<sup>\*\*\*</sup> Minimum of 30,000 square feet needed per unit for two-family (2) dwelling.

<sup>\*\*\*\*</sup> Multi-family dwellings shall not exceed four (4) units/acre in Zone B and six (6) units/acre in Zone C.

| Efficiency  | 370 square feet |
|-------------|-----------------|
| One-bedroom | 550 square feet |
| Two-bedroom |                 |

#### 4.11 Standards for Multi-Family Use:

The Commission, as a special use, may permit the development of multi-family housing for a minimum parcel size of two (2) acres in Residence B or Residence C Zones. Density shall not exceed four (4) dwelling units/acre in Zone B and six (6) dwelling units/acre in Zone C. (Effective 2-15-95)

### 4.12 Multi-Family Permitted Uses:

- A. Multiple family buildings, including garden apartments, townhouses, apartments, group buildings, or a combination thereof, for rental or for sale as individual dwelling units.
- B. Occupancy shall not exceed one (1) family per unit nor exceed the limits established by the Commission.

Accessory uses incidental to permitted uses for the use of owners or tenants and their guests, including garages, carports, off-street parking areas, laundry facilities are permitted. Community and recreational uses including tennis, swimming, and community, or club houses shall be limited to residents and guests, unless such facilities are operated as a non-profit club.

#### 4.13 Homeowners' Association:

A homeowners' association must be legally established if units are to be sold.

### 4.14 Criteria for Building and Sites

- A. Buildings shall be located to provide for the amenities of apartment living to assure the maximum of light and ventilation, to control the impact of traffic and parking on the site, and to integrate the development with the neighborhood.
- B. The scale of buildings shall be compatible with the residential scale of buildings within the town. No building facade or roof visible from public or private streets shall have a continuous surface exceeding fifty (50) feet in length between horizontal offsets or a projecting roof gable.
- C. Maximum lot coverage for buildings shall not exceed twenty-five percent (25%) of the developable land area and dwelling density shall not exceed four (4) units per acre in Residence B and six (6) units per acre in Residence C districts.
- D. The site shall be served by public sewer and water.
- E. In development of ten (10), or more units, a minimum of thirty-three percent (33%) shall be efficiency or one-bedroom units and not more than twenty percent (20%) may have three (3) or more bedrooms.

- F. A portion of a wall without openings in a principal building shall not be less than fifteen (15) feet from a portion of a parallel or opposing wall without openings in another principal building on the same lot or six (6) feet to a wall without openings in an accessory building, provided that any openings to a garage shall not be less than ten (10) feet to the nearest window in a principal building.
- G. A portion of a wall with openings in a principal building shall be not less than sixty (60) feet to any portion of a parallel or opposing wall with openings in another principal building on the same lot or fifty (50) feet minus the required side yard on the adjoining lot opposing such wall or forty (40) feet to any portion of a parallel, or opposing wall without openings in another principal building, or accessory building on the same lot or ten (10) feet to any portion of a wall without openings in an accessory building; the highest part of the roof of that is below the lowest window sills of a habitable room in such principal building facing such accessory building.
- H. Courts enclosed by buildings on all sides shall be prohibited. In courts formed by two L-shaped buildings, the distance between the nearest corners of the buildings shall be not less than twenty (20) feet measured in a straight line between such corners.
- I. The finished grades at front, side, and street elevations of buildings shall be not more than twenty-four (24) inches below the first story level, except that the Commission may modify this requirement in the case of natural slopes in existing ground levels.

## **4.15 Public Improvements:**

- A. Any required public improvements, including fire hydrants, sanitary sewers, and storm drainage, shall conform to the applicable sections of the Subdivision Regulations and the Town of Stafford standards.
- B. The owner(s) shall provide and maintain all private road and parking areas, as may be required. All main roads shall conform to the Town of Stafford specifications except that the minimum width may be fifteen (15) feet for one-way and twenty-four (24) feet for two-way traffic and a performance bond may be required initially.
- C. The owner(s) shall provide and maintain all private street lights and sidewalks, hydrants, signs, trash storage areas, and recreation areas.

## 4.16 Maintenance and Operation

- A. The owner(s) shall be responsible for the proper operation and maintenance of the grounds, private roads and parking areas, landscape buffers, and buildings. Lack of proper operation or maintenance shall constitute a violation of these Regulations.
- B. When an individual unit is sold, each deed of sale shall include suitable provisions requiring proportionate ownership of and responsibility for any community property owned in common with others. All such conditions and agreements shall be in a form satisfactory to the Town of Stafford and shall be recorded in the Stafford Town Records.

### 4.17 Parking, Landscaping, and Buffers

- A. Off-street parking spaces or garages shall be provided in a ratio of two (2) spaces for every dwelling unit with the exception of elderly housing (ratio to be determined for each project by the Commission). All parking and loading areas shall be located in side or rear yards only and shall be screened from the street and abutting residential areas by Type "C" screening.
- B. Parking areas shall not be located within twenty (20) feet of a building wall and shall be separated by landscaped areas that may contain appropriate pedestrian walkways.
- C. All yards shall be accessible for firefighting purposes and shall include Type "A" screening.
- D. Where appropriate, in the judgment of the Commission, suitable walls or fencing may modify or waive the landscape buffer requirements.
- E. Where two similar uses or accessory uses abut a common yard, the Commission may modify or waive the landscape buffer requirements.
- F. A parking area containing spaces for more than twenty-four (24) cars shall be designed to include not less than five percent (5%) of the gross area for landscape treatment. Landscaped area\* shall be not less than three (3) feet wide. In general, there shall be not more than ten (10) contiguous car spaces without separation by a landscaped area, suitably planted.

\*Areas: planted trees, shrubs, and/or flowering plants

### 4.18 Filing Requirements:

An approved site plan shall be filed with the Commission and an approved plan not so filed within ninety (90) days shall become null and void. For filing purposes, said plan shall be submitted on *Mylar*, or similar transparent material accompanied by seven (7) paper prints. Such filed plan may be declared null and void by the Commission unless construction of buildings is in progress and not less than fifty percent (50%) of building foundations are completed within one (1) year of the date of approval of such plan, unless an extension of time is granted by the Commission.

# Article V Commercial Districts

## 5.1 Explanation of Tables

- A. No structure shall be used, erected, or expanded; and no land use shall be established or expanded except in accordance with the Zoning Regulations. The schedules contained in Section IV list permitted uses of lands and buildings, and the maximum height of buildings, the required yards, open space, area of lots, and other requirements for the various districts in the Zoning Regulations.
- B. Any use marked "SU" in the following tables is a permitted use subject to the issuance of a building and/or zoning permit and subject to Sec. 8.11(A) "Special Use Permit."
- C. Any use marked "SP" in the following tables is a permitted use subject to the issuance of a building and/or zoning permit and subject to Sec. 8.3 "Site Plan."
- D. Any use marked "P" in the following tables is a permitted use subject to the issuance of a building and/or zoning permit.
- E. Any use not marked is not a permitted use in the particular zoning district. Uses that are not permitted are prohibited unless the Commission determines that a proposed use is sufficiently similar to a listed use.
- F. Any use marked "A" in the following tables is a permitted use as of right without a permit.

### 5.2 Schedule of Permitted Main Uses:

| Permitted Main (Principal) Uses   |     |    | Zoning | g Distr | icts |    |    |
|---|-----|----|--------|---------|------|----|----|
|   | WM  | OS | LB     | HB      | CB   | IN | HI |
| 1. One-family (1) dwelling, one (1)/lot                                     | SU* |    | P      |         |      |    |    |
| 2. Apartments above first floor   |     |    | SP     |         | SP   |    |    |
| 3. Agriculture  | Α   | A  | A      | Α       | P    | P  | P  |
| 4. forestry   | P   | P  | P      | P       | P    | P  | P  |
| 5. Poultry  | Α   | A  | A      | Α       | A    | Α  | Α  |
| 6. Government uses, cemeteries, places of worship, recreational club        | SU  | SU | SU     | SU      | SU   | SU | SU |
| 7. Wood cutting/sawmill   | SP  |    |        |         |      | SP | SP |
| 8. Group day care home, child play care center, nursery school, adult       |     |    | SU     | SU      |      | SU | SU |
| day care  |     |    |        |         |      |    |    |
| 9. Bed and breakfast  |     |    | SP     | SP      |      |    |    |
| 10. Clubhouse, lodge, or other non-profit social, civic, or fraternal house |     |    | SP     | SP      |      |    |    |
| 11. Utility transmission lines and substations                              | P   |    | P      | P       | P    | P  | P  |
| 12. Commercial Kennel   |     |    |        | SU      |      | SU | SU |
| 13. Noncommercial kennel (Effective December 19, 2003)                      |     |    | SP     | SP      |      |    |    |
| 14. Veterinary hospital (Effective December 19, 2010)                       |     |    | SU     | SU      |      | SU | SU |
| 15. Rental Car Business   |     |    |        | SP      |      | SP | SP |
| 16. Funeral home  |     |    | SP     | SP      |      |    | SP |
| 17. Retail, personal services   |     |    | SP     | SP      | SP   |    |    |
| 18. Restaurant  |     |    | SP     | SP      | SP   |    | SP |

| Permitted Main (Principal) Uses  | Zoning Districts |    |    |    |    |    |    |
|--|------------------|----|----|----|----|----|----|
| `  | WM               | OS | LB | НВ | СВ | IN | НІ |
| 19. The sale of alcoholic beverages  |                  |    |    |    |    |    |    |
| a. Café permit   |                  |    | SU | SU | SU |    | SU |
| b. Tavern permit   |                  |    | SU | SU | SU |    | SU |
| c. Club permit or nonprofit permit, brewpub  |                  |    | SU | SU | SU |    | SU |
| d. Hotel permit or hotel permit for beer only  |                  |    | SU | SU | SU |    | SU |
| e. Package store permit  |                  |    | SU | SU | SU |    | SU |
| f. Restaurant permit, restaurant permit for beer only, restaurant permit for wine and beer only, or restaurant permit for catering establishment |                  |    | SU | SU | SU |    | SU |
| 20. Parking *  |                  |    |    | SU |    | SP | SP |
| 21. Business and professional offices, banks   |                  |    | SP | SP | SP | SP | SP |
| 22. Auto sales, auto repair, service station, gasoline sales   |                  |    |    | SP |    | SP | SP |
| 23. Trade school, business school  |                  |    | SP | SP | SP | SP | SP |
| 24. Assembling, processing, and packaging of goods for sale (25% of floor)   |                  |    | SP | SP | SP | SP | SP |
| 25. Grooming facility  |                  |    | SP | SP | SP | SP | SP |
| 26. Drive-in, open air sales, commercial recreation, amusement arcade, café, indoor entertainment facility.                                      |                  |    | SU | SU | SU |    |    |
| 27. Bus station, hotel, motel, inn   |                  |    | SU | SU | SU | SU | SU |
| 28. Wholesale, storage, freight terminal, commercial processing, Brewery   |                  |    |    |    |    | SP | SP |
| 29. Manufacturing, processing, assembly  |                  |    |    | SP |    | SP | SP |
| 30. Outdoor storage, fuel storage, contractor's yard   |                  |    |    |    |    | SP | SP |
| 31. Salvage operations contained within a building   |                  |    |    |    |    | SU | SU |
| 32. Lawn care business   |                  |    |    | P  |    | P  |    |
| 33. Retail use limited to ten (10)% of floor area  |                  |    |    |    |    | P  | P  |
| 34. Retail uses having a gross floor area of more than thirty thousand (30,000) square feet  |                  |    | SU | SU | SU |    | SU |
| 35. Earth removal (Effective 4/25/2007)  |                  |    | SU | SU | SU | SU | SU |
| 36. Medical Marijuana Producer   |                  |    |    | SU | SU | SU | SU |
| 37. Medical Marijuana Dispensary   |                  |    | SP | SP | SP |    | SU |
| 38. Recreational Cannabis Cultivator   |                  |    |    |    |    | SU | SU |
| 39. Recreational Cannabis Retailer   |                  |    |    | SU |    |    | SU |
| 40. Recreational Cannabis Packager/Manufacturer  |                  |    | SU | SU | SU | SU | SU |
| 41. Roadside Selling pertaining to food trucks/trailers  |                  |    |    | SU |    |    |    |

- A. One family dwellings in WM zones are subject to site plan requirements. Such site plan shall include wetland, flood plain, and stream encroachment delineations.
- B. All buildings, or uses larger than thirty five thousand (35,000) square feet of gross leasable floor area require the issuance of a special use permit in accordance with the requirements of Section 8.11(A) of these regulations.
- C. A special use permit is required for all drive through facilities.
- \* All parking areas will comply with parking requirements of Section 6.2.
- \*\* The Commission shall determine, upon request, all like uses not specifically listed and, in its discretion, may allow such use under a special use permit whether such use is a main use or accessory use.

## 5.3 Schedule of Permitted Accessory Uses

|    | Permitted Accessory Uses   |    |    | Zoning Districts |    |    |    |    |
|----|--|----|----|------------------|----|----|----|----|
|    | ·  | WM | GS | LB               | HB | CB | IN | HI |
| A. | Keeping of not more than two (2) non-transient roomers or boarders   |    |    | Α                | A  |    |    |    |
|    | in any dwelling unit.  |    |    |                  |    |    |    |    |
| B. | Private garage for motor vehicles, not to include more than one (1)  | P  | P  | P                | P  | P  | P  | P  |
|    | commercial vehicle over thirteen thousand (13,000) GVW.              |    |    |                  |    |    |    |    |
| C. | Building used for storing, processing, and manufacture of            | P  | P  | P                | P  | P  | P  | P  |
|    | agricultural products.   |    |    |                  |    |    |    |    |
| D. | Building for storage and repair on a farm of any number of motor     | P  | P  | P                | P  | P  | P  | P  |
|    | vehicles and equipment for operation of such farm.                   |    |    |                  |    |    |    |    |
| E. | Swimming pools in accordance with all applicable state and local     | P  |    |                  | P  |    |    |    |
|    | regulations.   |    |    |                  |    |    |    |    |
| F. | Off-street parking and loading facilities.                           | P  | P  | P                | P  | P  | P  | P  |
| G. | Signs as specified in Sec. 6.3.                                      | P  | P  | P                | P  | P  | P  | р  |
| H. | Agriculture  | A  | Α  | Α                | A  | P  | P  | Α  |
| I. | Accessory uses incidental to a permitted use on the same lot subject | P  | P  | P                | P  | P  | P  | P  |
|    | to all other special conditions contained in these regulations.      |    |    |                  |    |    |    |    |
| J. | Amusement devices shall be limited to three (3) such devices at each |    |    | P                | P  | P  |    |    |
|    | location (Effective 11/15/2003)                                      |    |    |                  |    |    |    |    |

## 5.4 Supplementary Regulations to the Tables of Permitted Uses

- A. **Regulations for utility lines and substations:** Utility transmission lines and substations are permitted uses in all zoning districts, provided that:
  - 1. The location, construction, and right-of-way of any transmission line shall be such as to prevent hazard to the public and surrounding property.
  - 2. A utility substation shall be located on a lot at least ten thousand (10,000) square feet in area. There shall be suitable fencing to protect the public and landscaping to effectively screen the substation from surrounding property. Yard setbacks of the district, in which the substation is located, shall be met. In addition, there shall be adequate off-street parking areas.
  - 3. Minor structures, such as hydrants, telephone or light poles or similar equipment, shall not be subject to these regulations.

#### B. Uses expressly prohibited in all districts:

- 1. Distillation of bones, offal, or dead animal rendering, or dumping.
- 2. Blast furnaces or smelting of copper, iron, lead, tin, and zinc.
- 3. Coal or petroleum distillation or derivation of by-products.
- 4. Manufacture of cement, line, gypsum, or plaster of parts; or chlorine; or carbolic, hydrochloric, nitric, picric, or sulfuric acid.
- 5. Manufacture or storage of explosives.

- 6. Fertilizer manufacture.
- 7. Fat rendering in the manufacture of tallow, grease, and oil.
- 8. Refining and recovery of products from fish, animal refuse, or offal.
- 9. Petroleum gas manufacture and storage by others than a public utility, except that the storage for distributing purposes and the distribution of liquefied petroleum gas may be permitted by the Zoning Board of Appeals, provided that the standards established by the National Board of Fire Underwriters, NBFU pamphlet 58 and 59, and applicable state laws, including revisions, are complied with. Nothing shall prevent the storage for use on the premises of liquefied petroleum gas when installed and used in accordance with applicable Connecticut State laws.
- 10. Production, storage, and/or placement of any nuclear waste, hazardous materials, or hazardous waste, explosives, including pyrotechnic and incendiary devices (except pyrotechnic displays), and pollutants.
- 11. Junkyards of any kind, or storage, or baling of scrap not associated with a permitted manufacturing, or business use.
- 12. Trailer park, or mobile home park.
- 13. Commercial radio, television, microwave relay, or other communication towers whose height exceeds thirty-five (35) feet from the ground, except that permitted are:
  - d. Amateur radio station towers that do not exceed fifty (50) feet in height from the ground.
  - e. Private business radio towers solely for the use of the business, or its employees, provided that said towers are located at the business address of the applicant and provided that said towers do not exceed fifty (50) feet in height from the ground.
  - f. Other structures in excess of thirty-five (35) feet may be allowed by the Commission under the special permit procedure.
- 14. Apartments and/or living quarters in accessory buildings.
- 15. Similar uses to the above that are dangerous by reason of fire or explosion; or injurious, noxious, or detrimental to the neighborhood because of emission of dust, fumes, odor, smoke, wastes, noise, vibrations; or because of other objectionable features.
- C. Chemical and petroleum products storage: Commercial underground tanks and storage systems, of fiberglass construction, shall be designed, constructed, and installed in accordance with the standards of the state building and fire codes, National Fire Prevention Association, American Society for Testing Materials, and "Connecticut Regulations for Control of Non-Residential Underground Storage and Handling of Oil and Petroleum Liquids."

D. **Domestic fuel oil storage**: Domestic underground fuel oil storage tanks are not permitted under these Regulations. Fuel tanks shall be located in cellars/basements or in approved structures above ground.

#### 5.5 Performance standards — Business and Industrial Uses:

The existing use of land or buildings shall not be changed or expanded except in conformance with the following criteria:

- A. The use shall be carried on in such a manner and with such precautions against fire and explosion hazards as to produce no serious exposure hazard to adjacent property and the storage of all flammable, or explosive materials shall be in a manner approved by the Fire Marshall of the town of Stafford.
- B. The use shall emit no offensive odors perceptible from any property line of the lot on that the operation is located and shall emit no noxious, toxic or corrosive fumes or gases.
- C. The use shall not exhaust, or waste into the air dust created by an industrial operation in excess of one cubic centimeter, or settled matter per cubic meter of air or produce heat, or glare perceptible from any property line of the lot on that the operation is located for any three (3) continuous minutes.
- D. Industrial and exterior lighting shall not produce glare on public highways or neighboring property or conflict with any traffic signals.
- E. No vibration will be transmitted outside the property where it originates.
- F. No radiation will be detectable outside of the building where it originates.
- G. No offensive nor untreated wastes will be discharged onto the land, or into any stream, water course, or storm sewer.
- H. Smoke, or other air contaminant shall not be discharged into the atmosphere from any single source of emission for a period, or periods aggregating more than three (3) minutes in any one (1) hour, that is as dark, or darker in shade than as designated as No. 2 on the Ringleman Chart, as published by the United States Bureau of Mines;, or that is of such opacity as to obscure an observer's view to a degree equal to or greater than smoke designated as No. 2 on the Ringleman Chart.

#### **5.6 Work/Live Floating Zone District** (Effective 12/15/04)

- A. **Purpose:** The intent of this section is to set forth regulations and standards for establishing and operating work/live occupancies as a primary commercial/industrial use, in that the proprietor would be allowed to reside as a secondary land use activity. The purposes of these provisions are:
  - 1. To provide for and make feasible the reuse of existing commercial or industrial buildings and related sites;

- To provide cost-efficient alternative work space that will provide an incentive for entrepreneurs, business owners, artists, artisans, and other individuals to continue to work in Stafford and contribute to the revitalization of older commercial buildings and sites;
- 3. To promote the preservation and reuse of commercial or industrial buildings that contribute to the historic character of the community;
- 4. To allow activities that are compatible with and will not compromise or interfere with existing and potential industrial or commercial uses in the districts were such work/live occupancies are established;
- 5. To ensure that work/live occupancies will function predominantly as work spaces with incidental residential accommodations that meet basic habitability requirements;
- 6. To ensure that the exterior design of structures converted to work/live use reflects the predominant industrial or commercial character of such buildings and will be compatible with adjacent commercial or industrial uses;
- 7. To ensure that, where there is adjacent residentially zoned land, changes to the exterior of structures converted to work/live are designed to make the commercial or industrial building being converted more compatible with the adjacent residential area.

## **B.** Applicability

- 1. Work/live occupancies are only allowed in existing buildings that are proven to be inadequate for other practical commercial, or business uses. Such proof may be established by lack of ability to sell or rent for other commercial or business uses.
- 2. Work/live occupancies are a floating district that may be allowed, at the discretion of the commission, in any zoning district. In determining the applicability of any proposal for a work/live occupancy, the commission will consider the purposes, standards, and requirements as described in this section 5.6(A). The commission will also consider the impact of such proposal to surrounding property values and the effect on public health, safety, and welfare.

#### C. Definitions

- 1. Living space that portion of a work/live occupancy that is used for residential purposes including, but not limited to, a sleeping area that includes up to two bedrooms, a food preparation area with reasonable work space, and a full bathroom including a lavatory, a toilet, and a shower or bathtub. Use of living space is limited to immediate family members of the tenant.
- 2. Work space- that portion of a work/live occupancy that is used only for commercial, or business purposes.

- 3. Work/live occupancy- a commercial, or industrial unit with incidental residential accommodations occupying one (1) or more rooms, or floors in a building primarily designed and used for industrial, or commercial occupancy and providing:
  - a. Adequate working space reserved for commercial, or industrial use and regularly used for such purpose by one (1) or more persons residing therein;
  - b. Living space as defined above and in accordance with the provisions of this section.
- 4. Adjacent- properties sharing a common property boundary, or are directly across a street right of way.
- 5. Immediate family spouse, mother, father, mother-in-law, father-in-law, daughter, son, sister, or brother.

#### D. Standards

- 1. Minimum floor area Each work/live occupancy shall include at least one thousand (1000) square feet of gross floor area.
- 2. Maximum living space Not more than thirty (30) percent, or five hundred (500) square feet, whichever is greater, of the work/live occupancy shall be reserved for living space. The remaining area must be reserved and designated as work space.
- 3. Separation required Each work/live occupancy in a building must be separated from other work/live occupancies, or other uses in the building. Access to each work/live occupancy shall be provided from common access areas, common halls, corridors directly from the exterior of the building.
- 4. Each building that provides work/live occupancies may contain additional business, or commercial uses allowed under this section that do not include living spaces.
- 5. Parking- Each work/live occupancy shall have two (2) parking spaces for up to one thousand (1,000) square feet of floor area plus one (1) additional space for every additional thousand (1,000) square feet, or portion thereof, of floor area. Calculation of parking spaces for commercial, or business uses that do not include living space shall be provided according to Section6.2I.
- E. Uses allowed in Work/Live Floating Zone District Retail, wholesale, personal services, business and professional offices, assembling, processing and packaging of goods for sale. All other uses expressly prohibited.
- F. Application process- Applicants applying for WL Floating Zone approvals must submit a site plan in accordance with section 8.3 and a zone change application.
- G. Said permit shall be reviewed annually by the ZEO to determine compliance with standards of this section.
- H. Failure to meet any of the above requirements may be a cause for revocation of the permit.

#### 5.7 Standards for Business and Industrial Districts

|                                | Local    | Highway  | Central  | Industrial | Highway    |
|--------------------------------|----------|----------|----------|------------|------------|
|                                | Business | Business | Business | IN         | Industrial |
|                                | LB       | HB       | CB       |            | HI         |
| Lot area (minimum) square feet | 7,500    | 20,000   | 4,000    | 40,000     | 30,000     |
| Lot width (minimum) feet       | 75       | 125      | 40*      | 150        | 125        |
| Lot depth (minimum) feet       | 100      | 150      | -        | ı          | 125        |
| Building Coverage ratio        | 25%      | 25%      | 100%     | 40%        | 25%        |
| (Maximum)                      |          |          |          |            |            |
| Floor area ratio               | 20%      | 25%      | 200%     | 40%        | 35%        |
| (Maximum)                      |          |          |          |            |            |
| Impervious coverage ratio      | 60%      | 60%      | 100%     | 60%        | 60%        |
| (maximum)                      |          |          |          |            |            |
| Stories (maximum)              | 2        | 2        | 3        | 3          | 3          |
| Height (maximum) feet          | 25       | 30       | 35       | 35         | 35         |
| Front yard (feet)              | 35       | 55       | **       | 60         | 55         |
| Side yard (minimum) feet ***   | 20       | 20       | *        | 20         | 20         |
| Rear yard (minimum) feet       | 20       | 20       | **       | 20         | 20         |

<sup>\*</sup> Minimum, or no less than the average of all lots in the particular block.

#### 5.8 Consolidation of Parcels:

Land under more than one ownership may be consolidated for development purposes, and buildings may be constructed and used individually under a site plan meeting all of the requirements of these Regulations, provided each separate owner shall have granted to each other owner in the consolidated parcel by deed, easement, or agreement recorded in the Town Clerk's Office. Such permanent access, egress, passage, and use of parking and loading and open spaces as may be required to assure integrated use and maintenance of the consolidated parcel, provided that nothing in this section shall be construed to permit the erection of a building or structure on an unaccepted street.

#### 5.9 Smaller Lots in Business Zones:

Smaller lots of record under separate ownership may be developed and used for permitted uses, provided:

- A. Such smaller lots are consolidated for development purposes into a parcel having the minimum frontage and area required in the schedule Section 5.7;
- B. Each owner shall grant to the other owners of a consolidated parcel mutual use of common facilities.
- C. Such consolidated parcel shall not adversely affect the development potential and future use of other lots in the area;

<sup>\*\*</sup> No less than the average of all lots in the particular block.

<sup>\*\*\*</sup> Where two, or more parcels under separate ownership are consolidated for development purposes, side yards are required only at the outer side lot lines of the consolidated parcel, provided the buildings are built on the common lot line.

D. Where, in the judgment of the Commission, existing conditions prevent the consolidation of lots into a plan of development containing the minimum lot frontage and area, the Commission may modify the minimum lot frontage and area requirements and may approve a plan of development that is, or can be, integrated with adjacent business development, and that is consistent with the purposes of these Regulations.

#### **5.10** Commercial and Industrial Development. (Effective April 1, 2006)

- A. **Applicability:** The requirements of this section (5.10) shall apply to all uses within the LB Local Business; HB Highway Business; CB Central Business, IN Industrial and HI Highway Industrial districts.
- B. **Purpose:** The purpose of this regulation is to provide for sound commercial and industrial development within the Town of Stafford to achieve the following objectives:
  - 1. Improve the tax base of the Town.
  - 2. Provide long-term employment opportunities for Town residents.
  - 3. Facilitate attractive commercial and industrial development to serve Town residents and is complementary to the scale and character of the community.

## C. Preliminary Site Development Plan:

It is recommended that, prior to the submission of an official application for commercial, or industrial development approval, the applicant initiate a preapplication conference with the Commission and subsequently prepare and present a preliminary plan for informal consideration by the Commission. The preparation of the preliminary plan is recommended to facilitate the general consideration of factors and problems affecting the development of the land before the applicant proceeds with the official application and the preparation of final maps, plans, and documents required for formal consideration by the Commission. The presentation of a preliminary plan will more readily and economically facilitate alterations and changes recommended by the Commission.

Neither the pre-application conference nor the formal consideration of the preliminary plan shall be deemed to constitute any portion of the official and formal procedure of applying for a Preliminary Site Development and Land Use Concept Plan approval. Neither the proponent nor the Commission shall be in any way bound by statements made in such informal discussions, their purpose being only to minimize delay, expense, and inconvenience to the public, the proponent, and the Commission upon future receipt, if any, of a formal application. Following any formal discussion, the Commission may suggest that the proposal, or certain aspects thereof, be referred to other municipal, State, or Federal agencies for review and comment, or Federal agencies for review and comment or may suggest that additional Information is, or will be required.

## D. Standards of Design:

#### 1. Area and Bulk

- a. For purposes of this section, the gross floor area of a retail store shall include gross floor area of all portions of the site outside of the exterior walls, or buildings used for the display, storage, or sale of any goods, wares, or merchandise, except that the gross floor area of a retail store shall not include exterior areas of not more than three thousand (3,000) square feet used for seasonal or temporary sales events.
- b. The gross floor area of adjacent stores shall be aggregated in cases where the stores (A) are engaged in the selling of similar, or related goods, wares, or merchandise and operate under common ownership, or management; (B) share check stands, a warehouse, or a distribution facility; or (C) otherwise operate as associated integrated, or co-operative business enterprises.

#### 2. Signs

- a. All signs shall conform to a theme or design that is compatible with the architecture of the buildings and site.
- b. All signs shall comply with requirements of Section 6.3 of the regulations.

#### 3. Architectural guidelines

The following architectural guidelines shall apply to all new construction, reconstruction, or addition within all commercial and industrial districts. The Commission may waive all, or sections of these requirements for building additions of less than two thousand five hundred (2,500) square feet of gross floor area.

#### 1. Facades

- a. No uninterrupted length of any façade shall exceed one hundred (100) feet and shall incorporate wall plane projections, or recesses.
- b. The ground floor façade of a retail, or similar use facing a public street, or pedestrian area shall incorporate display windows, awnings, or other such features to create visual interest in the site.
- c. Windows shall be recessed and should include visually prominent sills, shutters, or other such forms of framing.
- d. All building facades that are visible from a public street must be attractively designed with windows and other architectural elements so that no visible elevation looks like the back of a building.

#### 2. Roofs

- a. Variations in roof lines shall be used in the design of buildings.
- b. Rooftop equipment such as HVAC units shall be screened from public view with parapets featuring three-dimensional cornice treatments.
- c. The appearance of a flat roof shall be avoided.

- 3. Materials, colors, and detail features
  - a. Building facades shall include a repeating pattern including color, texture, and materials.
  - b. Predominant exterior building materials shall be high-quality materials, such as brick, wood, sandstone, and other native stone or tinted and textured concrete masonry units.
  - c. Façade colors shall be low reflectance, subtle, neutral, or earth tone colors. The use of high intensity colors, metallic colors, black, or fluorescent colors is prohibited.
  - d. Predominant exterior building materials should not include smooth-faced concrete block, tilt-up panels, or pre-fabricated steel panels.

## 5.11 Criteria for Approval of Special Use Permit

The Commission shall approve an application for a special use permit only if it finds that the proposed development meets the following criteria:

- A. The proposed development shall be of such location, size, and character that, in general, it will be in harmony with the appropriate and, orderly development of the area in that it is proposed to be situated and will not be detrimental to the, orderly development of adjacent properties.
- B. The location and size of proposed uses, the intensity of operations involved in Connecticut with such uses, the site layout, and their relationship to access streets shall be such that vehicular and pedestrian traffic generated by the use, or uses, shall not be detrimental to the character of the neighborhood.
- C. The proposed development will not hinder, or discourage the appropriate development and use of adjacent land and buildings, or impair the value thereof.
- D. The proposed uses permit the development of the site without the destruction of valuable natural assets, or pollution of lakes, streams, and other water bodies while providing the best possible design of structures and land uses compatible with the shape, size, and topographic and natural character of the site.
- E. The proposed development will be compatible with the existing and future character of the neighborhood in that the commercial development is to be located. Particular attention shall be paid to the type and density of adjacent residential development, the character and uniqueness of the natural resources of the neighborhood, the character and use of existing highway facilities, and the Plan of Conservation and Development.
- F. The location and character of buildings shall create a harmonious grouping and shall be compatible with surrounding structures.
- G. The proposed development shall not create traffic safety hazards, or congestion not consistent with the character of the community. Any improvements to public roads necessitated by the proposed development shall be the responsibility of the developer.

- H. The proposed development shall be of a superior quality site and architectural design that enhances the surrounding neighborhood environment.
- I. The public water supply facilities and facilities for the disposal of sanitary waste have sufficient capacity to accommodate the needs of the proposed development.
- J. The proposed development shall have a positive impact on the overall economy of the community and produce a fiscal benefit to the Town of Stafford.

## 5.12 Village Overlay District

The use of land, buildings and other structures within the Village Overlay District shall be established and conducted in conformity with the underlying zoning classification subject to the additional requirements of this Section. No lot shall be used, any new building constructed, or any existing building substantially improved until the Commission makes a determination that it complies with the procedures and informational requirements of this Section.

The intent of the Village Overlay District is to encourage the development and redevelopment of village areas within Stafford in such a way that protects, maintains, and enhances their unique character, as described in the 2010 Plan of Conservation and Development, while allowing for investment that maximizes the economic viability for future use. The Village Overlay District is intended to:

- Protect, maintain, and enhance the historic and village character of these areas.
- Provide flexibility in site design, maximizing development potential while ensuring aesthetically pleasing residential, commercial, and mixed use development.
- Create greater opportunities for traditional community living, working, housing, and recreation.
- Encourage a more efficient use of land and public services by promoting compact development in appropriate locations.

The goal of this regulation is to provide a process by which the Commission can encourage and provide incentives that ensure the protection and enhancement of the unique villages that define Stafford's character and charm.

## A. Purpose

The purpose of the Village Overlay district is to provide a process of design review in accordance with the provisions of the Village District Act, Section 8-2j of the Connecticut General Statutes, while increasing awareness of the importance of high-quality design. This review is not intended to discourage or prohibit the use of property; rather its purpose is to enhance the character, landscape and architecture quality of village areas consistent with the Plan of Conservation and Development.

## B. Interpretation and Applicability

The Village Overlay District is not to be interpreted as historic preservation regulations or as architecture design regulations that impose strict regulatory requirements upon properties. The Village Overlay District is to be interpreted as a design framework aimed at protecting, maintaining, and enhancing the character of the designated village centers. Therefore, the regulatory provisions of the Village Overlay District shall not be interpreted as rigid regulatory requirements that must be met, but as a framework and set of recommendations to elevate standards for design and quality of investment, while protecting, maintaining, and enhancing the character of Stafford's unique villages. Therefore, the design standards set forth in the Village District Overlay District establish an expectation for greater attention to design and high-quality design that will be considered by the Commission when reviewing applications that require site plan (Section 8.3) and/or special use permit (Section 8.11) approval. Recognizing that higher expectations and standards of design can result in greater costs for new construction, renovations, and rehabilitation, the Commission provides great flexibility in bulk and area dimensional requirements (Section 5.7) and parking requirements (Section 6.2) to better achieve the aim of protecting character and encouraging village style development, which create greater economic value and opportunity.

#### C. Activities Covered

- 1. Any new building construction and substantial building or site improvement which requires site plan (Section 8.3) and/or special use permit (Section 8.11) approval as required by these Regulations and which is located within the Village Overlay District is subject to the design and architectural requirements of, and eligible for the special provisions of, the Village Overlay District. Review of each new building or site and substantial building or site improvement shall cover the following to the satisfaction of the Commission:
- 2. The design and placement of the building in relationship to the adjacent sites, buildings, and public streets.
- 3. The maintenance of public views.
- 4. Building façade improvements, including awnings, facing a public street.
- 5. The height, size, massing, and proportion of new buildings and proposed modifications to existing buildings.
- 6. Site design, building placement, landscaping, parking layout, lighting, sidewalk location, and pedestrian accessibility.

## D. General Design Standards and Requirements

1. Any new building construction and substantial building or site improvement which requires site plan (Section 8.3) and/or special use permit (Section 8.11) approval as required by these Regulations is subject to the following design requirements. The burden is on the applicant (owner, developer, and/or tenants) to demonstrate their best attempts to satisfy these standards:

- 2. The proposed improvements protect, maintain, and enhance the unique and historic character of the village.
- 3. New construction and associated site design are sensitive to the village character and respect the historic architecture within the village.
- 4. Improvements and renovations of historic buildings are sensitive to and respect the historic architecture. Best attempts should be made to follow the recommendations of "The Secretary of the Interior's Standards for Rehabilitation and Guidelines for Rehabilitating Historic Buildings", as amended, and excluding interior renovations. In doing so, the applicant shall explain when said recommendations can't be followed, the reason or reasons for not following said recommendations, and why the alternatives proposed satisfy the intent of the Village Overlay District.
- 5. The height, size, massing, proportion, and design of new buildings and proposed modifications to existing buildings are sensitive to and respect the same characteristics of adjacent properties.
- 6. Site design, building placement, landscaping, parking layout, sidewalk location, and pedestrian accessibility enhance the village aesthetic and experience.

## E. Architectural Character and Design Guidance

- 1. The overall character of the proposed site design and architectural character of proposed structures shall protect property values in the village and the Town; preserve the existing historic character in terms of scale, density, architecture, and materials used in construction of all site features; protect the existing historic patterns of arrangement of structural and natural features, including circulation patterns; and, preserve public access to scenic views and vistas, and water courses. In the interest of general welfare and to stabilize and protect property values, applicants are encouraged to respect the New England village character of Stafford.
- 2. New England Architecture Styles:
  - 1. Colonial, Georgian, Federal, Jeffersonian, Greek Revival, Gothic Revival, Italianate, Shingle Style, and Bungalows. General description of these architectural styles can be found on the Historic New England website's Architectural Style Guide:
    - i. http://www.historicnewengland.org/preservation/your-older-or-historic-home/architectural-style-guide
    - ii. Other architectural style guides may be considered.

#### F. Encouraged Materials and Practices:

1. Materials, texture, and color used on the exterior walls and roof should emphasize the use of natural materials or should be those associated with traditional New England architecture. Preferred facade materials are brick, stone, and wood, including narrow-width siding, clapboards, wood shingles, or a reasonable equivalent. Metal, unfinished concrete, or concrete block, and asphalt siding is discouraged. Roofing

- materials, where visible, should be cedar shake, slate, copper, or a reasonable equivalent. Tarpaper, sheet metal, plastic, or fiberglass surfaces are discouraged.
- 2. Architectural details characteristic of the particular style and period proposed should be incorporated into the design for new construction and should relate harmoniously to adjacent buildings. Architectural details of a period need not be duplicated precisely, but they should suggest the extent, nature, and scale of the period.
- 3. Large structures should have well-articulated facades to reduce the appearance of significant bulk. Provision shall be made to coordinate site architecture with the character of the district. Rooflines should be varied to provide architectural interest.

#### **G.** Administration and Procedures

In accordance with the Village District Act, Section 8-2j(d) of the Connecticut General Statutes all applications for new construction and substantial reconstruction within the village district shall be subject to review and recommendation by the Village District Consultant. The Commission shall engage an architect or architectural firm, landscape architect, or planner who is a member of the American Institute of Certified Planners as the Village District Consultant, as needed, to provide such reviews. The fees for such Village District Consultant review shall be borne by the applicant in accordance with the Town Ordinance. The village district consultant shall review an application and report to the commission within 35 days of receipt of the application. Such report and recommendation shall be entered into the public hearing record and considered by the commission in making its decision. Failure of the village district consultant to report within the specified time shall not alter or delay any other time limit imposed by the regulations.

## **H. Pre-Application Consultation**

Applicants for site plan (Section 8.3) and/or special use permit (Section 8.11) approval within the Village District are required to submit a preliminary concept plan for review by staff. Applications involving new construction or substantial improvements, as determined by staff, may be referred to the Planning and Zoning Commission for a preapplication consultation. The preliminary concept plan shall include:

- 1. Preliminary Concept Plan Requirements:
  - a. The Concept Plan shall be submitted to the Planning and Zoning Office and shall be accompanied by plans and sufficient information so that staff and the Commission may informally review for general conformance with these Regulations.
  - b. The Concept Plan shall be considered only informational and advisory in nature and no development rights shall attach to the review or consideration of any Concept Plan.
  - c. Such review shall not be binding on the applicant or the Commission and any results or information obtained from it may not be appealed under any provision of the Connecticut General Statutes.

- d. The Concept Plan shall be placed on file in the Commission's office for continuing reference purposes for any subsequent application.
- e. No fee shall be required for the submission of a Concept Plan or preapplication meeting with the commission.

## I. Application

Application for site plan and/or special (cite site plan) use permit approval within the Village Overlay District shall include the following information:

- 1. **Description of materials:** Color and texture of major building materials, exterior wall elevations of all adjacent structures, drawn at an acceptable scale that allows for design review. When suitable, photos may be submitted, in lieu of drawings.
- 2. **Streetscape Facades:** Detailed drawing showing design, color, texture and type of materials of proposed building façade facing the public street.
- 3. **Signs:** Detailed drawing showing design, size, color, texture and type of materials. Lettering style, size and color for wall signs shall also be included as well as the illumination and light intensity. The uniformity of the signs, placement, and size consistency with adjacent signs shall also be presented.
- 4. **Utilities:** The location of all utilities serving the building such as transformers, HVAC units, gas and electric meters.
- 5. **Location:** The placement of new or substantially improved buildings in relationship to surrounding properties, public views from roadways and nearby public spaces.
- 6. **Lighting:** The design, location and illumination of buildings and light standard fixtures and their consistency with existing light styles.
- 7. **Compatibility:** A statement of the proposed new building's or substantially improved building's compatibility with the surrounding area or with the "The Secretary of the Interior's Standards for Rehabilitation and Guidelines for Rehabilitating Historic Buildings, as revised; and the standards set forth the Village Overlay District.
- **J. Bulk and Area Requirements Special Provisions:** Properties which satisfy the standards and requirements of the Village Overlay District are eligible to use the following incentives to facilitate their improvements subject to the approval of the Commission:

Table 1. Village District Bulk and Area Requirements

| Residential Zones – Minimum Dimensions | AA     | A      | B**    | C**    |
|--|--------|--------|--------|--------|
| Lot area – square feet                 | 44,000 | 15,000 | 40,000 | 44,000 |
| Lot frontage – feet                    | 125    | 75     | 175    | 75     |
| Lot depth – feet                       | 125    | 100    | 125    | 100    |

Table 1. Village District Bulk and Area Requirements

| Residential Zones – Minimum Dimensions | AA     | A      | B**    | C**   |
|--|--------|--------|--------|-------|
| Front yard – feet                      | 50     | 25     | 50     | 15    |
| Side yard – feet                       |        |        |        |       |
| Main building                          | 20     | 20     | 30     | 10    |
| Accessory building *                   | 20     | 12     | 30     | 10    |
| Parking area                           | 5      | 2      | 5      | 5     |
| Rear yard – Feet                       |        |        |        |       |
| Main building                          | 35     | 35     | 50     | 25    |
| Accessory building *                   | 35     | 35     | 50     | 25    |
| Parking area                           | 5      | 2      | 5      | 5     |
| Maximum height                         |        |        |        |       |
| Main building – stories                | 21/2   | 21/2   | 21/2   | 3     |
| Main building – feet                   | 30     | 30     | 35     | 35    |
| Accessory building – stories*          | 30     | 30     | 35     | 35    |
| Lot area/dwelling unit                 |        |        |        |       |
| Square feet                            | 30,000 | 15,000 | 11,000 | 7,330 |
| Lot coverage – all buildings           |        |        |        |       |
| Percentage                             | 20     | 20     | 20     | 20    |

<sup>\*</sup> Accessory buildings that are not more than two hundred (200) square feet in size and not more than twelve (12) feet in height shall comply with Section 3.17.

Table 2. Village District Bulk and Area Requirements

| Commercial Zones – Minimum Dimensions | LB   | НВ     | СВ    | IN     |
|---------------------------------------|------|--------|-------|--------|
| Lot area (minimum) square feet        | 7500 | 20,000 | 4,000 | 40,000 |
| Lot width (minimum) feet              | 50   | 100    | 40*   | 125    |
| Lot depth (minimum) feet              | 75   | 125    | 0     | 0      |
| Lot coverage (maximum)                | 35   | 35     | **    | 50     |
| Stories (maximum)                     | 3    | 3      | 4     | 4      |
| Height (maximum) feet                 | 35   | 35     | 45    | 45     |
| Front yard (feet)                     | 20   | 30     | **    | 50     |
| Side yard (minimum) feet ***          | 10   | 10     | *     | 15     |
| Rear yard (minimum) feet              | 15   | 15     | **    | 15     |

<sup>\*</sup> Minimum or no less than the average of all lots in the particular block

## **K.** Parking – Special Provision:

<sup>\*\*</sup> Multi-family dwellings shall not exceed four (4) units/acre in Zone B and six (6) units/acre in Zone C.

<sup>\*\*</sup> No less than the average of all lots in the particular block

<sup>\*\*\*</sup> Where two or more parcels under separate ownership are consolidated for development purposes, side yards are required only at the outer side lot lines of the consolidated parcel, provided the buildings are built on the common lot line.

The parking ratio in Village Overlay Zone shall not be applied to changes in use to an existing site and building, unless the gross building area is increased by more than (500) five hundred square feet.

#### L. Application Decisions:

If the Commission grants or denies an application, it shall state upon the record the reasons for its decision. If the Commission denies an application, the reason for the denial shall cite the specific regulations under which the application was denied. Notice of the decision shall be published in a newspaper having a substantial circulation in the municipality. When the application is approved it shall become effective in accordance with subsection (b) of Section 8-3c of the Connecticut General Statutes.

No approval of the Commission under this section shall be effective until a copy thereof, certified by the Commission, containing the name of the owner of record, a description of the premises to which it relates, and specifying the reasons for its decision, is recorded in the land records of the town in which such premises are located.

#### 5.13 ADAPTIVE REUSE OVERLAY.

### 5.13 A General Requirements.

The Commission may allow, by Special Permit, the adaptive-reuse conversion of structures for uses not otherwise permitted in the zoning district, provided the following elements are met:

- 1. Structure must be located in a Residential Zone.
- 2. Structure must be designated as historically significant by (a) registry in the National Register of Historic Places or the State Register of Historic Places or (b) by the Planning and Zoning Commission upon presentation by the applicant of adequate evidence of historic significance such as a Historic Resources Survey.
- 3. Architectural integrity of the structure must be preserved, and the proposed use must be in harmony with surrounding uses.

#### a. B Allowed Uses.

- 1. The Commission may only allow uses that are allowed in other districts in Stafford.
- 2. If proposed uses or activities are not allowed in Residential Districts, then the proposed uses and all activity must occur within the historically significant structure.

### b. C Special Procedural Requirement.

- 1. The applicant shall notify the Stafford Historical Advisory Commission of all adaptive-reuse conversion applications at the time of filing such applications so that the Society may comment on:
- (a) The appropriateness of the use, and
- (b) Planned restoration / preservation of architectural integrity.

2. If comments are not received from the Stafford Historical Advisory Commission within Thirty (30) days after written notice is provided by the applicant, the Commission may assume that the Society has no comment.

## THIS PAGE INTENTIONALLY LEFT BLANK

## **Article VI**

## Site Development Standards

#### 6.1 Screening, Parking, Landscaping, Storm Water and Utility Standards

A. **Statement of Purposes:** The following standards are intended to enhance the appearance and natural beauty of the town and to protect and increase property values through preservation of existing vegetation and planting of new screening and landscaping material. Specifically, these standards are intended to reduce excessive heat, glare, and accumulation of dust; to provide privacy from noise and visual intrusion; and to prevent the erosion of soil, excessive runoff of drainage water, and the consequent depletion of the ground water table and the pollution of water bodies.

#### **B. Definitions:**

- 1. **Effect:** The visual impression desired from screening and landscaping.
- 2. **Large trees:** Deciduous shade trees such as sugar maple, pin oak, 59anadi plane, or linden, and conifers such as white pine, 59anadian pine, or 59anadian hemlock. Three-quarters (3/4) of the required large trees shall be at least two-and-one-half to three inches (2½"-3") in caliper at the time of planting and one-quarter (1/4) shall be at least four to four-and-one-half inches (4"-4½") in caliper at the time of planting.
- 3. **Flowering trees:** Trees such as dogwood, redbud, or crabapple.
- 4. **Ground cover:** A medium used in a confined area to check or prohibit the growth of undesirable plant materials. Ground cover may consist of plants such as pachysandra and myrtle or of materials such as white gravel, brick, or stone paving, in combination with live planting materials.
- 5. **Planting materials with Low Growth Habits:** Low growing evergreen shrubs such as junipers and shrubs such as goldmound spirea; seasonal plants such as lilies or hastas; and ornamental grasses.
- 6. **Hedge:** A hedge shall provide complete visual screening and consist of evergreens at least four (4) feet in height at the time of planting and it shall be maintained at a height of at least six (6) feet.
- 7. **Screening fence, or screening wall:** These are devices for complete visual screening. They shall be at least six (6) feet in height and three-quarters (3/4) solid.
- 8. **Partial visual screening:** A type of screening through that the screened object is partially visible.
- 9. **Complete visual screening:** A type of screening that affords a year-round effect and through that the screened object is obscured.
- 10. **Landscaped:** The term landscaped or landscaping shall mean that an area be at least covered with grass or ground cover. Any additional planting is either

specifically required by the ordinance or left to the discretion of the property owner.

## C. General Screening Standards:

- 1. Landscaping, trees, and screening plants required by the Zoning Regulation shall be planted and in a growing condition according to accepted horticultural practices, and they shall be maintained in a healthy growing condition. Any landscaping, trees, and screening plants that are in a condition that does not fulfill the intent of this ordinance shall be replaced by the property owner during the next planting season for the particular plant material.
- 2. A screening fence or wall required by this ordinance shall be maintained by the property owner in good condition throughout the period of the use of the lot.
- 3. All landscaping, trees, and screening material adjacent to parking areas, loading areas, or driveways shall be properly protected by barriers, curbs, or other means from damage by vehicles.
- 4. To the extent that existing healthy trees, if properly located, are preserved, they shall be fully credited against the requirements of the Zoning Regulations. The Commission may determine that trees shall be preserved as part of the site plan.
- D. **Screening Specifications:** The following specifications are considered as prototypes. In order to comply with the stated "desired effect," the number and spacing of required trees and the width of the screening strip may be varied.
  - 1. **Type "A" screening:** The desired effect is partial visual screening. Plant material shall consist of large trees spaced forty (40) feet on centers or flowering trees spaced twenty (20) feet on centers or a mixture of both. Where a continuous landscaped screening strip is impractical, the trees may be located in islands at least twenty (20) square feet in area.
  - 2. **Type "B" screening:** The desired effect is partial visual screening between zoning districts of different classification. Plant material shall be the same as specified for Type "A" screening except that one-half (½) of the trees shall be evergreens and the number of trees required is doubled.
  - 3. **Type "C" screening:** The desired effect is complete visual screening of parking and loading areas. The screening material shall consist of a hedge, screening fence, screening wall, or a combination thereof.
  - 4. **Type "D" screening:** The desired effect is separation and partial visual screening without creation of visual obstructions for traffic. The screening material shall consist of planting materials with low growth habits interspersed with flowering trees at twenty (20) feet on center and deciduous shade trees spaced forty (40) feet on center.
- E. **Screening and Landscaping:** The Commission may consider and approve such modifications in the above standards in a specific case where the Commission makes a finding that equivalent or superior screening will be provided.

#### **Buffers:**

The following tables outline the buffer sizes and planting types required between zones and along side lines.

|                            | <b>BUFFER PLANTING</b>  | BUFFER PLANTING       |
|----------------------------|-------------------------|-----------------------|
|                            | TYPE                    | TYPE                  |
|                            | Commercial/Industrial   | Commercial/Industrial |
| USE                        | То:                     | To:                   |
|                            | Residential Zone or Use | Non Residential Zone  |
| LOCATION                   | Residential Zone of Ose | or Use                |
| Building Front (street)    | D                       | D                     |
| Parking Lot Front (street) | D                       | A                     |
| Building Side and Rear     | C                       | D                     |
| Parking Side and Rear      | C                       | В                     |
| Tarking Side and Rear      | C                       | C                     |
| Loading Areas              | C                       | C                     |

|    |        | BUFFER STRIP |      |     |     |     |
|----|--------|--------------|------|-----|-----|-----|
|    |        | WIDTH        |      |     |     |     |
|    | STREET | Between      |      |     |     |     |
|    | LINE   | Residential  | LB   | HB  | I   | HI  |
| LB | 10'    | 10'          | 0    | 10' | 50' | 25' |
| HB | 25'    | 25'          | 10'' | 10' | 25' | 25' |
| I  | 50'    | 50'          | 50'  | 30' | 0   | 30' |
| HI | 30'    | 30'          | 25'  | 25' | 30' | 10' |

The Commission may consider and approve such modifications in the above standards in a specific case where the Commission makes a finding that equivalent or superior screening will be provided.

### **6.2** Off-street parking and loading regulations:

Parking facilities shall be provided off the street or highway right-of-way, on all premises, sufficient to accommodate the motor vehicles of all occupants, employees, customers, and other persons normally visiting the premises at any one time.

Properties in the Central Business District shall not be required to adhere to these requirements when new construction is involved and/or uses are changed. However, such activities may be subject to review and approval by the Commission to determine that such proposals do not create parking, or loading problems in the district. Each proposal shall be considered on its own merit. (Effective 7/15/95)

#### A. Submission and Approval of Plans:

Applications for zoning permits (effective 11/15/03) except for one (1)- and two (2)-family houses shall be accompanied by a plot plan drawn to scale, showing the location, size, and arrangement of off-street parking and loading facilities required by these regulations and the means of access to said facilities from the public street and any separate egress from such facilities. Such parking plan shall also show proposed screening, landscaping, lighting fixtures, drainage, and other improvements. The parking plan shall be submitted to the Zoning Enforcement Officer for approval as complying with the Zoning Regulations, including adequate relationship of entrances and exits to the flow of traffic on the public streets, safeguarding of pedestrians on public sidewalks and in the parking facility itself, and the adequate access and circulation of the vehicles and pedestrians using the parking facility. The parking plan shall be approved before a building and/or zoning permit is issued.

- i. The applicant shall submit information on projected periods of parking usage to the Commission to assist in their determination.
- ii. The Commission may require that a component of the total parking area that would be utilized only during peak periods of demand be surfaces with an appropriate pervious surface.
- iii. Traffic lanes in parking lots shall facilitate traffic movement and maneuverability, especially for ambulances and fire fighting vehicles.
- iv. All areas used for parking and driveways within the zone, except the buffer strip and any other planting strip, shall be provided with a dust free, all-weather surface.
- v. The perimeter of all parking areas that abut any required buffer strip, another property line, or a street line shall be provided with wheel, or bumper guards, so situated, designed, and maintained, that no part of any vehicle parked therein will extend beyond the zone, or street line, or into the buffer strip. The Planning and Zoning Commission may also require guards, or fences within the area for the regulation of traffic and the alignment of parked vehicles, or for either purpose.

## **B.** Parking Lot Landscaping Requirements

- Landscaped end islands and interior islands and planting strips shall be provided in parking lots to avoid the visual and environmental impacts of large expanses of asphalt.
- ii. Landscaping within the parking lots shall contain a minimum of one (1) shade tree, three (3) inch caliper and three (3) shrubs (plantings with low growth habits), with a minimum height of twenty-four (24) inches for each ten (10) parking spaces, planted within islands, planting strips, or buffer areas, as provided below.
- iii. There shall be no more than twenty-five (25) contiguous parking spaces without an interior or end island.

- iv. End islands shall be provided at each end of each row of parking spaces, curbed and landscaped, and a minimum of ten (10) feet wide by eighteen (18) feet long for a single row of spaces and ten (10) feet wide by thirty-six (36) feet long for a double row of spaces.
- v. Planting strips shall be provided in every third (3<sup>rd</sup>) set of interior parking spaces between abutting rows of spaces, curbed and landscaped, at a minimum width of ten (10) feet. Such planting strips may contain a pedestrian walkway.
- vi. Trees in or adjacent to parking lots shall be of a variety suitable for a parking lot environment that provide shade or are capable of providing shade at maturity.
- vii. No parking area or driveways shall be closer than ten (10) feet from any portion of a building other than its garage entrance, or loading area apron. This ten (10) foot area shall be used for walkways and/or landscaping.
- viii. All curbing to be Cape Cod style curbing.

## C. Variations to Landscape Requirements

- Additional Landscaping The Commission may require additional landscaping, or more mature plantings when unusual conditions require more extensive screening or for noise abatement to prevent the depreciation of adjoining residential properties.
- b. Reduced Landscaping The Commission may reduce the landscape requirements by not more than twenty-five percent (25%) for excellence in building or space design, where site conditions are appropriate to a revised landscape plan. The Commission shall consider, among other features, the site characteristics, compatibility of proposed structures with surrounding architectural types, quality of building materials, and the size and quality of landscape materials.
  - 1. Completion of Landscaping
    - In new construction, all landscaping shown on the approved plan shall be completed before issuance of a bond in a form and amount satisfactory to the Commission assuring completion within a specific time (not to exceed one (1) year) shall be filed with the Commission. Such bond shall be forfeited if the work is not completed within such time limit.
  - 2. Landscape Plan Requirements

Landscape plans shall include a plant list, with plant names, quantities, size at planting, and size when mature. Typical sections may be shown. Existing planting shall be identified on the plan.

#### **D.** Pedestrian Circulation

- a. The development of the site should include facilities for safe and convenient pedestrian circulation.
- b. Each site shall include a comprehensive system of pedestrian circulation, linking key components of the site with relevant external areas. A concrete sidewalk with a minimum width of four feet may be required by the
- c. Commission if, in its opinion, circumstances warrant such facility along the frontage of the property, parallel to the public road(s). If there is insufficient room within the right-of-way to construct the sidewalk, it shall be within a public access pedestrian easement.
- d. Walkways with a minimum width of six (6) feet shall be constructed along the entrance of all retail and related uses that may be used by pedestrians.
- e. Pedestrian paths in the form of concrete walkways or areas of the parking area differentiated from vehicular travel areas by pavement shall link the building entrances with parking areas, sidewalks along the street, and adjacent properties, if required.

## E. Location of Parking Facilities:

The parking and loading facilities that are required by these regulations shall be provided on the same lot or premises with the structure or land use they are to serve. If a lot cannot accommodate the required parking facilities, the board may grant off-site parking. Parking to the rear or side of buildings is encouraged.

#### F. Parking Specifications:

- Parking and loading facilities shall be designed to provide for safe circulation of vehicular and pedestrian traffic within the parking area and in relation to adjacent streets.
- 2. Parking and loading facilities shall be laid out so that vehicles shall enter and leave the public street or right-of-way only at the approved entrances and exits.
- 3. All parking spaces in parking areas of more than twenty-five (25) cars and all loading spaces shall be so located that vehicles entering or leaving such spaces do not block any entrance drive to the parking facility within twenty (20) feet of any street.
- 4. Any area of one hundred (100) square feet or more in a parking lot that is not required for a parking space, loading space, aisle, driveway, or walkway shall be landscaped.
- 5. Provisions shall be made to prevent vehicles from overhanging any walkway and from damaging trees or other landscaping materials.
- 6. Off street parking must provide for handicapped parking in conformance with ADA regulations. Each space must be as a minimum, one handicapped parking space for

up to twenty-five (25) spaces and one (1) handicapped parking space for each twenty-five (25) additional spaces.

7. Dimensions of parking spaces and aisles shall be at least as follows:

| Width of space: nine (9) feet or ten (10) feet if adjacent to wall or column.  Width of handicapped space: fifteen (15) feet with additional 6-foot-wide cross hatch aisle located on right side of vehicle's approach to the space. |         |         |  |  |  |
|--|---------|---------|--|--|--|
| Length of space: 18 feet   |         |         |  |  |  |
| Width of Aisle Two-way One-way   |         |         |  |  |  |
| If spaces are at an angle of 80 degrees or more to the aisle.  | 24 feet | 24 feet |  |  |  |
| If spaces are at an angle of 70 degrees to 79 degrees to the aisle. 20 feet 19 feet  |         |         |  |  |  |
| If spaces are at an angle of 60 degrees to 69 degrees to the aisle.  | 20 feet | 18 feet |  |  |  |
| If spaces are at an angle of 40 degrees to 59 degrees to the aisle. 20 feet 13 feet  |         |         |  |  |  |
| If spaces are at an angle of less than 40 degrees to the aisle.  | 20 feet | 12 feet |  |  |  |

- 8. Dimensions of loading spaces shall be at least as follows:

  - b. One-half loading space ...... 12 feet x 22.5 feet

## G. Temporary waiver of complete installation of parking facilities:

The Commission may waive the immediate installation of up to one-half (½) of the parking requirement for any proposed use, if, upon application and submission, in the commission's opinion sufficient evidence as to the reasonableness of the application was shown and the reduced parking supply will adequately serve said land use. The plot plan of the land use, however, shall show the complete layout for the full parking requirement in conformance with the reasons for the waiver. The Commission shall require the installation of additional parking facilities up to the full parking requirement if, in the Commission's opinion, such installation becomes necessary. Before the approval of the waiver, the applicant shall file an agreement with the Commission stating that he, his heirs, and assigns will install such additional parking facilities within six (6) months after the date of the Commission's vote to require such installation.

### H. Schedule of Off-Street Parking Requirements:

The following schedule of parking requirements shall apply: singularly for a single use and in combinations for more than one use:

|    | Use  | Spaces required                                      |
|----|--|--|
| 1. | Dwelling   | Two (2) per dwelling unit                            |
| 2. | Dwelling with medical or dental office           | Two (2) per dwelling unit, one (1) per employee plus |
|    |  | four (4)   |
| 3. | Dwelling with professional office                | Two (2) per dwelling unit, one (1) per employee plus |
|    |  | two (2)  |
| 4. | Dwelling with home occupation                    | Two (2) per dwelling unit plus one (1)               |
| 5. | Temporary Farm Stands                            | Minimum of four (4)                                  |
| 6. | Theater, assembly hall, or auditorium with fixed | One (1) per three (3) seats                          |
|    | seats  |  |
| 7. | Other spaces of public assembly or recreation    | One (1) per six (6) legal occupants                  |

| Use |   | Spaces required  |  |  |
|-----|---|--|--|--|
| 8.  | Hotel or boarding, rooming, or tourist home   | One (1) per guest sleeping room plus office and restaurant requirement where applicable.  Meeting and conference rooms: one (1) per three (3) persons of design capacity |  |  |
| 9.  | Convalescent or nursing home; hospital  | One (1) per three (3) beds   |  |  |
| 10. | Bowling alley   | Three (3) per lane   |  |  |
| 11. | Business offices and banks  | One (1) per two hundred and fifty (250) square feet of gross leasable area. ( <b>Effective 4/1/06</b> )  |  |  |
| 12. | Retail and personal service stores  | One (1) per two hundred and fifty (250) square feet of gross leasable area. ( <b>Effective 4/1/06</b> )  |  |  |
| 13. | Gasoline service station  | Minimum of five (5) spaces   |  |  |
| 14. | School  | Twenty (20) plus one (1) per staff member. For 11 <sup>th</sup> and 12 <sup>th</sup> grades or above, an additional one (1) per four (4) pupils.                         |  |  |
| 15. | Restaurants   | One (1) per three (3) seats.   |  |  |
| 16. | Physician's or dentist's office, clinics  | One (1) per doctor one (1) per employee and two (2) per examining room   |  |  |
| 17. | Industrial uses including wholesale and storage   | Two (2) per each three (3) employees employed or intended to be employed on the largest shift when the capacity of building is in full use.                              |  |  |
| 18. | B. A drive-in window shall have at least five (5) waiting positions between the street line and said window for cars approaching and at least one (1) waiting position for cars leaving said window. Waiting spaces shall not block other circulatory aisles or obstruct entrance or exit visibility and shall not cross pedestrian ways. |  |  |  |
| 19. | An attendant-operated car wash shall have at least ten (10) waiting positions on the property for each bay entrance and at least two (2) waiting positions for cars leaving a bay.  |  |  |  |
| 20. | <ol> <li>For uses not mentioned above, the minimum number of parking spaces shall be determined by the<br/>Commission.</li> </ol>   |  |  |  |

## I. Off-Street Loading Space Requirements:

| Use Classification and Building Size                                      |              |  |  |
|---|--------------|--|--|
| Retail store buildings:   |              |  |  |
| 2,000-5,000 square feet   | 1/2          |  |  |
| 5,000-25,000 square feet  | 1            |  |  |
| 25,000-50,000 square feet   | 2            |  |  |
| Each additional 50,000 square feet or part thereof                        | 1 additional |  |  |
| Office, apartment, institution, hotel, theater, public assembly building: |              |  |  |
| 5,000-25,000 square feet  | 1/2          |  |  |
| 25,000-50,000 square feet   | 1            |  |  |
| Each additional 100,000 square feet or part thereof                       | 1 additional |  |  |
| Industrial and warehousing buildings:                                     |              |  |  |
| 0-50,000 square feet  | 1            |  |  |
| 50,000-100,000 square feet  | 2            |  |  |
| Each additional 100,000 square feet or part thereof                       | 1 additional |  |  |

Note: All building areas are gross floor areas.

## J. Driveways and Access

a. The number of access drives and intersections shall be the minimum necessary to provide reasonable access to the proposed development.

- b. For all development within the HB and LB districts, there shall be a maximum of one driveway access where there is three hundred (300) feet or less of road frontage. If there is more than three hundred (300) feet of road frontage, there shall be a maximum of one (1) driveway access for each three hundred (300) feet of road frontage.
- c. The use of shared access for adjacent properties shall be provided where feasible, as determined by the Commission.
- d. The maximum width of the driveway shall be the minimum necessary to provide safe and efficient access to the property.

A common driveway is a driveway that provides access to two (2) or more lots. Common driveways serving residential building lots shall be limited to providing access to no more than two (2) lots. (Effective July 27, 2005)

#### K. Utilities

- a. No commercial or industrial development shall be approved unless adequate public utilities are to be provided by the developer.
- b. All development within the HB, LB, and CB districts with a gross floor area of twenty thousand (20,000) square feet or greater shall be served with public water and sanitary sewage.
- c. No development shall occur without an adequate supply of potable water for consumption and firefighting.
- d. There shall be adequate capacity for sanitary waste disposal generated by the proposed development.
- e. Electric power, telephone, and other cable systems shall be placed underground.
- f. Fire hydrants shall be placed at locations determined by the Commission upon advice from the Fire Marshal.
- g. The design of the proposed development must include storm water management so that the completed development will cause no net increase in storm water discharging from the site during twenty-five (25) year and fifty (50) year storm events when compared to pre-development conditions during such storm events. Storm water design will be in accordance with the DEEP Storm Water Quality Manual.
- h. Surface Water Drainage: Special attention shall be given to proper surface drainage so that surface waters will not adversely affect neighboring properties or public storm drainage facilities; will not obstruct the flow of vehicular or pedestrian traffic, and will not create standing water in paved or pedestrian areas. All surface water drained from roofs, streets, parking lots, and other site features shall be disposed of in a safe and efficient manner that will not create problems of water runoff or erosion on the site or on neighboring sites or pollution of surface water or groundwater. Insofar as possible, natural drainage courses and swales shall be properly stabilized and drainage-impounding areas shall be utilized to dispose of water on the site through natural percolation to a

degree equivalent to that prior to development. Also, appropriate erosion control measures shall be employed, including slope stabilization measures and the seeding of exposed areas to replace vegetative cover. Low Impact Design in accordance with the CT Stormwater Quality manual is encouraged.

## L. Outdoor Lighting:

The purpose of this regulation is to provide specific outdoor lighting standards in order to maximize the effectiveness of site lighting, to enhance public safety and welfare, to raise public awareness of energy conservation, to avoid upward illumination, to reduce glare, and to avoid light trespass onto adjacent properties and public streets. These regulations shall apply to the installation or improvement of outdoor lighting fixtures for non-residential land uses located in all zoning districts.

- a. Submission of Lighting Plan. Applicants proposing installation of outdoor lighting fixtures for nonresidential land uses shall file with a Lighting Plan subject to site plan review, unless waived by the Commission. Said plan shall be designed in accordance with the lighting standards and requirements and shall contain the following:
  - 1) Location, height and type of outdoor lighting fixtures, including building mounted.
  - 2) Fixture manufacturer's specification data, including lumen output and photometric data showing cutoff angles, and type of lamp.
  - 3) Isodiagram or photometric plan showing the intensity of illumination expressed in foot-candles at ground level, including at adjacent property lines.

#### b. Lighting Standards and Requirements.

- Outdoor lights and illuminated signs shall be designed, located, installed, shielded, and directed to prevent direct light at (and glare across) the abutting property lines and the public roads. The "maintained horizontal illuminance recommendations" as set by the Illumination Engineering Society of North America shall not be exceeded.
- 2) Outdoor lighting shall be full cut-off type fixtures, defined as a luminaire or light fixture that by design does not allow any light dispersion or direct glare to shine above a ninety (90) degree horizon plane from the base.
- 3) Outdoor lighting fixtures shall be limited to a maximum of fourteen (14) feet in height, unless otherwise designated by special permit.

#### **6.3 Sign Regulations**

The sign regulations are designed and adopted in compliance with Section 8-2 of the Connecticut General Statutes and with the aim of being content neutral. Where distinctions are made between the types of signs, such distinctions are made with the aim of protecting public health, safety, and welfare and are not intended to regulate the content of speech.

## A. Measurement of sign area:

The area of a sign shall be considered to be that of the smallest rectangle or triangle that encompasses all lettering, wording, design, or symbols together with any background different from the balance of the wall on that it is located, if such background is designed as an integral part of and obviously related to the sign. The supports that affix a sign to the ground or to a building shall not be included in the area of the sign unless such supports are obviously designed to be part of the sign as defined in these Regulations.

## **B. Sign Location:**

- 1. Signs projecting over a sidewalk or any other public walkway shall not exceed 8 Sq. Ft. in total area, nor shall the sign project over the sidewalk more than four (4) feet from the property line. All signs over sidewalks shall be located as to eliminate any hazard to the pedestrians.
- 2. Each unit of occupancy above the first floor may only display a sign on the inside of each window serving said unit of occupancy provided that the combined area of such signs shall not exceed six (6) square feet.

## C. Height of Signs:

- 1. No sign attached to a structure shall project more than two (2) feet over the top of the exterior wall of such structure at the location of the sign, except that in an industrial district, a sign may project up to six (6) feet over the top of the exterior wall of such structure provided the sign is not within five hundred (500) feet of the boundary of a residential district.
- 2. Where more than one (1) sign is permitted on a building, such signs shall be of uniform height and shape.

#### **D. Temporary Signs:**

- 1. Temporary yard signs may be installed before a specified special event. Such signs may remain for the duration of the event and shall be removed immediately after—within forty eight (48) hours—the termination of such special event. There shall be only one temporary sign within three hundred (300) feet of another in any zone; said sign shall not exceed nine (9) square feet and shall be located ten (10) feet or more back from the street line. Signs may not be placed on public property without permission. The Zoning Enforcement Officer will issue one (1) permit for the special event signs.
- 2. Temporary directional signs for roadside stands may be installed during the tenure of the roadside stands. Directional signs shall not exceed nine (9) square feet and shall be located ten (10) feet or more back from the street line. There shall be only one temporary sign within three hundred (300) feet of another in any zone. Signs may not be placed on public property without permission. The Zoning Enforcement Officer will issue one permit for the directional signs.

## E. Illuminated and Moving Signs:

- 1. A sign may be illuminated if illumination is confined to or directed to the surface of the sign. No flashing, rotating, or intermittent illumination shall be permitted except signs indicating time and/or temperature by means of white, intermittent lighting provided the longest dimension of such a sign does not exceed five (5) feet. Except for electronic message/changeable copy signs for municipal or civic purposes as described in 6.3.H.6.
- 2. No sign shall be mechanically rotated or moved provided the longest dimension of such signs does not exceed three (3) feet.
- 3. The light sources of signs shall be so designed and shielded that the sign does not shed light or create undue glare beyond the property lines of property on that the sign is located.

### F. Sign Permits and Tenure:

All signs, including temporary signs, under this Regulation or special use permit shall require the approval of the Zoning Enforcement Officer unless otherwise noted and shall meet all requirements of the Building Code and the Zoning Regulations. Applications for permits shall be made on forms supplied by the Zoning Enforcement Officer. Failure to maintain signs shall be considered a violation of this Regulation. Permits for temporary signs shall include a statement describing the special event or activity and the start and end dates of said event or activity. Upon discontinuance of a tenancy, it shall be the responsibility of the tenant and/or property owner to eliminate signs pertaining to a removed use, business, or proprietor within thirty (30) days of such discontinuance of tenancy.

#### G. Architectural Harmony of Signs:

All signs attached to a building shall be designed to be in harmony with the architecture of the building and with each other.

#### H. Signs in Residential Districts:

The following signs shall be permitted in residential districts and all other signs are expressly prohibited. (Effective 12/15/04)

- 1. Signs giving the name and/or address of the property are permitted provide there is only one such sign, it does not exceed two (2) square feet, and is located 10 feet or more back from the street line. Address numbers on a mailbox are excluded from this provision. Such name and address signs related to property address, including the name of the property owners, are allowed and regulated specifically to protect public health, safety, and welfare in regards to providing emergency services.
- 2. Common interest communities and apartment developments, where multiple residential units exist on private drives, may have no more than two (2) entrance signs, not to exceed thirty (30) square feet each, and located ten (10) feet or more back from the street line at the entrance to development. Such entrance signs are

- allowed and regulated specifically to protect public health, safety, and welfare in regards to providing emergency services.
- 3. Non-residential uses in residential zones may have two (2) signs, a pole or monument sign not to exceed twenty-four (24) square feet and a wall sign not to exceed thirty (30) square feet. The monument sign shall be located 15 feet or more back from the street line.
- 4. Temporary yard signs may be installed two-weeks before the specified special event. Such signs may remain for the duration of the event and shall be removed immediately after—within 48 hours—the termination of such special event. There shall be only one temporary signs in residential zones, said sign shall not exceed nine (9) square feet, and shall be located ten (10) feet or more back from the street line.
- 5. Traffic and roadway signs, as required by local, state, and federal law. Local, state, and federal highway standards shall dictate the size and location of said sign in the street right of way.
- 6. Electronic message/changeable copy signs for municipal or civic purposes may be permitted by Special Permit provided that;
  - a. Copy does not change more than once every 7 seconds; the electronic message center does not exceed 40 (forty) square feet and the height does not exceed 9 (nine) feet.
  - b. All EMCs are required to have automatic dimming capability that adjusts the brightness to the ambient light at all times of the day and night.

## I. Signs in Business and Industrial Districts:

The following signs are permitted in any business or industrial district and all other signs are expressly prohibited:

- 1. Signs permitted in a residential district:
- 2. Signs advertising the use of the land and building upon that displayed and the sale of goods or services on the premises and the name and location of the proprietor as follows:

|    | Maximum area of sign   | Location (See also<br>Sec. 3.42 General<br>Regulations) | Permit<br>Required |
|----|--|---|--------------------|
| a. | The total combined permitted exterior and permanent ground mounted signs area for each unit of occupancy on a lot shall not exceed two (2) square feet for each lineal front foot of exterior building wall. Maximum sign coverage is 200 Sq. Ft.              | Back of street line                                     | Yes                |
| b. | One detached double-faced sign may be erected on each lot. Such signs shall not exceed thirty-two (32) square feet in total area per sign face. A detached sign shall not exceed twenty (20) feet in height nor be within twenty (20) feet of a side lot line. | Ten (10) feet back<br>of street line                    | Yes                |
| c. | Within ten (10) feet of each entrance of a building, one (1) common sign may be provided allowing not more than two (2) square feet of sign area for each unit of occupancy that is served by  | Back of street line                                     | Yes                |

|    | Maximum area of sign   | Location (See also<br>Sec. 3.42 General<br>Regulations) | Permit<br>Required |
|----|--|---|--------------------|
|    | said entrance. The total area of each common sign shall not exceed twenty-four (24) square feet.   |   |                    |
| d. | Each unit of occupancy above the first (1 <sup>st</sup> ) floor may only display a sign on the inside of each window serving said unit of occupancy provided that the combined area of such signs shall not exceed six (6) square feet.  | -   | No                 |
| e. | One (1) portable sign of up to twelve (12) square feet in total area may be located on each street frontage of a property and shall be removed daily. The sign shall be removed immediately if, in the opinion of the Zoning Enforcement Officer, it endangers pedestrian or vehicle movement or is not maintained in good condition. Lighted portable signs are not permitted except by government agencies in conjunction with road construction or maintenance. | Back of street line                                     | No                 |
| f. | Other outdoor advertising devices: Outdoor advertising devices including, but not limited to, plaques, banners, pennants, streamers are permitted for a period of not more than two (2) weeks after the opening of a new business.   | Back of street line                                     | No                 |
| g. | A unit of occupancy that does not have a free-standing sign displayed on a property may join with other units to provide a common sign. Such sign shall be within the total square footage allowed on the premises. Said sign may be attached to a double-faced sign as permitted under #b. above. Each unit of occupancy shall be limited to four (4) square feet per face.   | Ten (10) feet back<br>of street line                    | Yes                |

#### J. Design Incentive in Business and Industrial Districts:

In Business and Industrial Districts, a fifteen percent (15%) increase in the total allowable size of signs as allowed in Section 6.1(I)(2)(a) and (b) may be granted, provided the following standards are met:

- 1. The sign or signs are not made of plastic.
- 2. The sign or signs is not internally illuminated.
  - a. Halo illumination is acceptable if reverse pan-channel letters with an internal light source reflecting off of the building are used for "halo" illumination and the light source is not visible.
- 3. The sign or signs is designed with the overall context of the building and character of the surrounding area.
- 4. The sign or signs is in proportion to the building, so they do not dominate the building appearance.
- 5. The sign or signs are made of wood (or a synthetic wood-like a material), metal, or other material that is architecturally consistent with the building facade.
- 6. The sign or signs are exterior lit by a shielded lamp directed at the sign and/or building façade in such a way as to prevent glare and the leakage of light of the property.

## c. Severability:

The provisions of these Sign Regulations are declared severable, to the maximum extent permitted by law. If any provision or provisions of these Sign Regulations or the application thereof to any person or circumstance is held invalid or unlawful, it is the intent of the Planning and Zoning Commission that said invalidity shall not affect other provisions of these Regulations, which shall remain in full force and effect as if such portion so declared invalid or unlawful were not originally part of these Regulations, even if the surviving parts of the Regulations result in greater restrictions after any unlawful provisions are stricken. In particular, and without limitation, in the event any portion of these Regulations are declared invalid as applied to noncommercial signs, the Regulations, or any surviving portions thereof, shall remain in full force and effect as applied to commercial signs.

## **Article VII**

# **Special Regulations**

## 7.1 Seasonal Dwellings

As used in this Regulation, a seasonal dwelling is a dwelling that is not designed or constructed for year-round occupancy, which may be lacking one or more of the following: 1) central heating; 2) no basement or cellar; 3) indoor plumbing; 4) potable water supply; 5) electricity; or 6) insulation.

## A. Seasonal Use for Campgrounds and Existing Cottages:

Seasonal dwellings may be used during the period of April 15 through October 15. From October 15 to April 15 no person or persons may reside therein for a total of more than thirty (30) days.

## **B.** Conversion of Seasonal Dwellings:

Nothing in these Regulations shall prevent the expansion or alteration or conversion to year-round occupancy of existing seasonal dwellings providing that:

- 1. All such enlargements and conversions meet the applicable yard requirements and do not exceed the maximum allowable lot coverage of the district in that it is located; and
- 2. That, in the written opinion of the Stafford Health District, the existing or proposed septic system meets the current State Health Code including but not limited to the size of the septic tank and the capacity of the leaching system for the number of bedrooms of said structure or said dwelling is connected to a public sewer system; and
- 3. That all conversions to year-round use shall meet the minimum livable floor area of the district in which it is located.

## 7.2 Earth Removal/Blasting

**A. Definitions:** (Effective April 25, 2007)

- 1. **Minor Landscaping:** Less than one (1) acre of disturbed area, associated with construction, building
  - a. Alterations or installation of subsurface utilities.
- 2. Gravel Bank: Commercial earth removal or mining operation for the purposes of excavation, processing, and/or blasting from the premises on a continuing basis, excluding bona fide farming operations who manufacture loam on site and permitted retail operations where products are delivered to and sold on the premises, retail only.

- **B. Temporary Earth Removal Operation:** An operation associated with landscaping or agricultural use where the disturbed area exceeds one (1) acre, excluding minor private residential landscaping of less than one half (1/2) acre not associated with a commercial operation.
  - 1. **General:** There shall be no excavation or removal from the premises in any district of sand, gravel, stone, loam, dirt, or other earth products except in connection with necessary excavation or removal of surplus material resulting from a bona fide construction or alteration of a building structure and/or installation of subsurface utilities, minor landscaping associated with such construction or alterations or minor landscaping associated with residential use or agricultural operation to be executed on the premises disturbing less than one (1) acre or except in connection with a special use permit use granted by the Commission authorizing such excavation or removal under the conditions hereafter specified. Special use permit is required for any operation where one (1) or more acres are to be disturbed throughout the life of the project. (Effective 4/25/2007)
  - 2. **Application:** An application in writing for a special use permit shall be submitted to the Commission for any excavation or removal of material not part of a bona fide construction project. Said application shall be submitted by the property owner or his/her authorized agent; it shall be accompanied by a plan of operation including maps and drawings prepared by a Connecticut-licensed engineer and land surveyor; and it shall include the following information:
    - a. A map showing the location of boundaries of the property on that the excavation is to take place, the location of area to be disturbed, and the estimated amount of material to be removed. If any excavation is within one hundred (100) feet of an abutting boundary, the map is required to be derived from an A2 survey. If the excavation is over one hundred (100) feet from an abutting boundary, the Commission may, at their discretion, allow the map to be derived from a Class D survey. (Effective 4/25/2007)
    - b. Grading plan at a scale of one (1) inch equals forty (40) feet (1" = 40') or less prepared from an actual field survey and based on a bench mark noted and described on said map showing existing topography in the area to be excavated and proposed future topography and indicating contour lines at not more than two (2)-foot intervals within the premises and within one hundred (100) feet of the area surrounding the excavation;
    - c. Existing and proposed drainage on the site and ground water level as determined by subsurface explorations;
    - d. Proposed truck access to the excavation, the number and type of trucks, or other machinery to be used on the site, the type of processing to be used, and the location and type of any buildings to be erected;

- e. Written permission for the inspection of the site at any reasonable time by a duly authorized representative of the Commission;
- f. In addition to the above requirements, the Zoning Commission may request that data related to soil conditions, locations, and depth to rock ledge be submitted.
- 3. **Approval:** The Commission, after public hearing, may grant a special use permit or renewal of a permit to excavate for a period not to exceed three (3) years for a gravel bank and one (1) year non-renewable for a temporary earth removal operation only when it is satisfied that the following conditions will be complied with: (Effective 4/25/2007)
  - d. That the proposed excavation will not impair the future use of the property in accordance with these Zoning Regulations and that the slopes and banks will not impair good development and safe use of the property after excavation;
  - e. That the premises will be excavated and graded in conformity with the plan of operation as approved; and any deviation from said plan, without the consent of the Commission, shall be cause for the Commission to revoke the permit;
  - f. That there will be no excavation or removal within fifty (50) feet of any property or street line except to an elevation equal or above the established grade of the adjoining street or property;
  - g. That there will be no excavation to a depth of less than six (6) feet above natural water table;
  - h. That processing such as screening, sifting, washing, crushing, or other forms of processing may be conducted on the premises of a commercial gravel bank within the guidelines set in a special use permit in concert with the Special Use criteria as written under Section 8.11(A) of the Zoning regulations; (Effective 4/24/2007)
  - i. Processing such as screening, sifting, washing, crushing, or other forms of processing may be conducted on the premises of a temporary earth removal operation or bona fide construction project by a temporary processing permit, providing all operations comply with the regulations herein. (Effective 4/25/2007)
    - i. Temporary processing permits will be good for six (6) business days from the date issued. This permit may be modified or renewed at the discretion of the Commission. Temporary processing permits may be renewed up to three (3) times for a total of four (4) permits per calendar year and must be used within thirty (30) days after approval by the Commission.

- ii. In the case of a residential subdivision development, a temporary processing permit can be issued, for processing of material on the overall construction site, until seventy (70) percent of the homes in the subdivision have been issued a certificate of occupancy by the building department. Once seventy (70) percent of homes in the subdivision have issued a certificate of occupancy, no temporary processing permits shall be issued. Activities related to the bona fide construction of a single family home, when less than two (2) acres are being disturbed, are exempt from this subsection.
- j. That, if arable soil exists within the site, the applicant has made provisions for storing and retaining within the premises at least six (6) inches of the top layer of said soil for use, after excavation and/or removal operations are completed, as specified herein;
- k. That at all stages of operation, proper drainage will be provided to prevent the collection and stagnation of water to prevent soil erosion and to prevent harmful effects upon surrounding properties, and proper slopes will be maintained to prevent dangerous overhang;
- 1. That, during the period of excavation and removal, such barricades or fences will be erected as may be necessary for the protection of pedestrians and vehicles:
- m. That truck access to the excavation will be arranged to provide a minimum site line of two hundred and fifty (250) feet as to minimize danger to traffic. Proposed site lines must be approved by the Stafford Department of Public Works or the State of Connecticut DOT, whichever applies. Proper provisions shall be made to control the nuisance of dust and mud to the surrounding properties and road ways where equipment and trucks will be entering and exiting the site. Anti-tracking aprons shall be installed and properly maintained at the site entrance of all commercial gravel band or temporary earth removal operations. (Effective 4/25/2007)
- n. That, when excavation and removal operations or either of them are completed or if a permit has expired and not been renewed, the excavated area will be graded so that gradients in disturbed earth will not be steeper than a slope of three-to-one (3–1) (horizontal–vertical) or whichever lesser slope is necessary to maintain stability under particular soil conditions; that a layer of arable soil obtained from material stockpiled on the site, or from elsewhere if no on-site stockpile is available, will be spread over the excavated area to a minimum depth of three (3) inches; that the area will be covered with a perennial rye grass or other equally suitable vegetation; and that it will be maintained until accepted by the Commission.
- o. The following table of minimum setback requirements shall be applicable to any use falling under the requirements of Section 7.2: (Effective 4/25/2007)

Minimum Required Distances Residential Home Abutting Property Line

| Excavation | 100 feet  | 50 feet  |
|------------|-----------|----------|
| Screening  | 200 feet  | 100 feet |
| Crushing   | 1000 feet | 250 feet |

g. The following table for hours of operation shall be applicable to any use falling under the requirements of Section 7.2: (Effective 4/25/2007)

## **Hours of Operation:**

- i. Commercial Gravel Operation: Mon-Sat 8:00 a.m. to 4:00 p.m. for excavating and processing
- ii. Mon-Sat 7:00 a.m. to 5:00 p.m., Sun 10:00 a.m. to 2:00 p.m.
- iii. For all other operations: Sunday operations allowed in HB, LB and IN zoning districts **only.**
- iv. No operations on New Year's Day, Easter, Memorial Day, July 4<sup>th</sup>, Labor Day, Thanksgiving, and Christmas.
- v. Temporary Earth Removal: Mon-Sat 9:00 a.m. to 4:00 p.m. for all operations.
- vi. No operations on New Year's Day, Easter, Memorial Day, July 4<sup>th</sup>, Labor Day, Thanksgiving, and Christmas.
- **C. Blasting:** Blasting to remove earth products is permitted under the following conditions: (Effective 5/4//1998)
  - 1. Blasting is permitted as a special use in conjunction with a valid earth removal permit. Blasting shall be included as part of earth renewal permit process. If blasting is proposed for a new area on a site, applicant must first apply for a special use permit, submitting a plan of operation as outlined in Section 7.2(B)(2), to include area of proposed blasting.
  - 2. No blasting shall be allowed within two hundred and fifty (250) feet of a residential dwelling or well without written consent of property owner.
  - 3. Blasting may be conducted Monday through Friday during the hours of 8:00 a.m. to 2:30 p.m. No blasting on holidays.
  - 4. All Blasting shall be done under the direction of the Town of Stafford Fire Marshall or the Assistant Fire Marshall, and all applicable State laws and regulations promulgated by the Office of the State of Connecticut Fire Marshall shall be controlling.
  - 5. Blasting for utilities and building foundations shall not be governed by these regulations.
- **D. Performance bond:** Upon approval of the special use permit and prior to the start of any excavation, the applicant shall file a performance bond with the Treasurer of the Town of Stafford in form approved by the Town Attorney and in an amount established by the Commission. The Commission may, at the applicant's expense, hire a professional engineer to review the applicant's site plan and provide a cost analysis for reclamation of the land as a guide for determining said bond. Prior to full or partial refund of said bond, an inspection of the disturbed area by a duly appointed

- official of the Town of Stafford or said Engineer, shall be completed to insure compliance with the conditions set forth herein. (Effective 4/25/2007)
- **E. Enforcement:** As a condition for granting a special use permit, the Commission may require the applicant to submit periodic reports of progress with the excavation or removal, including contours and cross sections, prepared and certified by an engineer or land surveyor licensed in the State of Connecticut. If at any time the Commission finds that the excavation is not being conducted in accordance with the plan of operation as approved, the Commission may order the applicant to cease operations.
- 7.3 Regulations concerning kennels, veterinary hospitals, and animal agriculture uses: (Effective 12/19/2003) (Rev. 9/28/10).

The purpose of these controls is to insure the protection of the public health, safety, and welfare of the residents of the Town of Stafford. Effective January 1, 2004, dogs kept at locations not being used as commercial kennels, grooming facilities, or training facilities are exempt from these regulations. Effective February 1, 2004, the exemption to Section 7.3 shall only apply to households with five (5) or fewer dogs and will include all household dogs registered with the Town Clerk as of January 30, 2004. Households having more than five (5) registered dogs as of January 30, 2004, shall be permitted to renew the registrations for the lives of the dogs, but such households shall not be permitted to keep any additional dogs until attrition has reduced the total number of dogs to four (4) or less.

- A. Commercial Kennel and/or Dog Training Facility: Said uses may be permitted as special permit uses within the corresponding district providing:
  - 1. That the lot area is no less than ten (10) acres;
  - 2. That a maximum total of thirty (30) dogs are kept in the kennel building(s);
  - 3. That no building for the boarding, breeding, or selling of dogs, or exercise run is within three hundred (300) feet of any property or street line;
  - 4. That any building housing dogs shall be of solid construction, be masonry, or conventional framed with sheathing and insulation, shall have finished interior walls, and all external doors shall be of solid core construction:
  - 5. That exercise runs shall have finished floors with covered drains and shall be separated by solid partitions of at least four (4) feet in height;
  - 6. That all ceilings shall be insulated and finished with sound absorbent materials;
  - 7. That kennel rooms shall be mechanically ventilated and shall not have windows that open;
  - 8. That no animals shall be housed in outside kennels, and
  - 9. That the special use permit shall be renewable after two years.
- B. **Noncommercial Kennel:** Said use may be permitted as special permit uses within the corresponding district providing;

- 1. That the lot area is no less than five (5) acres;
- 2. That a maximum of ten (10) dogs are to be kept at the residence;
- 3. That the dogs are kept inside the residence from 9 p.m. to 7 a.m.;
- 4. That a fence adequate to contain the dogs during exercise is provided and that no part of the fence is within fifty (50) feet of any property or street line;
- 5. That landscape buffers be included in the plan submitted with the application for a special permit, and
- 6. That the special use permit shall be renewable after two years.
- C. **Veterinary Hospital:** Said uses may be permitted as special permit uses within the corresponding district providing;
  - 1. That the lot area is no less than four (4) acres in AAA and AA districts and two (2) acres in LB and HB districts.
  - 2. That a maximum total of thirty (30) animals are kept in the kennel building(s);
  - 3. That no building for caring of animals is within fifty (50) feet of any property or street line in AAA and AA districts. Setbacks in LB and HB districts shall be twenty (20) feet on the side and rear and thirty-five (35) feet from the front property line;
  - 4. That any building housing animals shall be of solid construction, be masonry or conventional framed with sheathing and insulation, shall have finished interior walls, and all external doors shall be of solid core construction;
  - 5. That all ceilings shall be insulated and finished with sound absorbent materials;
  - 6. That rooms where animals are kept shall be mechanically ventilated and shall have no opening windows;
  - 7. That no dogs shall be housed in outside kennels, and
  - 8. That landscape buffers be included in the plan submitted with the application for a special permit.
- D. **Grooming Facility:** Said use may be permitted after site plan review by the Zoning Enforcement Officer when used only for grooming animals in HB and LB zones the following conditions are maintained:
  - 1. The facility is established in an existing commercial building designed for multiple tenants and;
  - 2. The total floor space of the facility does not exceed fifteen hundred (1,500) square feet and:
  - 3. Animals shall not be kept overnight and;
  - 4. All animals shall be kept inside a building constructed to reduce animal noises and;

- 5. All other facilities, when used only for the grooming, shall have no overnight boarding and must comply with all other standards of section 7.3I of these regulations.
- E. **Horse and pony farms, riding academies, or boarding stables:** Said uses may be permitted as commercial agriculture, under a special use permit within the corresponding district providing;
  - 1. The lot area is no less than ten (10) acres; and
  - 2. That the lot area allocated for the keeping of animals is provided at the rate of three (3) acres of land for the first horse or pony on the premises and one (1) additional acre for each additional animal thereafter; and
  - 3. That no riding ring or boarding stable or any similar structure or area for the keeping of animals or for the storage of fertilizer or manure will be located in an area that will affect the health and welfare of neighboring properties.
  - 4. A single dwelling for a full time caretaker or watchman may be allowed on the premises in addition to a primary dwelling but shall be limited to three (3) rooms totaling seven hundred fifty (750) square feet of living space. Said dwelling shall not be freestanding and shall be an integral part of a building housing horses or facilities.

Stafford is fortunate to have come from a long history of farming with some farmland deeded to their current family owners by the King of England. Prime farmland soils are limited in amount in town and most of Stafford soils are poor and rocky. Nonetheless, the remaining farmland is agriculturally productive promoting a visually attractive environment for people, animals, plants, and wildlife. For these reasons and in order to preserve our remaining farmland, it is the intended purpose of the Zoning Board of the Town of Stafford to allow agricultural use and activities within the boundaries of current regulations. We intend to do this, while at the same time, being considerate of neighboring residents and complying with Town agencies.

## 7.4 Regulation concerning the keeping of animals associated with agriculture

- A. Purpose: The purpose of these controls is in furtherance of the public health, safety, and welfare of the residents of the Town of Stafford and to protect and enhance the living environment and the value of properties thereof. Toward these ends, these controls provide minimum adequate standards for the keeping of domestic animals and the location and number of structures for the housing of said animals.
  Applicability: These controls apply to the keeping of farm animals in any zone.
  Commercial farms over ten (10) acres in size are exempt from the table in 7.4d.
- B. **Permit:** A permit shall be required when the keeping of such animals is on an acre or less.
- C. **The Zoning Enforcement Officer shal**l grant approval for an Animal Agricultural Permit when satisfied that the application complies with the criteria and standards presented in this section of the regulations.

## D. Table of Units Per Acre for Agricultural Animals:

The number and type of animal constituting an animal unit as well as the number of animals permitted by lot size is described in the following table:

| Type of animal                       | Number of animals per animal unit | Number of animals permitted on space |
|--------------------------------------|-----------------------------------|--------------------------------------|
| Horse, pony, mule, donkey, cattle,   | One (1)                           | One (1) unit for two (2) acres and   |
| llama                                |                                   | one half (½) acre for each           |
|                                      |                                   | additional unit                      |
| Sheep, goat, alpaca                  | Four (4)                          | One (1) unit for two (2) acres and   |
|                                      |                                   | one half (½) acre for each           |
|                                      |                                   | additional unit                      |
| Pig                                  | Two (2) mature pigs plus one (1)  | One (1) unit for two (2) acres and   |
|                                      | litter (3 months or less)         | one half (½) acre for each           |
|                                      |                                   | additional unit                      |
| Poultry including chickens, ducks,   | Twelve (12)                       | One (1) unit per acre when kept as   |
| geese, and turkeys (see also 5.03I4. |                                   | single type animal on the property   |
|                                      |                                   | or ½ unit on lots under an acre.     |
|                                      |                                   | When combined with other animals     |
|                                      |                                   | such as horses, sheep, or pigs, one  |
|                                      |                                   | half (½) unit will be allowed as a   |
|                                      |                                   | secondary unit for each primary      |
| 5.11.                                | T (10)                            | unit.                                |
| Rabbit                               | Ten (10)                          | One (1) unit per acre when kept as   |
|                                      |                                   | single type animal on the property   |
|                                      |                                   | or ½ unit on lots under an acre.     |
|                                      |                                   | When combined with other animals     |
|                                      |                                   | such as horses, sheep, or pigs, one  |
|                                      |                                   | half (½) unit will be allowed as a   |
|                                      |                                   | secondary unit for each primary      |
|                                      |                                   | unit.                                |

Note: Combinations of different animals and fowl are permitted provided total permitted animal units are not exceeded for available acreage. Example – two (2) sheep and two (2) goats would be permitted on two (2) acres. However, one (1) cow and one (1) goat would not.

## E. Minimum requirements

- 1. **Maintenance:** The care for and number of such animals and the maintenance of the area in that they are to be kept shall be such that no environmental health hazard or nuisance will be created and that the values of abutting properties are not adversely affected.
- 2. **Poultry:** (Effective 4/26/05) The keeping of these animals shall meet all other requirements of Section 7.4. Roosters and guinea hens are only allowed on lots over three (3) acres.
- 3. For the "A, B and C" zones, a special use permit is required and will comply with the following standards:
  - a. Maximum of twelve (12) birds
  - b. A coop and fenced in play yard that is a complete enclosure on all sides including the top;

- c. An adequate landscape buffer surrounding the "fenced structure" to provide noise and sight relief to neighboring properties; and
- d. Coop/fenced area must be at least twenty-five (25) feet from the side and rear property lines.
- 4. That the keeping of pigs or the building of a pigsty on marshy ground or land subject to over flow or within three hundred (300) feet of any inhabited house upon property other than that of the proprietor of the pigsty is prohibited.
- The animals are kept in a manner that conforms to all applicable regulations of the Public Health Code, the Department of Environmental Protection, the Connecticut Department of Agriculture, and the General Statutes of Connecticut.
- F. Shelters, stables, barns, or similar structures, buildings, or appurtenance used for the keeping or housing of animals:

No such structure used for the keeping of any such permissible animal, or animals, or for the storage of feed shall be located:

- 1. Closer to the side and rear property lines than the minimum setback requirements of the zoning district in that the premises are located.
- 2. Closer than one hundred fifty (150) feet from a front lot line.
- 3. Within one hundred fifty (150) feet from an existing dwelling under separate ownership.
- G. **Storage of manure:** The storage of manure shall meet all requirements of local and state health officials, but in no case will it be located closer than one hundred fifty (150) from any neighboring residence.
  - p. All other animals not covered under these regulations may be permitted at the

discretion of the commission or the zoning enforcement officer.

q. Students whose projects involve the keeping of farm animals are authorized

provided they present a statement of use as well as an animal management plan (see definition). This must be found acceptable with respect to animal welfare and potential environmental/neighborhood impacts by a 4-H instructor and or school project manager.

## 7.5 Regulations Concerning Mobile Homes and Camping Areas

#### A. Mobile homes

- 1. With the exception of a mobile home that is clearly designed or intended for use as a temporary field office or a trailer field office, a mobile home shall be considered as a dwelling occupying a lot and shall be subject to the same lot area, floor area, yard setback, and lot area per dwelling unit requirements applicable to dwellings in the zoning district in which such mobile home is located.
- 2. No mobile home shall be used for human habitation or commercial purposes when parked upon a public highway.
- 3. Not more than one (1) mobile home at a time may be parked, placed, or stationed on one (1) lot. Adjoining lots having common ownership shall be considered one (1) lot for the purpose of this restriction.

## B. Camping areas

- 1. No camping areas may be expanded or operated unless a special use permit for such purposes has been issued by the Commission after a public hearing.
- 2. Application for a special use permit shall be made to the Commission on a form prescribed by it. The application shall include, in addition to requirements of Section 8.3, a site plan in duplicate at a scale not less than forty (40) feet to the inch showing: lot lines, dimensions of the lot, setback dimensions; location and description of shelter and toilet facilities; location and capacity of parking facilities; general layout of campsites including overflow field access and egress roads; and such other information as may be required by the Commission to determine that the proposed facility complies with these Regulations. The site plan shall also show the names of the owners of lands abutting the property for that a camping area permit is being requested.
- 3. No permit for the expansion and/or operation of a camping area will be issued by the Commission unless the following minimum requirements are met:
  - a. **Minimum area of lot:** Forty (40) acres;
  - b. **Maximum density:** Eight (8) campsites per acre or at the rate of five thousand (5,000) square feet per campsite;
  - c. **Minimum toilet facilities:** One (1) men's room and one (1) ladies' room for each fifteen (15) campsites or fraction thereof. Each men's room shall include at least: a shower room, two (2) lavatories, a utility sink, two (2) water closets, and two (2) urinals. Each ladies' room shall include at least: a shower room, two (2) lavatories, a utility sink, and three (3) water closets. Each toilet facility shall have adequate provisions for the supply of running hot and cold water to corresponding fixtures. All toilet facilities proposed shall meet the approval of the town Health Officer.

In those camping areas where the use is restricted to self-contained camping vehicles with full sanitary facilities, the above requirements shall not apply;

but sanitary facilities shall be as required by the Public Health Code of the State of Connecticut as to number, location, physical features, and construction. A self-contained camping vehicle means a camping vehicle equipped with a sink, shower, toilet, a water supply tank and a holding tank, or tank for sink, shower and toilet waste.

- d. **Refuse disposal:** The site plan shall also indicate areas designated for the location of refuse containers. Said containers shall be rodent and odor proof and located within suitable visual barriers away from campsites
- e. **Shelter:** All-weather shelter structures shall be provided. Said shelter shall be of adequate size and design and located within walking distance from the camping sites.
- f. **Buffer zone/strip:** Each camping area shall be provided with a buffer zone/strip of at least fifty (50) feet around its boundaries. If natural vegetation does not exist, said buffer zone/strip shall be properly planted with evergreen trees and/or shrubbery of no less than six (6) feet in height so planted as to provided adequate year-round screening.
- g. **Open space recreational area:** Each camping area shall be provided with open space recreational areas (exclusive of overflow fields, campsites, and circulation spaces) at the rate of no less than twenty (20) percent of the total area of the lot.
- h. Standard permit is from April 15 to October 15
- 4. Camping areas shall be provided with driveways of adequate design and paving and shall meet the approval of the Commission.
- 5. Permittees of camping areas shall maintain an up-to-date register indicating the names of campsite occupants including number of license plates of all motor vehicles and trailers in the campgrounds. A copy of said register shall be available for inspection by the Commission or its authorized agent.
- 6. Initial permits for operation of camping areas shall meet the approval of the Commission after public hearing. Permits will be issued on a yearly basis and each renewal thereafter shall also meet the approval of the Commission.
- C. **Trailers and mobile homes for special purposes**: Trailers and mobile homes may be utilized for the following:
  - 1. Trailers may be used for emergency storage necessitated by natural disaster, special function, or other good cause following application to and approval by the Commission or its agent.
  - 2. Mobile homes may be used as an office to service construction projects that may include facilities for a watchman following approval by the Commission or its agent.
  - 3. Temporary uses such as for emergency purposes that, in the judgement of the Commission or its agent, warrant a permit.

4. Mobile homes or trailers shall not be placed in a special flood hazard area under any circumstances.

The Commission or its agent shall establish reasonable requirements in respect to screening, time limits, hours of use, and other factors that control such use.

# 7.6 Regulations Concerning Automotive Sales and Repair, Service Garage and Gasoline Stations and compliance with CGS Secs. 14-51, 14-54, 14-321, as amended.

# **Applicability:**

- 1. No person, firm, or corporation shall establish, expand, or alter in use or structure any business concerned with: sale, repair, or servicing of automobiles and/or sales unless said businesses or uses are developed in accordance with the requirements of this section and all other applicable controls in these Regulations.
- 2. All sale, repair, or servicing of automobile and/or gasoline sales legally developed prior to the effective date of these Regulations shall be considered legal notwithstanding the provisions of this section.
- 3. Furthermore, no business or use referred to in this section shall become nonconforming by reasons of development of any park, or playground, school subsequent to the establishing of the business or use.

## **Issuance of permits:**

- 1. Issuance of a permit for the establishment, the expansion, or the alteration in use or structure, shall be contingent upon the approval or the disapproval by the Zoning Board of Appeals for sales and repair licenses, and by the zoning commission for gasoline sales.
- 2. The Zoning Board of Appeals or the Zoning Commission shall not give approval unless it finds the location is suitable for the use or business intended, due consideration being given to the proximity of schools, places of worship, libraries, theaters, playhouses, or other places of public gathering; or intersecting streets, traffic conditions, width of highway; and effect of public travel and that such use or business will not imperil the safety of the public. In no event shall a business or use referred to in this section be established if any part of the proposed building or structure will be within one hundred (100) feet of any entrance to a public park or playground. No permit shall be granted by the Zoning Board of Appeals unless the proposed use or business is within a district where such use or business may be permitted.

#### **General requirements:**

1. Safeguards shall be provided against surface and subsurface leakage of gas and oil

- 2. All applications to the Zoning Board of Appeals or the Zoning Commission shall include a report from the Zoning Enforcement Officer confirming that the site plan required as part of the application meets all zoning requirements.
- 3. All applications shall be accompanied with maps showing information as detailed in Section 8.3 of these Regulations.
- 4. Off-street parking shall conform to Sections 6.2.
- 5. There shall be at least ten (10) feet of landscaped area along the frontage of the premise; and where adjacent to paved areas and service roads, said areas shall be clearly defined by curbing.
- 6. There shall be a buffer zone/strip of at least twenty-five (25) feet between any of the uses or buildings described in this section and adjacent to lots zoned for or in residential use. Said buffer zone shall be properly landscaped and planted including Type "C" screening.
- 7. There shall be no exits or entrances closer than one hundred (100) feet to any road intersection nor shall there be any business or uses referred to in this section proposed for a location within one hundred (100) feet of any residential dwelling structure that is situated in a residential district.
- 8. Except for new or used auto sales operations, vehicles parked, or stored on the premises for periods exceeding ten (10) days and all damaged vehicles on the premises shall be housed within a fenced and obscured enclosure at least six (6) feet in height.

# D. Convenience store retail trade shall be permitted as an accessory use, provided that:

- 1. Adequate vehicular and pedestrian traffic flow patterns are established which prevent conflict with gasoline and service patrons.
- 2. Adequate parking for the additional retail trade use is provided in accordance with Section 6.2
- 3. To insure adequate supervision for both the gasoline sale and retail trade uses, there shall be at least one (1) employee on duty for the sale of gasoline, and one (1) additional employee for the retail trade use. Depending on the size of the facility, and the anticipated volume of vehicular traffic and public activity, the Commission may require additional security measures as a condition of the retail trade use.
- 4. Restroom facilities shall be provided for employees and may be required by the Commission for customers
- 5. There shall be no overnight parking associated with the retail trade use, other than for employees.

#### 7.7 Car Wash:

A car washing facility as a main use may be permitted by the Commission only as a special permit use. Plans for a proposed facility shall be submitted as in Section 8.3. In addition, the following shall apply:

- A. Vehicular circulation shall be controlled for safe entrance and exit and all vehicles awaiting service or being served shall be parked or stacked behind the building line. Entrance and exit driveways shall be not over fifteen (15) feet wide each between the street line and the building line, shall have adequate radii for junction with existing travelled ways, and shall be separated by a landscaped median not less than eight (8) feet wide.
- B. All front yards shall be at least twenty-five (25) feet in depth from the street line and covered with a natural landscape material such as turf, ground cover or stone or slate paving, but not bituminous material. Each street yard shall have Type "A" screening. (Ref. Sec. 6.1(D))
- C. Landscape strips shall contain Type "C" screening. (Ref. Sec. 6.1(D))
- D. All operations, except final hand polishing, shall be confined within the building.
- E. All waste water and liquids used in car wash operations shall be collected by a self-contained treatment system on the property. Said system shall be so designed as to remove all sand and other solids from the used wash water. In addition, if wash water is to be discharged into a public sewer, approval of the Water Pollution Control Authority (WPCA) shall be required. Said self-contained treatment system shall be subject to approval by the Commission or its designated agent.

## 7.9 Pistol, rifle and skeet trap shooting: Available upon request.

## 7.10 Forestry-Repealed 8-16-18 see Wetlands Regulations

## 7.11 **Day care**

Group day care homes and child day care centers as defined by CGS 8-3 and nursery schools and adult day care facilities may be allowed as a special permit use by the Commission after public hearing and subject to the following:

- A. The size of the property shall be such that off-street parking will be accommodated; outside activity areas will be provided; and on-site pick-up and discharge are provided.
- B. The street serving the proposed facility shall be adequate to provide a free flow of two-way traffic that will not create congestion.
- C. Driveways serving the pick-up/drop-off point shall be wide enough to allow two one-way lanes for at least thirty (30) feet in each direction from the subject point.
- D. Appropriate screening of parking facilities shall be provided.

- E. Proper lighting shall be provided without being a nuisance to neighboring properties.
- F. All state and local fire and health codes shall be continuously met during the period of the permit.

#### 7.12 "Adult" bookstores

May be allowed as a special permit use by the Commission after public hearing and subject to the following:

- A. The center of the main entrance of the "adult" bookstore is not closer than one thousand, five hundred (1,500) feet (measured in a straight line) from the center of the main entrance of any other "adult" bookstore or the nearest property line of any lot used as a place of worship, a public library, a day care facility, a community recreation building, a playground, a sports field, a park, and a school (as defined by standards recommended by the State Board of Education and State Department of Education). Said distance requirement may be waived by a two-thirds (2/3) vote of the Commission provided that the applicant has submitted a petition, at or before the public hearing, signed by the owners of fifty-one percent (51%) of the residences and commercial establishments within a one thousand, five hundred (1,500) foot distance of the proposed location and provided that the Commission makes the following findings:
  - 1. The proposed use will not be contrary to the public interest or injurious to nearby properties, and the spirit and intent of this regulation will be observed;
  - 2. The establishment of an additional use of this type in the area will not be contrary to any program of neighborhood conservation or improvement, either residential or non-residential.
- B. Advertisements, displays, or other promotional materials shall not be shown or exhibited so as to be visible to the public from sidewalks or walkways or from other public and semi-public areas; and such displays shall be considered signs.
- C. All building openings, entries, windows, etc., for "adult" bookstores shall be located, covered, or screened in such a manner as to prevent viewing into the interior from any public or semi-public area.
- D. All public floor areas within the "adult" bookstore shall be adequately lighted and constructed so that every portion thereof is readily visible to the clerk or other supervisory personnel from the counter or other regular station. In order to protect the public health and safety, booths for previewing or viewing films, videos, etc., whether coin-operated or otherwise are prohibited. Nothing herein is intended to prohibit the uncompensated presentation of films or videos in such a manner as to be visible from the public floor areas
- E. Applications under this section shall be in accordance with the relevant subsections of Section 8.11 of these Regulations and the approval of any such application shall be for a period not to exceed one (1) year.

- F. In the event that at any time it shall appear to the Zoning Enforcement Officer that an existing bookstore whose stock in trade was previously unrestricted to minors has become an "adult" bookstore, said premises will be subject to special use approval under the conditions of this section.
- G. In the event that at any time it shall appear to the Zoning Enforcement Officer that an "adult" bookstore has become a nuisance or hazard, he/she shall notify the Commission. The Commission, after public notice and hearing, may take whatever action it deems appropriate to abate the nuisance or hazard including revoking the special use permit.

# 7.13 Alcoholic Beverages:

Applicability – The provisions of this regulation shall apply to the following classes of permit as described in Secs. 30-15 to 30-37h inclusive, of the Connecticut General Statutes:

- 1. Café permit.
- 2. Club permit, or nonprofit club permit and brewpub.
- 3. Hotel permit, or hotel permit for beer only.
- 4. Package store permit.
- 5. Restaurant permit, restaurant permit for beer only, restaurant permit for wine and beer only, or restaurant permit for catering establishment.
- 6. Special sporting facility bar permit, special sporting facility concession permit, special sporting facility employee recreational permit, special sporting facility guest permit, or special sporting facility restaurant permit.
- 7. Tavern permit.
- 8. Bowling establishment permit, bowling establishment permit for beer only, or racquetball facility permit.
- 9. Temporary sales/tasting permit not to exceed four (4) days may be permitted by the Zoning Enforcement Officer
- 10. Manufacturers permit.
- A. **Permits:** Use of a building or other premises for the sale or dispensing of alcoholic beverages shall require the issuance of a special permit by the Commission in accordance with the Zoning Regulations of the Town of Stafford, Special Use Permit Section 8.11. For a special permit under this Regulation, the Commission may consider the following in addition to the standards of the Town of Stafford's Zoning Regulations:
  - 1. The proximity of the establishment to schools, religious places of worship, residential neighborhoods, and charitable institutions such that, in the judgment

- of the Commission, the trade associated with the establishment does not disrupt the quiet pursuit of education and religion.
- 2. The proximity of the establishment to an establishment operating under another permit listed in the Alcoholic Beverages Applicability previous such that, in the judgment of the Commission, there not be created a cluster of undue concentration of establishments where the sale or consumption of alcoholic beverages may be perceived as a dominant characteristic of the neighborhood or area.

#### 7.14 Flood Plain District:

There is hereby established a Flood Plain District. The Flood Plain District includes all areas designated as Zone A and Zones A1-A30 on maps entitled "Flood Insurance Rate Map, Town of Stafford, Connecticut, Tolland County Map Index Panels Printed: 15, 18, 20, 22, 24, 25, 26, 30, Community Panel Numbers 090152-0001-0030" and "090113-0001B (Stafford Springs – Only Panel Printed)." These maps as well as the Flood Boundary and Floodway Maps and accompanying Flood Insurance Studies are incorporated by reference. In any area designated as a Flood Plain District, all regulations in this section will apply, together with all of the regulations for the zone of said property as shown on the Zoning Map of the Town of Stafford.

## A. Purpose and Objectives:

It is the purpose of this ordinance to promote the public health, safety, and general welfare and to minimize public and private losses due to flood conditions in specific areas. The flood hazard areas of Stafford are subject to periodic inundation that results in loss of life and property, health and safety hazards, disruption of commerce and governmental services, extraordinary public expenditures for flood protection and relief, and impairment of the tax base, all of that adversely affect the public health, safety, and general welfare.

#### **B. Definitions:**

Unless specifically defined below, words or phrases used in this ordinance shall be interpreted so as to give them the meaning they have in common usage and to give this ordinance its most reasonable application:

- 1. **Area of special flood hazard** is the land in the flood plain within a community subject to one percent (1%) or greater chance of flooding in any given year.
- 2. **Base flood** means the flood having a one percent (1%) chance of being equaled or exceeded in any given year.
- 3. **Basement** means any area of the building having its floor sub-grade below ground level on all sides.
- 4. **Development** means any man-made change to improved or unimproved real estate, including but not limited to buildings or other structures, mining, dredging,

- filling, grading, paving, excavation, or drilling operation or storage of equipment or materials.
- 5. **Flood Boundary** and **Floodway Map** means an official map of a community on which the Federal Emergency Management Agency has delineated the boundaries of the floodway.
- 6. **Flood Insurance Rate Map** (FIRM) means an official map of a community, on that the Federal Emergency Management Agency has delineated both the areas of special flood hazard and the risk premium zones applicable to the community.
- 7. **Flood Insurance Study** (FIS) means the official report from the Federal Emergency Management Agency (FEMA) that contains examination, evaluation, and determination of flood hazards and, if appropriate, corresponding water surface elevations.
- 8. **Floodway** means the channel of a river or other watercourse and the adjacent land areas that must be reserved in order to discharge the base flood without cumulatively increasing the water surface elevation more than one (1) foot.
- 9. **Lowest floor** means the lowest floor of the lowest enclosed area (including basement). An unfinished or flood resistant enclosure, usable solely for parking of vehicles, building access, or storage, in an area other than a basement area is not considered a building's lowest floor.
- 10. **Manufactured home** is a structure, transportable in one or more sections, that is built on a permanent chassis and designed to be used with or without a permanent foundation when connected to the required utilities. The term also includes recreational vehicles and similar transportable structures placed on a site for one hundred eighty (180) consecutive days or longer and intended to be improved property.
- 11. **Mean sea level** means, for purposes for the National Flood Insurance Program, the National Geodetic Vertical Datum (NGVD) of 1929 or other datum, to which base flood elevations shown on a community's Flood Insurance Rate Map are referenced.
- 12. **New construction** means structures for that the "start of construction" commenced on or after the effective date of the initial FIRM (latest edition) and includes any subsequent improvements to such structures.
- 13. **Recreational vehicle** means a vehicle that is: a. built on a single chassis; b. four hundred (400) square feet, or less when measured at the longest horizontal projections; c. designed to be self-propelled, or permanently towable by a light duty truck; and d. designed primarily not for use as a permanent dwelling, but as temporary living quarters for recreational, camping, travel, or seasonal use.
- 14. **Start of construction** includes substantial improvement and means the date the building permit was issued, provided the actual start of construction, repair, reconstruction, or improvement was within one hundred and eighty (180) days of the permit date. The actual start means the first placement of permanent

construction of a structure (including a manufactured home) on a site, such as the pouring of slabs or footings, installation of piles, construction of columns, or any work beyond the stage of excavation or placement of a manufactured home on a foundation. Permanent construction does not include land preparation, such as clearing, grading and filling; nor does it include the installation of streets and/or walkways; nor does it include excavation for a basement, footings, piers, or foundations, or the erection of temporary forms; nor does it include the installation on the property of accessory buildings, such as garages or sheds not occupied as dwelling units or not part of the main structure. For a substantial improvement, the actual start of construction means the first alteration of any wall, ceiling, floor, or other structural part of a building, whether or not that alteration affects the external dimensions of the building.

- 15. **Structure** means a walled and roofed building, including a gas or liquid storage tank that is principally above ground, as well as a manufactured home.
- 16. **Substantial improvement** means any combination of repairs, reconstruction, alteration, or improvements to a structure taking place over a one (1) year period in that the cumulative cost equals or exceeds fifty percent (50%) of the market value of the structure. The market value of the structure should be: a. the appraised value of the structure using the cost approach to value method prior to the start of the initial repair or improvement; or b. in the case of damage, the value of the structure prior to the damage occurring. For the purposes of this definition, "substantial improvement" is considered to occur when the first alteration of any wall, ceiling, floor, or other structural part of the building commences, whether or not that alteration affects the external dimensions of the structure. The term does not, however, include any project for improvement of a structure required to comply with existing health, sanitary, or safety code specifications that are solely necessary to assure safe living conditions.
- 17. **Water surface elevation** means the height, in relation to the National Geodetic Vertical Datum (NGVD) of 1929, (or other datum, where specified), of floods of various magnitudes and frequencies in the flood plain of coastal, or riverine areas.

## C. General Provisions:

These regulations shall apply to all areas of special flood hazard within the jurisdiction of Stafford. The areas of special flood hazard identified by the Federal Emergency Management Agency in its Flood Insurance Studies (FIS), or Flood Insurance Rate Maps (FIRM), latest edition, with accompanying floodway maps and other supporting data, and any revisions thereto, are adopted by reference and declared to be a part of these regulations. A development permit shall be required in conformance with the provisions of these regulations prior to the commencement of any development activities.

#### **D. Provisions for Flood Hazard Reduction:**

1. **General standards:** In all areas of special flood hazard, the following provisions are required:

- a. New construction and substantial improvements: New construction and substantial improvements shall be: a. anchored to prevent flotation, collapse, or lateral movement of the structure; b. constructed with materials resistant to flood damage; c. constructed by methods and practices that minimize flood damage; d. constructed with electrical, heating, ventilation, plumbing, air conditioning equipment, and other service facilities that are designed and/or located so as to prevent water from entering or accumulating within the components during conditions of flooding.
- b. Water supply and sanitary sewage systems: New and replacement water supply systems shall be designed to minimize or eliminate infiltration of flood waters into the system. New and replacement sanitary sewage systems shall be designed to minimize or eliminate infiltration of flood waters into the systems and discharges from the system into flood waters. On-site waste disposal systems shall be located and constructed to avoid impairment to them or contamination from them during flooding.
- c. **Manufactured homes:** All manufactured homes to be placed within Zone A on a community's FIRM shall be installed using methods and practices that minimize flood damage. For the purposes of this requirement, manufactured homes must be elevated and anchored to resist flotation, collapse, or lateral movement. Methods of anchoring may include, but are not to be limited to, over-the-top or frame ties to ground anchors.
- 2. **Specific standards:** In all areas of special flood hazard where base flood elevation data has been provided, as set forth in Section D, or as determined in Section G.4 the following provisions, in addition to those in Section C are required:
  - a. **Residential structures:** New construction or substantial improvement of any residential structure shall have the lowest floor, including basement, elevated to or above the base flood elevation.
  - b. **Non-residential structures:** New construction or substantial improvement of any commercial, industrial, or non-residential structure located in a special flood hazard area shall have the lowest floor, including basement, elevated to, or above the level of the base flood elevation.
- 3. **Flood Proofing:** Non-residential structures located in all "A" Zones may be flood proofed (to one (1) foot above the base flood elevation) in lieu of being elevated provided that, together with all attendant utilities and sanitary facilities, the areas of the structure below the required elevation are watertight with walls substantially impermeable to the passage of water and use structural components having the capability of resisting hydrostatic and hydrodynamic loads and the effect of buoyancy. A registered professional engineer or architect shall review and/or develop structural design, specifications, and plans for the construction and shall certify that the design and methods of construction are in accordance with acceptable standards of practice for meeting the provisions of this subsection.

- 4. **Enclosed areas below base flood elevation:** New construction or substantial improvement of buildings that include fully enclosed areas formed by foundation and other exterior walls below the base flood elevation shall be designed to preclude finished living space and designed to allow for entry and exit of floodwaters to automatically equalize hydrostatic flood forces on exterior walls. Designs for complying with this requirement must either be certified by a professional engineer or architect or meet the following minimum criteria:
  - a. Provide a minimum of two (2) openings having a total net area of not less than one (1) square inch for every square foot of enclosed area subject to flooding;
  - b. The bottom of all openings shall be no higher than one (1) foot above grade;
  - c. Openings may be equipped with screens, louvers, valves, or other coverings or devices provided they permit the automatic flow of floodwaters in both directions.

#### E. Administration:

In addition to the other information that must be provided in an application for a zoning permit, a development permit shall be required in conformance with the provisions of these regulations prior to the commencement of any development activities. The Zoning Enforcement Officer (Z.E.O.) is hereby appointed to administer and implement the provisions of these regulations. Any applicant for a permit in the Flood Plain District shall provide the following information to the Z.E.O:

- 1. The elevation in relation to mean sea level of the lowest floor, including the basement, of all new or substantially improved structures;
- 2. The elevation in relation to mean sea level to that any new or substantially improved structure is flood proofed;
- 3. Description of the extent to that any watercourse will be altered or relocated as a result of proposed development;
- 4. A statement as to whether or not proposed alterations to an existing structure meet the criteria of the substantial improvement definition;
- 5. Zones A1-30 and AE require, until a regulatory floodway is designated, that no development may increase flood levels more than one (1) foot at any point.

## F. Manufactures Homes:

- 1. All manufactured homes to be placed or substantially improved shall be elevated so that the lowest floor is above the base flood elevation.
- 2. Manufactured homes shall be placed on a permanent foundation that itself is securely anchored and to that the structure is securely anchored so that it will resist flotation, lateral movement and hydrostatic, and hydrodynamic pressures. Anchoring may include, but not be limited to, the use of over-the-top or frame ties to ground anchors. Specific requirements shall be that:

- a. Over-the-top ties be provided at each of the four (4) corners of the manufactured home, with two (2) additional homes less than fifty (50) feet long requiring one (1) additional tie per side;
- b. Frame ties be provided at each corner of the home with five (5) additional ties per side at intermediate points and manufactured homes less than fifty (50) feet long requiring four (4) additional ties per side;
- c. All components of the anchoring system be capable of carrying a force of 4,800 pounds;
- d. Any additions to the manufactured home be similarly anchored.
- 3. Manufactured homes are prohibited in the floodway.
- 4. All manufactured homes where the repair, reconstruction, or improvement of streets, utilities, and pad are less than fifty percent (50%) of the value before the repair, reconstruction, or improvement has commenced shall have the lowest floor elevated to or above the base flood level. They shall be placed on a permanent foundation that itself is securely anchored and to that the structure is securely anchored.

## G. Duties and Responsibilities of the Zoning Enforcement Officer:

The duties of the Zoning Enforcement Officers shall include, but not be limited to:

- 1. Review all permit applications to determine whether proposed building sites will be reasonably safe from flooding.
- 2. Advise permittee that additional federal or state permits may be required. Notify adjacent communities and the Department of Environmental Protection, Water Resources Unit prior to any alteration, or relocation of a watercourse. Assure that maintenance is provided within the altered or relocated portion of said watercourse so that the flood-carrying capacity is not diminished.
- 3. Record the "as built" elevation (in relation to mean sea level) of the lowest flood (including basement) of all new or substantially improved or flood-proofed structures. When flood proofing is utilized for a particular structure, the Zoning Enforcement Officer shall obtain certification from a registered professional engineer or architect.
- 4. When base flood elevation data or floodway data have not been provided, then the Zoning Enforcement Officer shall obtain, review, and reasonably utilize any base flood elevation and floodway data available from a federal, state, or other source including data developed pursuant to Sec. 4.7.5 of the Subdivision Regulations in order to administer the provisions of Sections 7.15(D) and 7.15(G).
- 5. Where interpretation is needed as to the exact location of boundaries of the areas of special flood hazard, the Zoning Enforcement Officer shall make the necessary interpretation. All records pertaining to the provision of this ordinance shall be maintained in the office of the Zoning Enforcement Officer.

- 6. In "A" zones, where base flood elevations have been determined, but before a floodway is designated, require that no new construction, substantial improvement, or other development (including fill) be permitted that would increase base flood elevations more than one (1) foot at any point along the watercourse when all anticipated development is considered cumulatively with the proposed development.
- 7. In areas where floodways have been designated or determined: Prohibit encroachments, including fill, new construction, substantial improvements, and other developments unless certification (with supporting technical data) by a registered professional engineer is provided demonstrating that encroachments shall not result in any (0.00) increase in flood levels during occurrence of the base flood discharge. When utilizing data other than that provided by the Federal Emergency Management Agency, the following standard applies: Select and adopt a regulatory floodway based on the principle that the area chosen for the regulatory floodway must be designed to carry the waters of the base flood, without increasing the water surface elevation of that flood more than one (1) foot at any one point.

#### H. Variance Procedures:

The Stafford Zoning Board of Appeals shall hear and decide appeals and requests for variances from the requirements of these regulations.

- 1. Variances shall not be issued within any designated floodway if any increase in flood levels during the base flood discharge would result.
- 2. Variances shall only be issued upon a determination that the variance is the minimum necessary, considering the flood hazard, to afford relief; and in the instance of a historical building, a determination that the variance is the minimum necessary as not to destroy the historic character and design of the building.
- 3. Variances may only be issued upon: a. a showing of good and sufficient cause; b. a determination that failure to grant the variance would result in exceptional hardship, and; c. a determination that the granting of a variance will not result in increased flood heights, additional threats to public safety, extraordinary public expense, create a nuisance, cause fraud on or victimization of the public, or conflict with existing local laws, or, ordinances. Only hardships that are based on unusual physical characteristics that are not shared by adjacent parcels shall qualify to meet subsection b. above. Claims of hardship based on the structure, on economic, or on personal circumstances are not sufficient cause for the granting of a variance under these regulations.
- 4. Any applicant to whom a variance is granted shall be given written notice specifying the difference between the base flood elevation and the elevation to that the structure is to be built and stating that the cost of flood insurance will be commensurate with the increased risk resulting from the lowest floor elevation up to amounts as high as twenty-five dollars (\$25) for one hundred dollars (\$100) of insurance coverage.

 The Zoning Enforcement Officer shall maintain the records of all appeal actions and report any variances to the Federal Emergency Management Agency upon request.

## 7.15 Soil Erosion and Sedimentation Control Regulations

The Planning and Zoning Commission shall either determine that the soil erosion and sediment control plan, as filed, complies with the requirements and objectives of this section, or deny any special permit, or permit when the development proposal does not so comply. Prior to approval, the Commission may submit any proposed Soil Erosion and Sediment Control Plan to the Tolland County Soil and Water Conservation District for its recommendations.

The Planning and Zoning Commission may, as a condition of approval, require that the applicant post a performance bond or other security acceptable to the Treasurer of the Town of Stafford, to ensure that the applicant complete construction in accordance with the Soil Erosion and Sediment Control Plan submitted to the Commission.

Construction of any project approved in accordance with those standards shall be in accordance with the following:

- Site development shall not begin unless the soil erosion and sediment control plan is approved and those control measures and facilities in the plan scheduled for installation prior to site development are installed and functional.
- Planned soil erosion and sediment control measures and facilities shall be installed as scheduled according to the certified plan.
- All control measures and facilities shall be maintained in effective condition to ensure the compliance of the certified plan.
- Inspections shall be made by the authorized agent of the Planning and Zoning
  Commission during development to ensure compliance with the plan and that
  control measures and facilities are properly performed or installed and
  maintained. The Planning and Zoning Commission may require the applicant to
  verify through progress reports that soil erosion and sediment control measures
  and facilities have been performed or installed according to the plan and are being
  operated and maintained.

# A. No Permit, or special use permit shall be issued in accordance with these Regulations unless:

- 1. The permit is for a single-family residence not being built by a developer as part of a subdivision of land;
- 2. The Planning and Zoning Commission or its authorized agent has determined that the ground cover to be disturbed as a result of the permitted activity is less than one-half ( $\frac{1}{2}$ ) acre; or

The Planning and Zoning Commission has determined that the permitted activity
will be conducted in accordance with a Soil Erosion and Sediment Control Plan
designed to minimize soil erosion and sedimentation resulting from the permitted
activity.

## **B. Application Requirements:**

- 1. A narrative describing:
  - a. The development;
  - b. The schedule for grading and construction activities including:
    - i. Start and completion dates;
    - ii. Sequence of grading and construction activities;
    - iii. Sequence for installation and/or application of soil erosion and sediment control measures;
    - iv. Sequence for final stabilization of the project site;
  - c. The design criteria for proposed soil erosion sediment control measures and storm water management facilities;
  - d. The construction details for proposed soil erosion and sediment control measures and storm water management facilities;
  - e. The installation and/or application procedures for proposed soil erosion and sediment control measures and storm water management facilities;
  - f. The operations and maintenance program for proposed soil erosion and sediment control measures and storm water management facilities.
- 2. A site plan map at a sufficient scale to show:
  - a. The location of the proposed development and adjacent properties;
  - b. The existing and proposed topography, including soil types, wetlands, watercourses, and water bodies;
  - c. The existing structures on the project site, if any;
  - d. The proposed area alterations including cleared, excavated, filled, or graded areas and proposed structures, utilities, roads, and, if applicable, new property lines;
  - e. The locations of and design details for all proposed soil erosion and sediment control measures and storm water management facilities;
  - f. The sequence of grading and construction activities;
  - g. The sequence for installation and/or application of soil erosion and sediment control measures;
  - h. The sequence for final stabilization of the development site.

3. Any other information deemed necessary by the Planning and Zoning commission, or its authorized agent.

# C. Application Compliance:

No permit or special use permit shall be approved for any activity to that these Regulations are applicable unless it complies with the following:

- 1. Plans for soil erosion and sediment control shall be in accordance with the principles outlined in Chapters 3 and 4 of the *Connecticut Guidelines for Soil Erosion and Sediment Control* (1985), as amended. Soil erosion and sediment control plans shall result in a development that minimizes erosion and sedimentation during construction is stabilized and protected from erosion when completed and does not cause off-site erosion and/or sedimentation.
- 2. The minimum standards for individual measures are those in the *Connecticut Guidelines for Soil Erosion and Sediment Control* (1985), as amended. The Commission may grant exceptions when requested by the applicant if technically sound reasons are presented.
- 3. The appropriate method from Chapter 9 of the *Connecticut Guidelines for Soil Erosion and Sediment Control* (1985), as amended, shall be used in determining peak flow rates and volumes of runoff unless an alternative method is approved by the Commission.

## 7.16 **Telecommunications Towers** (Effective 5/1/2000)

#### A. Definitions:

For the purpose of applying the provisions of this section the terms below shall be defined as follows:

- 1. **Antenna** means a device used to receive or transmit electromagnetic waves. Examples include, but are not limited to, whip antennas, panel antennas, and dish antennas.
- 2. **Co-location** means locating wireless communication facilities from more than one provider on a single site.
- 3. Commercial wireless telecommunication services mean a licensed commercial wireless telecommunication services including, but not limited to cellular, personal communication services (PCS), specialized mobilized radio (SMR), enhanced specialized mobilized radio (ESMR), and paging and similar services that are marketed to the general public.
- 4. **Commercial wireless telecommunication site** means a facility that consists of the equipment and structures involved in receiving or transmitting electromagnetic waves associated with wireless telecommunication services.
- 5. **Height of tower** means the vertical distance measured in feet from the ground elevation of such tower to the topmost point of the tower including any antenna or other appurtenances.

6. **Tower** means a structure that is intended to support equipment used to receive and/or transmit electromagnetic waves. Design examples of towers include: self-supporting lattice, guyed, and monopole.

#### **B. Regulations:**

- 1. Public and private telecommunications facilities, communications towers, antennae, and accessory equipment shall be permitted used in all zones by Special Use Permit subject to the requirements of Section 8.11 of these regulations and the following specific criteria.
- 2. No wireless telecommunication tower shall be constructed or used except as specifically provided in this section.
- 3. These regulations cover towers used for communications (both receiving and transmitting) and other such purposes. Towers may be freestanding ground mounted (with guys or self-supporting) or attached to or on the roof of a building (also with guys or self-supporting). Poles and towers for power transmission regulated by public utilities are exempted from these zoning regulations.
- 4. The applicant will be required to take reasonable steps to mitigate any adverse visual impact from all new telecommunications facilities. Steps include landscaping, fencing, painting, or similar measures as appropriate to camouflage the tower.
- 5. A tower must comply with the setback requirements of the zone in that it is located or be set back from property lines a distance equal to the height of the tower, whichever is greater.
- 6. Towers shall not exceed one hundred and eighty (180) feet in height above the ground.
- 7. To discourage unauthorized trespassing and provide for the public safety, the base of any ground mounted tower shall be secured by fence enclosure to a height of six (6) feet above the ground.
- 8. No lighting of any telecommunications towers will be permitted, with the exception of ground lighting for maintenance purposes, except as required by the Federal Communications Commission, Federal Aviation Administration, or the Connecticut Siting Council.
- 9. No advertising or signs shall be permitted on any telecommunications towers.
- 10. To avoid the unnecessary proliferation of antenna towers, new antenna towers for cellular and personal communication services will not be approved unless all available space on existing, or approved antenna towers covering the same geographic area has been utilized.
- 11. To protect the public health from the unknown effects of electromagnetic fields, all cellular and personal communication service transmitters must comply with FCC emissions regulations.

- 12. The Commission may condition, restrict, or limit any special permit issued for a telecommunications facility, communication tower, or antennae in any manner consistent with federal law.
- 13. Abandonment: A wireless telecommunication facility not in use for twelve (12) consecutive months shall be removed by the facility owner at their expense. This removal shall occur within ninety (90) days of the end of such twelve (12)-month period. The Commission may require a bond, or other surety satisfactory to the Town of Stafford, to guarantee removal, that shall be reviewed and renewed every two (2) years. If there are two, or more users of a single tower, this provision shall not become effective until all users cease utilizing the tower.
- 14. No wireless telecommunication site shall be located within two hundred (200) feet of a residence.
- 15. No commercial wireless telecommunication site shall be located within five hundred (500) feet of a playground, school, or daycare facility property line.
- 16. In the case where a wireless telecommunication site is proposed to be located on, or within one thousand (1000) feet of a property designated on the National Historic Register or historic district that such proposal will preserve the historic and/or architectural character of the landscape, or any structure.
- 17. All utilities proposed to serve a wireless telecommunication site shall be installed underground unless otherwise approved by the Commission.
- 18. All generators installed in conjunction with any wireless telecommunication site shall comply with Section 5.5 of the Town Zoning Regulations.

#### 7.17 **Age-restricted Housing** (Effective 6/11/08)

#### A. Applicability and Purpose:

- 1. Age-restricted Housing Developments may be permitted subject to the issuance of a special use permit in accordance with the requirements of Section 8.1 of these regulations within the AAA, AA, and A zoning districts.
- 2. The purpose of these regulations shall be to provide for the development of housing for residents fifty-five (55) and over in a comprehensively planned manner that will provide a superior residential environment in conformance with the sound and, orderly development of the Town.

#### **B.** Location Criteria:

Age Restricted Housing developments may be permitted by Special Use Permit on parcels of land meeting the following locational criteria as well as the requirements specified in Section 7.15b.

- 1. The site is at least five (5) acres in area;
- 2. The site shall have a minimum of two hundred (200) feet of frontage on a public road.

#### C. General Criteria:

The Commission shall approve an age-restricted housing development only when it finds that the following criteria are met:

- 1. That the location and size of the proposed use and intensity of use in relation to the size of the site will be in harmony with the orderly development of the area and compatible with other existing uses.
- 2. The proposed plans provide for the conservation and protection of natural features.
- 3. The design elements of the proposed development are attractive and conducive to the creation of a neighborhood residential environment.
- 4. The streets that provide access to the proposed use are adequate in width, grade, alignment, and visibility and have adequate capacity for the additional traffic generated by the proposed use.
- 5. The proposed use shall have easy accessibility for fire apparatus and police protection and is designed and equipped to further the provision of emergency services.
- 6. The proposed use will not have any detrimental impacts upon the public health, safety, or welfare and not conflict with the purpose of these regulations.
- 7. The proposed use conforms to all the requirements of Section 7.18(D) of these regulations.

## **D.** Age Restriction:

- 1. Occupancy of all dwelling units within Age-restricted Housing Development shall comply with the requirements of the "55 and over housing" exemption as set forth in the Fair Housing Amendments Act (42 U.S.C., Sec 3601 et. Seq.), the Housing for Older Persons Act of 1955, and in accordance with Federal Law.
- 2. The "housing facility or community" shall provide to the Zoning Enforcement Officer each year, on the anniversary of the approval of the site plan, documentation regarding age verification of the occupants of all dwelling units sufficient for the Zoning Enforcement Officer to determine there is compliance with the requirements of the "over 55 housing" exemption.
- 3. Such age restrictions shall be enforced by governing document and restrictive covenant, as reviewed and approved by the Town Attorney and shall not be subject to revocation.
- 4. Further, by governing document and restrictive covenant, that shall be binding upon all occupants of all dwellings in the development parcel, the applicant shall agree and covenant so that no children under the age of eighteen (18) years shall be permitted to reside in any dwelling, except by hardship exception granted by the Board of Directors or other governing body of the community. At least fifteen (15) days prior to the hearing concerning such hardship, the Board of Directors shall notify the Zoning Enforcement Officer, by certified mail, of the date, time, and place of such hearing and make available all appropriate information. Such

hardship exception shall be granted only for children, or grandchildren of an existing occupant, provided that visitor occupants of any ages shall be permitted to visit for up to four (4) weeks of any calendar year. In the event the Board of Directors or other governing body of the "housing facility community" shall grant a hardship exception, a copy of the application for the exception and of the Board's decision thereon shall be forwarded to the Stafford Zoning Enforcement Officer within thirty (30) days after such decision is made.

## E. Building Size and Coverage

- 1. There shall be a maximum of four dwelling units within each building.
- 2. No building shall exceed a height of thirty-five (35) feet.
- 3. No building shall be located within fifty (50) feet of any lot line.
- 4. The maximum building coverage shall not exceed twenty-five percent (25%) of the total area of the site.
- 5. Maximum impervious surface coverage shall not exceed thirty-five percent (35%) of the total area of the site.

#### F. Parking

- 1. There shall be a minimum of two (2) off street parking spaces per dwelling unit, with at least one (1) space located within an enclosed garage, plus adequate visitor parking as determined by the Commission.
- 2. All parking areas shall be a minimum of fifty (50) feet from all property lines, except driveways and access ways that provide access to the parking areas or the dwelling units.

#### G. Circulation

- 1. All interior roadways and sidewalks shall remain in private ownership in perpetuity. The owner(s) successors or assigns of the development shall, by formal agreement, agree to keep all internal roadways properly maintained, open, and passable at all times.
- 2. All interior roadways shall be suitably paved with a minimum width of twenty-four (24) feet, except that the Commission may permit at its discretion one-way roads of sixteen (16) feet in width if it finds that it will result in less environmental disruption of the land, be consistent with protection of public safety, and provide adequate and safe traffic circulation.
- 3. A pedestrian circulation system shall be provided within the development to provide safe access to all dwelling units and facilities. The system shall consist of the following:
  - a. Sidewalks along one side of the major streets within the development elsewhere, as determined by the Commission. The Commission may permit alternative pedestrian facilities, where it finds it will provide alternative pedestrian circulation.

- b. Where suitable, a walking trail of suitable surfacing material (i.e., stone dust, wood) shall be provided in locations that give access to the surrounding environmental resources. When applicable, this trail system shall be accessible to people of all abilities.
- c. The Commission may require that on-site or off-site pedestrian circulation facilities be installed to link the proposed development with other uses within the community.
- 4. The minimum grade for all roads shall be consistent with the standards of the Town of Stafford subdivision regulations.

#### H. Utilities

- 1. The development shall be served by a public water system; or the Commission may permit water service to be supplied by a private system, if it finds that it would be beneficial to the Town and consistent with the protection of public health.
- 2. The Development shall be served by the existing and established public sanitary sewer system, owned and operated by the Stafford Water Pollution Control Authority or The Commission may permit on-site waste water disposal if it finds that it would be consistent with beneficial patterns of development and would protect public health and preserve the quality of the water resources of the Town.
- 3. Where a system of on-site wastewater disposal is to be utilized, the Commission shall require sufficient legal and financial assurance to provide for the proper operation and maintenance of the system and assure that there be no additional costs by the Town of Stafford for maintenance, repair, or replacement of the system. Such shall include deed covenants and stipulations within the association documents that all costs of maintenance, repair, and/or replacement of such systems incurred by the Town shall be paid for by the property owners association. The Town Attorney shall review and advise the Commission as to the specific requirements of such financial and legal assurances.
- 4. All electrical, cable, telephone, and other service utilities shall be placed underground.
- 5. All storm water management systems will be included that incorporates all best management practices so that there are no negative impacts upon surrounding property or nearby watercourses.
- 6. All utilities shall conform to the applicable sections of the Subdivision Regulations and the Town of Stafford standards.

#### I. Design

1. The physical layout of the proposed development shall be designed to foster a neighborhood-oriented development.

- 2. The proposed development shall include areas for use by the residents for active and passive recreational purpose. These areas can include clubhouses and recreational facilities such as tennis courts or bocce courts.
- 3. The architecture of the buildings within the proposed development shall reflect the traditional New England village concept, as determined by the Commission. The Commission may, at its discretion, permit an alternative architectural theme, providing that they find that it is compatible with the surrounding neighborhood environment.

## J. Open Space

- 1. There shall be a minimum of thirty-five percent (35%) of the total area of the site preserved as open space to enhance and preserve the rural character of the Town of Stafford.
- 2. Open space shall be undisturbed areas, left in their natural state, except for walking trails and other passive recreation features.
- 3. Land designated as open space shall be set aside in accordance with the plan submitted and in the manner proposed by the applicant and as approved by the Commission, in accordance with one of the following methods as deemed appropriate by the Planning and Zoning Commission.
  - a. Ownership by a non-profit land trust or other conservation, organization, or the Town of Stafford.
  - b. Ownership by a homeowner's association, with a conservation easement in favor of the Town of Stafford.
  - c. If the land is transferred, all zoning compliance determinations shall be based upon the parcel size prior to the time of the application, and prior to the transference of open space.

#### K. Ownership

- 1. The development shall remain under common ownership. Where the plan proposes creation of common ownership interests subject to the Common Ownership Interest Act (COIA), Chapter 82B of the Connecticut General Statutes shall be reviewed by the Town Attorney prior to recording on the land records.
- 2. All streets, walkways, recreational facilities, utilities, lighting, parking areas, and landscaping within the proposed development shall remain in private ownership. The continued ownership and private maintenance of all internal streets and utilities shall be enforced by covenants, notations on the land records, inclusion within the common area declarations, and all other measures deemed appropriate by the Commission.

## L. Preliminary Site Development Concept Plan

It is recommended that, prior to the submission of an official application for approval, the applicant initiate a pre-application conference with the Commission and subsequently prepare and present a preliminary plan for informal consideration by the

Commission. The preparation of the preliminary plan is recommended to facilitate the general consideration of factors and problems affecting the development of the land before the applicant proceeds with the official application and the preparation of final maps, plans, and documents required for formal consideration by the Commission. The presentation of a preliminary plan will more readily and economically facilitate alterations and changes recommended by the Commission.

Neither the pre-application conference nor the formal consideration of the preliminary plan shall be deemed to constitute any portion of the official and formal procedure of applying for a Preliminary Site Development and Land Use Concept Plan approval. Neither the proponent nor the Commission shall be in any way bound by statements made in such informal discussions, their purpose being only to minimize delay, expense and inconvenience to the public, the proponent, and the Commission upon the future receipt, if any, of a formal application. Following any informal discussion, the Commission may suggest that the proposal, or certain aspects thereof, be referred to other municipal, State, or Federal agencies for review and comment, or Federal agencies for review and comment or may suggest that additional information is or will be required.

## M. Accessory Uses

The following uses may be permitted as accessory to the age-restricted housing development at the discretion of the Commission.

- i. Clubhouse, including health and fitness facility, athletic facilities, swimming pool, and meeting rooms.
- ii. Tennis courts, bocce courts, and other recreational facilities.

## 7.18 **Solar Energy Systems** (Effective 08-11-2015)

A. **Purpose**: The purpose of this subsection is to provide for the regulation of the construction and operation of Solar Energy Facilities in the Town of Stafford subject to reasonable conditions that will protect the environment, public health, safety, and welfare.

#### **B. Definitions:**

- 1. **Solar Energy System, Large:** A solar energy collection system that can be used to serve all or part of the electric load at one or more properties and consumers.
- 2. **Solar Energy System, Roof-Mounted:** A solar collection system, that is installed upon or is part of the roof of a building or structure located on the subject property. Systems integrated as awnings or attached to the roofs of porches, sheds, carports, and covered parking structures also fall under this distinction.
- 3. **Solar Energy System, Small:** An accessory solar energy collection system or photovoltaic system.

#### C. Small-Scale Solar Energy Systems

Small solar energy systems shall be a permitted as an accessory use by right in all zoning districts subject to the requirements set forth in this section: Solar energy systems include ground, pole, and roof-mounted systems.

1. The construction of the small solar energy system shall be in accordance with an approved zoning permit and building permit application.

#### 2. Setback:

- a. In residential districts ground- or pole-mounted small-scale solar energy systems shall be placed so that no individual component is closer than the height of the array from the rear lot line or side lot lines. Ground or pole-mounted solar energy system shall only be allowed in the rear or the side yard behind the front building line.
- b. In zones LB and CB ground- or pole-mounted small-scale solar energy systems shall be placed so that no individual component of the solar system may extend into the front, side, or rear setback for the district. Ground mounted small-scale solar systems shall be screened from adjoining residential districts by arborvitae or similar evergreen hedge planted six feet on center. The Commission may allow additional or alternative screening methods such as berms and opaque fencing when it is determined that such alternatives are more appropriate for that site.
- c. In zones HB, HI and HB, ground- or pole-mounted small-scale solar energy systems shall be placed so that the minimum distance to the side or rear setback is equal to or greater than the height of any individual component of the solar system, including any mounts. Ground mounted small-scale solar systems shall be screened from adjoining residential districts by arborvitae or similar evergreen hedge planted six feet on center. The Commission may allow additional or alternative screening methods such as berms and opaque fencing when it is determined that such alternatives are more appropriate for that site.

### D. Ground-Mounted Small Solar Energy Systems:

- 1. The total height of the solar energy system including any mounts shall not exceed the maximum height of accessory structures permitted in that district.
- 2. The system shall meet all applicable current building codes.

## E. Roof-Mounted Small Solar Energy Systems:

Roof-mounted small solar energy systems shall include integrated solar shingles tiles or panels as the surface layer of the roof structure with no additional apparent change in relief or projection (the preferred installation) or separate flush or rack-mounted solar panels mechanically fastened to and/or secured with ballast on the roof surface in conformance with applicable current building codes.

1. Separate flush or rack-mounted small solar energy systems installed on the roof of a building or structure shall not project vertically more than eight (8) feet above a flat roof installation.

2. It shall be demonstrated that the placement of the system shall not adversely affect safe access to the roof, pathways to specific areas of the roof, and safe egress from the roof.

## F. Appearance:

- 1. Appearance, color, and finish. The small solar energy system shall remain painted or finished the color or finish that was originally applied by the manufacturer.
- 2. All signs, other than the manufacturer's or installer's identification, appropriate warning signs, or owner identification on a small solar energy system shall be prohibited. Installer and/or developer are limited to one (1) sign indicating their role in the system installation.
- 3. The maximum area of signage is one (1) square foot.
- 4. No signage shall pertain to advertising.

## G. Code Compliance:

A small solar energy system shall comply with all applicable current Building codes.

#### H. Removal:

All obsolete or unused systems shall be removed within twelve (12) months of cessation of operations without cost to the Town.

#### I. Violation:

Subsequent to the effective date of this ordinance, it is unlawful for any person to construct, install, or operate a small solar energy system that is not in compliance with this chapter or with any condition contained in a building permit issued pursuant to this chapter.

### J. Large Scale Energy Systems

Any system greater than 100 kW AC shall require a special use permit.

## 1. Compliance with Laws, Ordinances and Regulations

The construction and operation of all large scale Solar Energy System shall be consistent with all applicable current local, state, and federal requirements, including but not limited to all applicable safety, construction, electrical, and communications requirements.

## 2. Building Permit and Building Inspection

No large scale Solar Energy System shall be constructed installed or modified as provided in this section without first obtaining a zoning permit and building permit.

### 3. Site Plan Review

Ground-mounted large scale Solar Energy System with 100 kW AC or larger of rated nameplate capacity are allowed in Business and Industrial Districts and on lots 20 (twenty) acres or more in size in residential districts subject to a Special Use Permit

by the Stafford Planning and Zoning Commission prior to construction, installation, or modification as provided in this section

#### a. General

All plans and maps shall be prepared, stamped, and signed by a Professional Engineer licensed to practice in Connecticut.

## b. Required Documents

Pursuant to the site plan review process, the project proponent shall provide the following documents in addition to the Site Plan submission requirements of Section 8.3:

- Blueprints or drawings of the Solar Energy System signed by a Professional Engineer licensed to practice in Connecticut showing the proposed layout of the system and any potential shading from nearby structures.
- ii. Manufacturer's data sheets or similar documentation of the major system components to be used, including the PV panels, mounting system, and inverter.
- iii. Full contact information, including name, address, phone number, and e-mail address for proposed system installer.
- iv. Name, address, phone number, and signature of the project proponent, as well as all co-proponents or property owners, if any.
- v. The name, contact information, and signature of any agents representing the project proponent; and
- vi. Documentation of actual or prospective access and control of the project site.
- vii. An operation and maintenance plan.
- viii. Zoning district designation for the parcel(s) of land comprising the project site.
- ix. Proof of liability insurance.

The Stafford Planning and Zoning Commission may waive documentary requirements that it finds are unnecessary to determine compliance with these regulations, as it deems appropriate.

- 4. **Site Control:** The project proponent shall submit documentation of actual or prospective access and control of the project site sufficient to allow for construction and operation of the proposed Solar Energy System.
- 5. **Operation and Maintenance Plan:** The project proponent shall submit a plan for the operation and maintenance of the large-scale, ground-mounted Solar Energy System, which shall include measures for maintaining safe access to the

installation, storm water controls, as well as general procedures for operational maintenance of the installation.

## 6. Dimension and Density Requirements

#### a. Setbacks

For large – scale ground-mounted Solar Energy System, front, side and rear setbacks shall be as follows:

- i. Front yard: The front yard depth shall be at least 10 feet; provided, however, where the lot is across from a Residential district, the front yard shall not be less than 50 feet.
- ii. Side yard. Each side yard shall have a depth at least 25 feet; provided, however, where the lot abuts a Residential district, the side yard shall not be less than 50 feet.
- iii. Rear yard. The rear yard depth shall be at least 20 feet; provided, however, where the lot abuts a Residential district, the rear yard shall not be less than 50 feet
- 7. Accessory Structures: All accessory structures to large- scale ground-mounted Solar Energy System shall be subject to the underlying zoning requirements concerning the bulk and height of structures, lot area, setbacks, open space, parking and building coverage requirements. All such accessory structures including, but not limited to, equipment shelters, storage facilities, transformers, and substations shall be architecturally compatible with each other. Whenever reasonable, structures should be shaded from view by vegetation and/or joined or clustered to avoid adverse visual impacts as determined by the Commission.

### 8. Design Standards

### a. **Lighting**

Lighting of large-scale Solar Energy System shall be consistent with local, state, and federal law. Lighting of other parts of the installation, such as accessory structures, shall be limited to that required for safety and operational purposes, and shall be reasonably shielded from abutting properties. Lighting of the Solar Energy System shall be directed downward and shall incorporate full cut-off fixtures to reduce light pollution.

## b. Signage

Signs on large-scale ground-mounted Solar Energy System shall comply with the Sign regulations. A sign consistent with the regulations shall be required to identify the owner and provide a 24-hour emergency contact phone number.

### c. Utility Connections

Reasonable efforts, as determined by the Stafford Planning and Zoning Commission, shall be made to place all wiring from the Solar Energy System underground, depending on appropriate soil conditions, shape, and topography

of the site and any requirements of the utility provider. Electrical transformers, inverters, and switchgears for utility interconnections may be above ground.

### d. Screening

A ground mounted large solar energy system shall be screened from adjoining residential districts by arborvitae or similar evergreen hedge planted six feet on center. The Commission may allow additional or alternative screening methods including existing vegetation when it is determined that such alternatives are more appropriate for the particular site. The Commission may also allow fencing greater than six (6) feet in height where deemed appropriate.

### 9. Safety and Environmental Standards

## a. Emergency Services

The large scale Solar Energy System owner or operator shall provide a copy of the project summary, electrical schematic, and site plan to the local fire marshal. Upon request, the owner or operator shall cooperate with local emergency services in developing an emergency response plan. All means of shutting down the Solar Energy System shall be clearly marked. The owner or operator shall identify a responsible person for public inquiries throughout the life of the installation.

## b. Land Clearing and Soil Erosion Impacts

Clearing of natural vegetation shall be limited to what is necessary for the construction, operation, and maintenance of the large—scale ground-mounted Solar Energy System or otherwise prescribed by applicable laws, regulations, and bylaws

### 10. Monitoring and Maintenance

### a. Solar Energy System Conditions

The large-scale ground-mounted Solar Energy System owner or operator shall maintain the facility in good condition. Maintenance shall include, but not be limited to, painting, structural repairs, and integrity of security measures. Site access shall be maintained to a level acceptable to the local Fire Marshal and Emergency Medical Services. The owner or operator shall be responsible for the cost of maintaining the Solar Energy System and any access road(s), unless accepted as a public way.

### 11. Abandonment or Decommissioning

### a. Removal Requirements

Any large- scale ground-mounted Solar Energy System that has reached the end of its useful life or has been abandoned consistent with section 5.60.2.12 b of this regulation shall be removed. The owner or operator shall physically remove the installation no more than 150 days after the date of discontinued

operations. The owner or operator shall notify the Stafford Planning and Zoning Commission by certified mail of the proposed date of discontinued operations and plans for removal. Decommissioning shall consist of:

- i. Physical removal of all large- scale ground-mounted Solar Energy System, structures, equipment, security barriers, and transmission lines from the site.
- ii. Disposal of all solid and hazardous waste in accordance with local, state, and federal waste disposal regulations.
- iii. Stabilization or re-vegetation of the site as necessary to minimize erosion. The Stafford Planning and Zoning Commission may allow the owner or operator to leave landscaping or designated below-grade foundations in order to minimize erosion and disruption to vegetation.
- iv. A stabilization/re-vegetation plan shall be submitted along with the Site Plan application.

#### b. Abandonment

Absent notice of a proposed date of decommissioning or written notice of extenuating circumstances, the Solar Energy System shall be considered abandoned when it fails to operate for more than one year without the written consent of the Stafford Planning and Zoning Commission. If the owner or operator of the large- scale ground-mounted Solar Energy System fails to remove the installation in accordance with the requirements of this section within 150 days of abandonment or the proposed date of decommissioning, the town may enter the property and physically remove the installation.

## 7.19 Accessory Agricultural Activities

- A. Accessory Agricultural Activities (Ag-tivities) The following fee-based activities are considered to be accessory to an established agricultural operation. A Site Plan Review shall be required when the following uses may reasonably be expected to require parking for twenty (20) or more motor vehicles, used by visitors to the farm, at any given time. Otherwise, such uses require only an Agricultural Activity "Agtivity" Permit.
  - Agricultural Engagement- The opening to the public of a working farm or any
    agricultural, horticultural, or agribusiness operation for the purpose of enjoyment,
    education, or active involvement in the activities of the farm or operation,
    including (but not limited to) "U-Pick", Community Supported
    Agriculture (CSA), Wine tours & associated tastings, agricultural workshops,
    classes, and demonstrations.
  - 2. **Agricultural Events-** Events of limited duration on a farm that are incidental to agricultural uses, including (but not limited to) hayrides, corn mazes, festivals, petting zoos, and other similar activities.

- B. **Non-agriculturally Related Uses-** Fee-based activities that are part of an agricultural operation's overall offering, but are not incidental to agriculture, are allowed by permit. Such uses may include (but not limited to) outdoor recreation such as bird watching, snow-shoeing, cross country skiing, mountain biking, and other passive recreational activities, as well as rental of farm property for small and large-scale events, such as parties, weddings, and other non-agricultural events.
  - 1. "Small-scale Farm Events" are allowed with an Agricultural Activity "Agtivity" Permit. Only low-volume amplified sound or music (as determined by the ZEO) will be allowed during these events. A Site Plan Review will only be required if these events happen more than ten (10) times per year, or they require parking for more than twenty (20) vehicles, at any given time.
  - 2. "Large-scale Farm Events" at which the use of loud amplified music is planned, and more than twenty (20) vehicles are expected, require a Special Permit. In order to satisfy the conditions of a Special Permit, the Planning & Zoning Commission may limit the number of events per year. The Planning & Zoning Commission may also impose additional restrictions on the use if it determines, after a hearing for which the owner of the farm is given written notice that the noise levels, or other factors, are causing a nuisance to nearby properties. The ZEO retains the right to meet annually with the farm owner to discuss any possible concerns. In the case of events when more than twenty (20) vehicles are expected, but loud amplified music is not a consideration, only a Site Plan Review is required.

## r. General Requirements

- 1. Parking. Parking for all permitted agricultural accessory uses shall, to the maximum extent possible, be located in areas on the site where they will be the least visible from access roads and adjoining properties but must allow for adequate sightlines for drivers traveling on the adjacent road and entering and exiting such spaces from such road. Setback requirements apply to all parking areas. Due consideration shall be given to the posted speed limit and other pertinent factors. The ZEO and/or Commission may require additional buffering to meet the intent of the parking and landscaping regulations.
- 2. **Location, Lighting and Noise**. The location of outdoor events and activities associated with "Ag-tivities" on the farm shall take into consideration the current use of surrounding properties. The Commission and/or ZEO may require a specific separating distance and/or an appropriate buffer strip that screens any such activity from adjacent properties. If the proposed activity or agricultural use regularly employs the use of an amplified sound system, there shall be a sound-

absorption plan in place that confirms compliance with the State of Connecticut DEEP noise standards (§22a- 69-1 through §22a 69-7-4) as amended. All lighting must be shielded so that it does not impact neighboring properties. The Commission may impose additional restrictions on the use if it determines, after a hearing for which the owner of the farm is given written notice, the noise levels are causing a nuisance to nearby properties.

### 7.20 Recreational Club

An applicant proposing to establish an outdoor recreational club on private land shall comply with the following:

The recreational club shall be a not for profit corporation.

The applicant shall submit a statement describing:

- a. The proposed use in detail including at least: rules; hours of operation; and maximum number of users.
- b. The effect the proposed facility will have on the town's infrastructure
- c. A site plan in accordance with section 8.3
- d. A maintenance plan of the facility

The Commission may impose conditions that are deemed necessary to protect the welfare of the community, such as, but not limited to the following:

- a. The effect of the use on the surrounding areas;
- b. Requirements of front, side, or rear yards greater than the minimum required by the regulations;
- c. Requirements of screening from adjoining premises or from the streets, by walls, fences, plantings, or other devices:
- d. Limitation on size, number of occupants, method or time of operation, extent of facility, outdoor activity or noise.
- e. Regulation of number, design, and location of access drives or other traffic flow features

The Commission may modify or waive by resolution any item of the required submission under this section if it finds that the information is not necessary in order to decide on the application, or if it does not affect the public health and welfare.

## 7.21 Medical Marijuana

### 1. Producer

Licensed medical marijuana producers (producer), when and where permitted by these Regulations, shall conform to the following standards in addition to other applicable standards of these Regulations.

- A. The property on which a Producer is located shall not be permitted within 500 feet of any portion of a property on which a duly organized school, municipal park/recreation facility. Place of worship is located; or within 1,000 feet of a dwelling.
- B. The property on which a Producer is located shall not be permitted within 1,000 feet of any portion of a property on which another Producer is located.
- C. Marijuana and Marijuana paraphernalia and products made thereof shall not be clearly visible from the exterior of the production facility.
- D. Producers shall comply the Connecticut General Statues Section 21a-408 et seq. and the Regulations of State Agencies Sections 21a-408-1 to 21a-408-70 inclusive, both of which as may be amended from time to time.

## 8. Dispensary

Licensed medical marijuana dispensaries (dispensary), when and where permitted by these Regulations, shall conform to the following standards in addition to other applicable standards of these Regulations.

- A. Marijuana and marijuana paraphernalia and products made thereof shall not be clearly visible from the exterior of the dispensary facility.
- B. Dispensaries shall comply with Connecticut General Statutes Section 21a-408 et seq. and the Regulations of State Agencies 21a-408-1 to 21a-408-70 inclusive, both of which as may be amended from time to time.

### 7.22 Recreational Cannabis Establishments

**Statement of Purpose**. The purpose of this section is to regulate the location and operation of recreational cannabis establishments in accordance with state statutes. The intent is to minimize any adverse impacts of such facilities and to protect and preserve the town's neighborhoods, commercial districts, property values and quality of life.

### 1. Definitions.

- a. Cannabis. Marijuana as defined in Section 21a-240, C.G.S.
- b. Cannabis Establishment. Producer, dispensary facility, cultivator, microcultivator, retailer, hybrid retailer, food and beverage manufacturer, product manufacturer, product packager and or delivery service.
- c. Cultivator. A person that is licensed to engage in the cultivation, growing and propagation of the cannabis plant at an establishment with not less than fifteen thousand square feet of grow space.
- d. Delivery Service. A person that is licensed to deliver cannabis from (A) micro-cultivators, retailers and hybrid retailers to consumers and research program subjects, and (B) hybrid retailers and dispensary facilities to

- qualifying patients, caregivers and research program subjects, as defined in Section 21a-408, C.G.S., or to hospices or other inpatient care facilities licensed by the Department of Public Health pursuant to Chapter 368v, C.G.S. that have a protocol for the handling and distribution of cannabis that has been approved by the department, or a combination thereof.
- e. Dispensary Facility. Means a place of business where cannabis may be dispensed, sold or distributed in accordance with Chapter 420f, C.G.S. and any regulations adopted thereunder, to qualifying patients and caregivers, and to which the department has issued a dispensary facility license under Chapter 420f, C.G.S. and any regulations adopted thereunder.
- f. Food and Beverage Manufacturer. A person that is licensed to own and operate a place of business that acquires cannabis and creates food and beverages.
- g. Hybrid Retailer. A person that is licensed to purchase cannabis and sell cannabis and medical marijuana products.
- h. *Micro-cultivator*. A person licensed to engage in the cultivation, growing and propagation of the cannabis plant at an establishment containing not less than two thousand square feet and not more than ten thousand square feet of grow space, prior to any expansion authorized by the commissioner.
- *i.* Person. An individual, partnership, limited liability company, society, association, joint stock company, corporation, estate, receiver, trustee, assignee, referee or any other legal entity and any other person acting in a fiduciary or representative capacity, whether appointed by a court or otherwise, and any combination thereof.
- *j. Product Manufacturer.* A person, excluding a producer, that is licensed to obtain cannabis, extract and manufacture products exclusive to such license type and who may sell or transfer cannabis and cannabis products to laboratories, research programs and cannabis establishments.
- *k. Product Packager.* A person that is licensed to package and label cannabis and cannabis products.
- l. Retailer. A person, excluding a dispensary facility that is licensed to purchase cannabis and cannabis products from producers, cultivators, product manufacturers and food and beverage manufacturers and to sell cannabis and cannabis products to consumers and research programs.
- *m.* Transporter. Means a person licensed to transport cannabis between cannabis establishments, laboratories and research programs.

3. Adult -Use Cannabis and Recreational Marijuana Sales and Facilities Adult -Use Cannabis and Recreational Marijuana Sales and Facilities shall be permitted subject to the following:

Cannabis retail and hybrid-retail facilities shall be governed by The Responsible and Equitable Regulation of Adult-Use Cannabis Act ("RERACA"), the Public Act 21-1 / Senate Bill 1201 as amended inclusive of the Regulations of Connecticut State Agencies as they may be amended and permitted in the specified zone, subject to special permit.

Adult-use cannabis cultivator and micro-cultivator facilities shall be governed by the RERACA, the Public Act 21-1/ Senate Bill 1201 as amended inclusive of the Regulations of Connecticut State Agencies as they may be amended and permitted in the specified zones, subject to special permit approval.

## 2. Separation Requirements.

- i. No medical marijuana production, adult-use cannabis cultivator, micro-cultivator facility, dispensary, or retail or hybrid-retail facility, shall be allowed within 500 feet of a church, temple or other place used primarily for religious worship, public building, private recreation area, or a school, playground, park or child day care facility.
- ii. No medical marijuana production, adult-use cannabis cultivator, micro-cultivator facility, dispensary, shall be allowed on a site that is less than 500 feet from any residential dwelling structure.
- iii. No medical marijuana production, adult-use cannabis cultivator, micro-cultivator facility, dispensary, or retail or hybrid-retail facility, shall be allowed within the same building, structure or portion thereof that is used for residential purposes, or that contains another medical marijuana dispensary, production facility, or adult use cannabis retail, hybrid retail, cultivator or micro-cultivator facility.
- iv. No adult-use cannabis retail or hybrid-retail shall be located less than 1,500 feet from another adult-use cannabis retail or hybrid-retail. Distance shall be measured from the radius of the front door to front door of each establishment.
- v. All distances contained in this section, other than those specified in the subsection "iv" above shall be measured by taking the nearest straight line between the respective lot boundaries of each site.
- vi. No adult-use cannabis retail or hybrid-retail shall operate beyond the hours of 8 am to 10 pm Monday through Saturday and 10 am to 6 pm on Sundays.

#### 7.23 Lawn Care Business

A Minor Lawn Care Business shall conform to the definition in Section 2.1 and is further defined as: being conducted on property having not less than three (3) acres (1) Not having more than three (3) Commercial Vehicles stored upon the property in connection with business operations; and (2) Not having more than two (2) persons other than primary residents of the dwelling starting and finishing their workday from the premises in connection with business operations. Maximum of two employee parking spaces.

A Major Lawn Care Business shall conform to the definition in Section 2.1 and is further defined as: (1) being conducted on property having not less than ten (10) acres; (2) Having more than three (3) but not more than six (6) Commercial Vehicles stored upon the property in connection with business operations; and (3) Having more than two (2) but not more than four (4) persons other than primary residents of the dwelling starting and finishing their workday from the premises in connection with business operations. Maximum of four employee parking spaces.

The following conditions are applicable to both Minor and Major Lawn Care Businesses:

- 1. All Commercial Vehicles, employee parking and equipment stored on the property in connection with business operations shall be (a) stored inside; or (b) located at least 50 feet from all property lines, located to the side or rear of the dwelling, and screened with Type "C" screening so that they are not visible from adjacent residentially zoned properties or public streets:
- 2. No Commercial Vehicles or equipment may be idled for more than 10 minutes, except for maintenance purposes; and
- 3. Commercial Vehicles and equipment shall not be moved from the premises before 700:AM or later than ½ hour before sunset each day except in the event of a storm or emergency.
- 4. No processing of trees or stumps generated from business removal operations shall occur on the premises where the business is operated; and
- 5. Sales of equipment from the premises where the business is operated is prohibited; and
- 6. The operator of the lawn care business shall provide written documentation, certified under oath, on or before January 31 of each year, to the Zoning Enforcement Officer, attesting to the number of Commercial Vehicles maintained on the premises where the business is operated and the number of non-resident employees starting and finishing their workday from the premises where the business is operated.
- 7. No Wall mounted or free-standing signs advertising the business are permitted

## **Article VIII**

## Administration and Enforcement

## 8.1 Zoning permits:

No structure shall be erected, expanded, or structurally altered and no land use shall be established until a permit has been issued as required below:

- A. A special use permit shall be in accordance with the procedure set forth in Section 8.11;
- B. A use requiring site plan approval shall be in accordance with the procedure set forth in Section 8.3;
- C. All other permitted uses in accordance with the procedure set forth in Section 8.25.

Additionally, zoning permits for all structures or additions exceeding four hundred (400) square feet shall be valid for the building foundation only until such time as a building foundation as-build by a licensed land surveyor licensed in the State of Connecticut has been submitted and approved by the Zoning Enforcement Officer. Such as-build shall indicate the location of the building foundation in relations to all property lines within fifty (50) feet, except that the Zoning Enforcement Officer may waive this request if, in his sole discretion, he determines that all zoning regulations are satisfied.

## 8.2 Certificate of zoning compliance

- A. No structure or land or any part thereof shall hereafter be occupied or used or changed to another use until a certificate of zoning compliance shall have been issued by the Zoning Enforcement Officer certifying that the Zoning Regulations have been complied with in full, including any limitations or conditions attached to any special use permit, site plan approval, or variance for the property covered by the certificate of zoning compliance. The Zoning Enforcement Officer shall issue or deny a certificate of zoning compliance within ten (10) days after a request for the same shall have been filed in his/her office by an owner or his/her agent.
- B. The Zoning Enforcement Officer may issue a temporary certificate of zoning compliance for the period up to one (1) year, if only a portion of a structure or land use is completed at the time the request for a certificate is made; or if minor non-compliances exist at such time that can be corrected without impairing the use of the structure of land. In this case, the Zoning Enforcement Officer shall list all non-compliances and set a date by that they shall be corrected. A temporary certificate of zoning compliance may be renewed for periods up to six (6) months if sufficient cause has been shown to the Zoning Enforcement Officer to warrant such an extension.
- C. A certificate of zoning compliance issued for the structure or use subject to a variance permitted by the Zoning Board of Appeals shall include a description of such variance and any conditions under that such variance was granted and shall further specify whether or not such variance constitutes a nonconformity.

- D. A certificate of zoning compliance issued for a structure or use permitted as a special permit use after approval by the Commission shall include any conditions under that such special permit use was granted.
- E. When a proposed change, extension, enlargement, or alteration of land, buildings, or premises as to use or area or construction does not require a permit, a certificate of zoning compliance shall be applied for before such proposed changes are commenced.
- F. A certificate of zoning compliance will be issued within ten (10) days after completion and compliance with all applicable regulations, and record of all such certificates shall be kept on file in the office of the Zoning Enforcement Officer who will furnish certified copies upon payment of the required fee to any person having a proprietary or tenancy interest in the building, or premises affected.
- G. The Zoning Enforcement Officer may require evidence of continuing compliance from the owner of all structures and uses for that certificates of zoning compliance have been issued. The Zoning Enforcement Officer shall make periodic inspections of all structures and land uses for that permits have been issued under site plan approval or special use permit to determine continuing compliance. If the owner refuses to furnish such evidence, or if compliance with said permits has not continued, the Zoning Enforcement Officer shall take action according to Section 8.14 of these Regulations and, in addition thereto, may revoke said permits and certificates upon fifteen (15) days written notice to said owner or permit holder provided, however, said owner or permit holder may appeal said revocation order to the Zoning Board of Appeals within said fifteen (15) days and said revocation order shall be stayed until the decision of the Zoning Board of Appeals.

## 8.3 Site Plan Application Requirements:

An application for a zoning permit shall include a description of all proposed uses and four (4) prints of a site plan showing the subject parcel and abutting properties within one hundred (100) feet. Plans submitted for approval and subsequent recording shall be drawn on stable, transparent and reproducible material, at a scale of one inch equals 40 feet (1" = 40') or larger. Sheet size maximum is twenty-four inches by thirty-six inches (24" x 36"). The plan shall indicate the following:

- A. Name and address of owner-of-record and developer, scale of drawing, date of preparation, or revision; and, if a lot in a subdivision, a reference to the original subdivision plan. Abutting property owner names shall be indicated. Where a public hearing is required, abutters' mailing addresses shall also be submitted.
- B. A boundary survey of the lot. Where work is to be done on only a small portion of a parcel, then only that portion need normally be shown.
- C. A key map at a scale of not more than one inch equals one thousand feet (1" = 1,000') shall be included in order to assist in locating the property. Zoning boundaries shall be indicated thereon.
- D. The location, size, and height of all existing and proposed structures above and below the ground.

- E. The existing and proposed contours at two-foot (2) intervals or less, if any change in grading is proposed.
- F. The present and proposed use of all existing and proposed structures, if other than a single-family residence.
- G. The layout of existing and proposed parking and loading facilities and access thereto, including any parking barriers or walkways. Surface treatments to be specified.
- H. Location and details of proposed sanitary disposal system, whether by sewer hookup or subsurface disposal area, along with percolation test and deep pit data, if applicable. Said soil tests shall be approved by the Stafford Health District.
- I. Where surface or subsurface drainage facilities are to be constructed or modified, existing and proposed structures and piping with flow line elevations shall be shown.
- J. Location of existing and proposed water supply either by private well or tap from public water.
- K. The type and location of any exterior floodlighting or spotlighting, including provisions for shading.
- L. The type, size, and location of all signs where applicable.
- M. The type and location of any required and/or proposed screening or landscaping.
- N. Location of existing watercourses, marshes, wooded areas, ledge outcrops, single trees with a diameter of ten (10) inches, or more, measured three (3) feet above the base of the trunk with an indication of whether or not they are to be retained.
- O. In the case of uses requiring approvals by other agencies, such prior approvals shall be submitted.
- P. An erosion and sedimentation control plan and narrative shall accompany all applications where a proposal will disturb more than one-half (½) acre of land.
- Q. Space for review, comments, and signatures of pertinent officials.
- R. Such additional information as required by the Zoning Enforcement Officer or the Commission, where a special use permit or site plan approval is required, or where it is necessary to determine that the requirements of the Zoning Regulations are met. The Zoning Enforcement Officer or the Commission may excuse compliance with requirements for specific information otherwise required on the plot plan where such compliance is not necessary to determine that zoning or other code regulations are met.
- S. Two (2) copies of dimensional plans of floors and elevations of any proposed structure, and specifications to indicate the size, kind, and quality of the proposed construction.

## 8.4 Changes and Modifications to an Approved Site Plan or Special Use Permit:

A. The ZEO with the Chairman's concurrence may approve the following types of minor modifications or changes to a previously approved Site Plan or Special Use Permit:

- 1. Minor modifications and/or minor building renovations that will not significantly impact natural resources; lighting, drainage, traffic generation and pattern; parking demand; buildings or other structures; aesthetics or landscaping; buffers, community character, or quality of life.
- 2. Change from one conforming use to another conforming use that will not significantly impact natural resources; lighting, drainage, traffic generation and pattern; parking demand; buildings, or other structures; aesthetics or landscaping; buffers, community character, or quality of life.

Copies of approved minor modifications will be reported to the Commission at its next regular meeting. If the ZEO and the Chairman do not agree as to the impact of the modification, the request will be submitted to the commission for review.

Modifications to a previously approved Site Plan or Special Use Permit shall not be granted for proposals that in the opinion of the ZEO and Chairman will result in significant impacts to natural resources; lighting, drainage, traffic generation and pattern; parking demand; buildings, or other structures; aesthetics or landscaping; buffers, community character, or quality of life.

If impacts are deemed significant, the modification will be presented to the commission for review. The commission may deem the modification minor and allow it or approve it after a public hearing or request a new application.

- C. When requesting a minor modification, the applicant shall provide a written description of all proposed changes. A revised Site Plan drawing depicting all modifications shall be provided, unless it is determined that there are sufficient plans on file with the Commission demonstrating that no major site alterations are proposed.
- D. Any expansion of a use, or change of a use listed in these Regulations as a Special Use shall require a new Special Use Permit application for the project.

## 8.5 Time period of permits:

A zoning permit shall become void unless construction is commenced within six (6) months from the date of issuance unless such time shall have been extended in writing by the Zoning Enforcement Officer.

### 8.6 One- and two-family dwellings:

With the exception of one-family dwellings in WM zones, plot plans for one- and two-family dwellings may be limited to those requirements under section 8.3 above, which the Zoning Enforcement Officer shall determine as applicable.

## 8.7 Waiver of requirements:

The filing of building plans required in section 8.3 above and/or the submission of a plot plan required in section 8.5 above may be waived when the proposed work is of a minor nature or repairs provided the scope of the work is adequately described in the application.

## **8.8 Planning Review:** Effective May 1, 2000

- A. The ZEO shall review and act upon applications, not part of a special use permit, requiring site plan approval, or the ZEO may require that the site plan be submitted to the Commission for review and action.
- B. Upon approval by the Commission or ZEO, the applicant shall cause to be filed with the town clerk the approved transparent print of the site plan.

## 8.9 Applications:

All applications for required permits shall be made to the Zoning Enforcement Officer on forms provided by his/her office.

#### 8.10 Performance bond:

An owner may be required to post a bond in an amount sufficient to cover the cost of construction of street improvements, drainage, sewer and water supply, landscaping, and any other conditions required before a permit is issued. The performance bond shall not be released until the bonded work is completed to the satisfaction of all appropriate agencies.

## 8.11 Procedure for approval of applications:

Applications for required permits shall be reviewed and acted upon as follows:

### A. Special Use Permit

- 1. All special permit uses are declared to possess characteristics of such unique and distinct form that each specific use shall be considered as an individual case. Special permit uses shall be deemed to be permitted uses in their respective districts, subject to the satisfaction of the requirements and standards set forth therein, in addition to all other requirements of the Zoning Regulations.
- 2. The Zoning Enforcement Officer shall refer applications for special use permits to the Commission for receipt, hearing, and action as required by Sec. 8-2 of the Connecticut General Statues as may be amended from time to time.
- 3. The applicant shall post a sign, available from the Zoning Enforcement Officer, giving notice of the application, in a conspicuous place on the property for that a special use permit approval is sought, visible from a public street. Said sign shall be posted ten (10) days before the public hearing, and it shall be removed not later than three (3) days after the close of the public hearing.

- 4. The Commission shall make a finding within sixty-five (65) days of public hearing that each of the following standards is met and, where necessary, shall attach specific conditions to its approval of the special use permit, if in its opinion such conditions are essential to making the finding.
- 5. The location and size of the use, the nature and intensity of the operations connected with it, the size of the lot in relation to it, and the location of the lot with respect to streets giving access to it are such that it will be in harmony with the appropriate and orderly development of the district in that it is located.
- 6. The kind, location, and height of all structures and the nature and extent of the landscaping on the lot are such that the use will not hinder or discourage the appropriate development and use or impair the value of adjacent properties.
- 7. The pedestrian, parking, and loading facilities are adequate and properly located for the proposed use and the entrance and exit driveways are laid out so as to achieve maximum safety.
- 8. The proposed uses permit the development of the site without the destruction of valuable natural assets or pollution of lakes, streams, and other water bodies while providing the best possible design of structures and land uses compatible with the shape, size, and topographic and natural character of the site. (Effective 6/11/08)
- 9. The proposed development will be compatible with the existing and future character of the neighborhood in that the development is to be located. Particular attention shall be paid to the type and density of adjacent residential development, the character and use of existing highway facilities, and the Plan of Conservation and Development. (Effective 6/11/08)
- 10. The location and character of buildings shall create a harmonious grouping and shall be compatible with surrounding structures. (Effective 6/11/08)
- 11. The proposed development shall not create additional traffic not consistent with the character of the community nor any safety hazards or congestion upon the road system. Any improvements to public roads necessitated by the proposed development shall be the responsibility of the developer. (Effective 6/11/08)
- 12. The proposed development shall be of a superior site and architectural design that enhances the surrounding neighborhood environment. (Effective 6/11/08)
- 13. The proposed water supply facilities and facilities for the disposal of sanitary waste shall have sufficient capacity to accommodate the needs of the proposed development, maintain the environmental qualities of the community, and contain safeguards that would not have a negative impact upon the fiscal resources of the Town. (Effective 6/11/08)
- 14. After the approval, the applicant shall submit to the Zoning Commission five (5) prints and one (1) transparency of the plans as approved. The Commission shall file with the Zoning Enforcement Officer one (1) print and the transparency of the approved plan with the approval noted thereon and list of all conditions pertaining to

- the special use approval. One (1) print and said list shall be made available to the applicant.
- 15. Special Permit Requirements: An application for a special use permit for a commercial or industrial use with twenty thousand (20,000) square feet of gross floor area, or greater, or age restricted housing (Effective 6/11/08) shall include all information required for Site Plan application, as required by Section 8.3, plus the following: (Effective 4/1/06)
  - a. A statement indicating the specific types of proposed uses on the site.
  - b. The proposed timetable for development, including a description of phases, if any.
  - c. The pattern/method of ownership and maintenance of any interior roadways, public facilities, the sewerage disposal system(s), the water supply system(s), and other common elements.
  - d. A statement outlining how the proposed development conforms to the Comprehensive Plan embodied in these Regulations and the adopted Plan of Conservation and Development of the Town and how the proposal will utilize the resources of the site to the benefit of the neighborhood and the Town.
  - e. The methods by which site utilities will be provided; a statement outlining how each of the requirements set forth in these Regulations is met and how each of the criteria for evaluation of the application is satisfied.
  - f. *Key Map*. A key map at a scale of one (1) showing the location of the proposed development and its relationship to existing Town roads.
  - g. Adjacent Land Uses. The boundaries of the subject parcel to be developed, owners of these parcels and adjacent parcels, roadways, structures, and land uses within five hundred (500) feet of the boundaries of the parcel.
  - h. *Existing Site Features*. Existing structures, roads, land uses, topography at a contour interval of five (5) feet, or less, major and unique natural, scenic, historic, and open space features of the parcel and their relationship to the proposed development.
  - i. *Proposed Land Uses*. The proposed density of land uses intended for different parts of the parcel, including the number of buildings, and the amount of land to be devoted to each land use.
  - j. *Proposed Buildings*. The general height, bulk, use, and location of buildings.
  - k. *Circulation:* The proposed location of roads, parking, and pedestrian circulation including tie-ins with existing Town facilities and the impact of the proposed use on traffic, including the impact on public roads. For any development that includes the construction of one hundred (100) parking spaces, or twenty thousand (20,000) square feet of gross floor area, the study should be prepared by a traffic engineer and shall include at a minimum, data, and information concerning existing and potential impact of the proposed development upon

daily traffic (ADT) on town roads and state highways; peak hour traffic; adequacy of right of way and travel way widths; proposed traffic generation and distribution of proposed development; location of road cuts and driveways within three hundred (300) feet of the development; pedestrian circulation including safety considerations; and site lines of all intersections likely to be significantly impacted by the proposed development. If the Commission determines that the proposed project is likely to cause adverse impacts upon the pedestrian or traffic circulation or render any transportation facility inadequate for projected use, whether on Town roads adjacent to, on site, or off-site; the Commission may require that the developer make improvements to mitigate the impacts of the proposed development.

- 1. *Restrictions*. The substance of any proposed covenants, easements, restrictions, and organizations.
- m. *Water Supply*. The proposed method of supplying potable water to the development.
- n. *Sewage Disposal*. The proposed method for the collection and disposal of all sanitary waste.
- o. *Storm Water*. The proposed storm water handling concept including possible utilization of detention, aquifer recharge, sediment control, irrigation, and fire protection storage.
- p. *Surface Water Quality*. A statement indicating the quality of existing watercourses through or near the site.
- q. *Erosion Control*. A statement indicating the erodibility of the soils and a general indication of the need for erosion and sedimentation control.
- r. *Noise*. The expected intensity and frequency of noise.
- s. Fiscal Impact. The anticipated impact of the proposed development upon municipal revenues and expenditures, including projected expenditures for increased services.
- t. *Soils*. A detailed soil survey for the parcel prepared by the Conservation Service or its equivalent prepared by a qualified soils scientist.
- u. *Watercourses*. The location of any inland wetland and watercourse as defined by the Stafford Inland Wetlands and Watercourses Regulations, as amended.
- v. *Scheduling*. A general schedule of development in terms of time and site development area for all proposed phases.
- w. *Restrictions*. The substance of any proposed covenants. Easements, restrictions, and organizations.
- x. *Further Documentation*. Other documentation as may reasonably be required by the Commission.

- y. Architectural renderings of proposed buildings, as well as description of proposed architectural theme.
- z. For all applications that include storm water management facilities, water supply wells, sewers, or on-site wastewater treatment and disposal systems serving units located within five hundred (500) yards of a lake, pond, river, stream, or other water-dependent environmental resource, the applicant shall conduct an environmental impact study that would include testing and analysis to demonstrate that the proposed development would not cause adverse impacts to the environmental resource, including demonstration that the development will not adversely impact the hydrology or water quality of the resource. (Effective 6/11/08)

The Commission may have all and any of the information provided by the applicant reviewed by independent professional experts, as selected by the Commission, the cost of these experts to be paid by the applicant, in accordance with Section 8.178 of these regulations. (Effective 6/11/08)

## 8.12 All other applications for permits

- A. Upon determination of the Zoning Enforcement Officer that the proposed structure or use complies with these Regulations, the Zoning Enforcement Officer shall issue the required permit. One (1) copy of the approved plot plan shall be returned to the applicant with the approval and number of the permit noted thereon. A copy of the approved plans shall be available for inspection at the premises during regular working hours.
- B. Before the actual placement of the foundation of any structure, the contractor or owner may be required to stake out the location of the proposed foundation and supply the Zoning Enforcement Officer with a certificate by a licensed surveyor that the location of such foundation, or any part of a structure to be erected thereon, complies with the approved plot plan.

## **8.13 Public Hearings**

- A. Written Notice: Whenever an application is made pursuant to these Regulations that requires a public hearing or it is determined by the Commission or the Zoning Enforcement Officer the use or activity may have an effect on the abutters and a public hearing is required; the applicant shall in addition to any other requirements by the regulations of the Town of Stafford or Connecticut General Statute meet the following requirements:
  - 1. For special use permits, send written notice to the owner of record for each parcel within one hundred feet (100') of the property for that the permit is sought; notifying them of the filing of such application and the date and time at that the public hearing will be commenced on such application.
  - 2. For a zone change, send written notice to the owner of record for each parcel within five hundred feet (500') of the property for that the permit is sought; notifying them

- of the filing of such application and the date and time at that the public hearing will be commenced on such application.
- 3. For subdivision and resubdivision, send written notice to the owner of record for each abutting parcel of the property for that the application is sought; notifying them of the filing of such application and the date and time at that the public hearing will be commenced on such application.
- 4. For variances, send written notice to the owner of record for each abutting parcel of the property for that the application is sought; notifying them of the filing of such application and the date and time at that the public hearing will be commenced on such application.
- 5. Such notices shall be mailed by certified mail, return receipt requested, or by "certificate of mailing" at least ten (10) days before the commencement of the public hearing, and proof of mailing shall be provided to the Commission before the conclusion of the public hearing.

## **8.14 Zoning Amendments**

- A. A proposed amendment to the Zoning Regulations shall be submitted in correct legal form accompanied by a letter of explanation of why the change is being requested.
- B. A proposed amendment to the Zoning Map shall include a legal description of the area proposed to be changed and three (3) prints of a map drawn to scale showing the proposed change in the boundaries of the zoning map. All properties within five hundred (500) feet of the proposed change shall be identified as to use and owners' names and mailing addresses for all properties indicated. Uses and addresses may be supplied as attachments. Supplying of such information is not required for comprehensive or large area revisions.
- C. The applicant shall post a sign, available from the Zoning Enforcement Officer, giving notice of the application, in a conspicuous place on the property that a zone change is sought, visible from a public street. Said sign shall be posted ten (10) days before the public hearing, and it shall be removed not later than three (3) days after the close of the public hearing.
- D. Applicants requesting an amendment to the zoning map shall provide a preliminary site plan for property to be developed under the proposed zone. Said plan shall show proposed buildings and uses, parking, preliminary grading, driveway locations, and other features that will aid the Commission in its deliberations. Supporting reports such as traffic studies, feasibility studies, etc., shall be submitted at the time of application in order to allow adequate review by the Commission and its staff. These requirements shall not apply to proposals by the Commission.
- E. If a proposed change to the zoning map is approved, it shall be the responsibility of the applicant to provide a transparent print of the change on permanent material to the Commission for filing with the Town Clerk. Said map is to be submitted within seven (7) days of approval by the Commission accompanied by the required filing fee.

#### 8.15 Enforcement:

A Zoning Enforcement Officer designated by the Commission shall administer and enforce these Regulations. Said Officer may be provided with the assistance of such other persons as the Commission may direct.

If the Zoning Enforcement Officer shall find that any of the provisions of these Regulations are being violated, he/she shall notify the person responsible for such violations, indicating the nature of the violations, and ordering the action necessary to correct same. Said official shall, order, in writing, the discontinuance of illegal use of land, buildings, or structures; removal of illegal buildings, structures, or of additions, alterations, or other structural changes thereto; discontinuance of any illegal work being done; or shall take any other action authorized by these Regulations; or by provisions of the Connecticut General Statutes to ensure compliance with, or prevent a violation of its provision.

## 8.16 Zoning Board of Appeals:

The Zoning Board of Appeals, in a specific case, after public notice and hearing and subject to proper safeguards, may determine and vary the application of these Regulations herein prescribed in such manner as shall, in the interests of public welfare and in fairness to individuals, best carry out the general purpose and intent hereof.

#### A. Powers and Duties

- 1. To hear and decide appeals where it is alleged that there is an error in any order, requirements, or decision made by any official charged with the enforcement of these Regulations.
- 2. To hear and decide all matters upon that it is required to pass by the Zoning Regulations.
- 3. To determine and vary any requirement of the Zoning Regulations in harmony with their general purposes and intent so that substantial justice may be done. This authority shall be exercised solely in instances where there are practical difficulties or unnecessary hardships in the way of carrying out the strict letter of these Regulations and shall maintain the security of public health, safety, and welfare.

#### **B. Variances:**

No variance in the strict application of any of the requirements of the Zoning Regulations shall be granted unless the Zoning Board of Appeals shall find:

1. That there are conditions fully described in the findings of the Board, especially affecting the land or structure for that the variance is sought, that conditions are peculiar to such land or structure and not to the personal or financial circumstances of the appellant and that conditions are not affecting generally the area in that such land or structure is situated and have not resulted from any act of the appellant subsequent to the enactment of the requirement appealed from. Minor infractions of the location or height of a structure, or the dimension or area of a lot caused by

human error may be considered sufficient cause for an affirmative finding of this paragraph unless, in the opinion of the Board of Appeals, such infractions can be repaired without impairing the use of the land or structure.

- 2. That the aforesaid conditions are such that the strict application of the requirements of the Zoning Regulations would deprive the appellant of the reasonable use of the land or structure, and that the variance, as granted by the Board, is the minimum variance that will accomplish this purpose.
- 3. That the granting of the variance would not permit the property to be used for a purpose that is denied to the occupants of other properties in the same zoning district.
- 4. That the granting of the variance is in harmony with the general purpose and intent of the Zoning Regulations and will not be detrimental to public health, safety, convenience, welfare, and property values.
- 5. No use variance shall be granted unless the Board sets forth in its findings the reasons that the land or building(s) in question could not be appropriately used for any of the permitted uses of the zone in that the site is situated and the reasons that the proposed use is more appropriate than the permitted uses. In no case, however, shall the Board approve a use variance that would allow a use that is not permitted in any zone or that is specifically prohibited.
- C. If a variance is granted, it shall be the responsibility of the applicant to file required documentation, as required by CGS Sec. 8-3d, with the Town Clerk.

## Recording of Findings:

The records of the Board of Appeals shall include specific reasons for each of the findings above to be made before a variance may be granted. The Board of Appeals shall attach such conditions and safeguards as are necessary to insure continued compliance with the terms of the variance.

The applicant shall post a sign, available from the Zoning Enforcement Officer, giving notice of the application, in a conspicuous place on the property for that a variance or permit is sought visible from a public street. Said sign shall be posted ten (10) days before the public hearing, and it shall be removed not later than three (3) days after the close of the public hearing.

### D. Rules and Procedures:

The Zoning Board of Appeals shall adopt such rules and regulations as may be deemed necessary to carry out the provisions of this Section.

1. All appeals and applications made to the Board shall be in writing on forms prescribed by the Board. Each appeal or application shall fully set forth the circumstances of the case and, where applicable, shall reasonably illustrate with map(s) and other drawings the nature and location of the appeal or request. Every appeal or application shall refer to the specific provision of the regulation involved and shall exactly set forth, as the case may be, the interpretation that is claimed or

the details of the variance that is applied for and the grounds on that it is claimed that same should be granted.

- 2. The concurring vote of four (4) members of the Board shall be necessary to reverse any order, requirement, or decision of the Zoning Enforcement Officer or to decide in favor of the applicant any matter upon that it is required to pass under any provision of these regulations or to vary the application of these regulations.
- 3. The time limit for filing an appeal shall be as per Section 8-7, Connecticut General Statutes.
- 4. All applications and appeals shall be accompanied by a fee as set by Town ordinance

## 8.17 Schedule of fees: Repealed, See Town Ordinance

## 8.18 Fees for Outside Experts:

The Commission may, at its discretion, hire or engage outside experts to assist in its evaluation of any application for site plan approval, by special exception. The total costs for all outside expertise shall be borne by the applicant. The Commission may require an initial payment, to be determined, for the hiring of these experts, prior to its review of the application. This payment shall be considered as an integral component of the application, and the failure of the applicant to make this payment shall render the application incomplete. If the applicant fails to pay the fee within thirty (30) days of receiving an invoice, the Commission may revoke all approvals of the application and pursue all necessary action to receive payment.

## 8.19 Continuity clause

- A. All terms and conditions of special use permits granted by the Commission or by variances granted by the Board of Appeals or of any other conditions of approval granted prior to September 15, 1994, shall remain in force unless amended by any subsequent action under these Zoning Regulations.
- B. Except as hereafter stated, it is not intended by these Regulations to repeal or in any way impair any existing provisions of law or ordinance or any rule and regulation previously adopted or that shall be adopted; nor is it intended by these regulations to interfere with any covenants or other agreements between parties provided, however, that where these Regulations impose a greater restriction upon the use of land and structures or any other larger requirements than are imposed by any provisions of law or ordinance or by any such rules and regulations or by any such covenants or other agreements the provisions of these Regulations shall hereafter prevail.

## 8.20 Validity:

If any section, paragraph, subdivision, clause, or provision of this act shall be adjudged invalid or unconstitutional for any reason, such adjudication shall apply only to the section, paragraph, subdivision, clause, or provisions upon that such adjudication is based, and the remainder of this act shall be deemed to be and shall continue to be valid and in full force and effect.

#### **8.21 Amendments:**

The provisions of this act may from time to time be amended, changed, or repealed as provided in Section 8-3 of the Connecticut General Statutes and in accordance with Sec. 6.40 of these Regulations.

## 8.22 Repealing clause:

The provisions of the Zoning Regulations of the Town of Stafford as originally adopted on March 23, 1972, and the Zoning Regulations of the Borough of Stafford Springs as originally adopted on April 7, 1972, and subsequently revised and amended are repealed as of the effective date of these Regulations, except that all terms and conditions attached to special use permits approved by the Commission and variances approved by the Zoning Board of Appeal shall remain in effect.

### 8.23 Revised Date: March 9, 1999

- 1. Effective Date: March 31, 1999
- 2. Amended as noted October 28, 2003, effective date of November 15, 2003.
- 3. Amended as noted November 25, 2003, effective date of December 19, 2003.
- 4. Amended as noted November 23, 2004, effective date of December 15, 2004.
- 5. Amended as noted April 6, 2005, effective date of April 6, 2005.
- 6. Amended as noted April 26, 2005, effective date of April 26, 2005.
- 7. Amended as noted July 26, 2005, effective date of July 27, 2005.
- 8. Amended as noted March 14, 2006, effective date of April 1, 2006.
- 9. Amended as noted March 28, 2006, effective date of April 1, 2006.
- 10. Amended as noted April 25, 2007.
- 11. Amended as noted June 11, 2008.
- 12. Amended as noted July 8, 2008.
- 13. Amended as noted August 12, 2008.
- 14. Amended as noted September 28, 2010.
- 15. Addition of Highway Industrial District April 28, 2015.
- 16. Addition of Village District March 8, 2016.
- 17. Complete Reorganization Approved May 9, 2017, effective date of May 30, 2017.
- 18. Amended October 19, 2017.
- 19. Amended April 12, 2018.
- 20. Amended August 16, 2018.
- 21. Amended March 12, 2019
- 22. Amended July 31, 2019
- 23. Amended July 16, 2020
- 24. Amended October 1, 2020
- 25. Amended November 1, 2021
- 26. Amended March 1, 2022
- 27. Amended June 7, 2022
- 28. Amended August 3, 2023
- 29. Amended October 1, 2023
- 30. Amended December 1, 2023
- 31. Amended July 1, 2024 \_Heliports

- 32. Amended November 6, 2024 \_ Fees
- 33. Amended March 17, 2025 \_ food trucks
- 34. Amended April 17, 2025 \_3.21 Existing Non- Conforming Uses